# Queer Reproductive Decision-Making in Saskatoon: Pandemically Complicated

A Thesis Submitted to the College of
Graduate and Postdoctoral Studies
In Partial Fulfillment of the Requirements
For the Degree of Master of Arts
In the Department of Archaeology and Anthropology
University of Saskatchewan
Saskatoon

By Jessica Jack

© Copyright, Jessica Jack, March 2023. All rights reserved.

Unless otherwise noted, copyright of the material in this thesis belongs to the author.

#### PERMISSION TO USE

In presenting this thesis/dissertation in partial fulfillment of the requirements for a Postgraduate degree from the University of Saskatchewan, I agree that the Libraries of this University may make it freely available for inspection. I further agree that permission for copying of this thesis/dissertation in any manner, in whole or in part, for scholarly purposes may be granted by the professor or professors who supervised my thesis/dissertation work or, in their absence, by the Head of the Department or the Dean of the College in which my thesis work was done. It is understood that any copying or publication or use of this thesis/dissertation or parts thereof for financial gain shall not be allowed without my written permission. It is also understood that due recognition shall be given to me and to the University of Saskatchewan in any scholarly use which may be made of any material in my thesis/dissertation.

Requests for permission to copy or to make other uses of materials in this thesis/dissertation in whole or part should be addressed to:

Head of the Department of Archaeology and Anthropology 55 Campus Drive University of Saskatchewan Saskatoon, Saskatchewan S7N 5B1 Canada

OR

Dean

College of Graduate and Postdoctoral Studies University of Saskatchewan 116 Thorvaldson Building, 110 Science Place Saskatoon, Saskatchewan S7N 5C9 Canada

#### **ABSTRACT**

This thesis examines how eighteen queer people living in Saskatoon, Saskatchewan, made their reproductive decisions during the COVID-19 pandemic. Through the twin lenses of queer theory and Cheryl Mattingly's critical phenomenology 2.0, I have found that queer people in Saskatoon are influenced in their reproductive decision-making by meanings they have created around queer and reproductive safety, the family structures they feel are best and are supported making, and how COVID has impacted meanings of place. One-on-one remote interviews elicited three significant areas that influenced queer reproductive decision-making. The first was how queer family structures are formed, including an analysis of the nuclear family and the ways that approach does or does not work for queer families, the gendered problems queer people face when contemplating pregnancy, and the impacts of COVID-19 on how queer families are formed. The second centres on safety, with people born and raised in Saskatchewan prioritizing social safety and people born in different, sometimes less queer-supporting countries prioritizing physical safety when making reproductive decisions. The third is the relationship between COVID-19 and place, dissecting how the COVID-19 pandemic has changed the meanings of place through social distancing and isolation, and how the space between places (i.e. travel, remote connection software like Zoom) has changed meaning during this pandemic, followed by how COVID-19 has specifically impacted queer peoples' reproductive decisions. Further research on the current queer experience in the prairie provinces is needed to explore this topic, as qualitative research with queer communities in places like Saskatoon is lacking.

#### **ACKNOWLEDGEMENTS**

I would like to thank my supervisor, Pamela Downe, for her support, guidance, and assistance throughout this project and the writing of this thesis, as well as Sylvia Abonyi and Karen Lawson for their comments and advice as members of my committee. I would also like to thank James Waldram for his Socratic training that prepared me for the rigours of graduate-level research. In addition, I would like to thank my fellow graduate students Caitlin Zbeetnoff Woloschuk, Kelsey Marr, Alexandria Pavelich, and Shaneice Fletcher-Hildebrand for being my research community through the isolation brought on by the COVID-19 pandemic – their support and insights have been invaluable in this work. I would also like to thank my husband David for his endless love and care. Finally, I would like to thank everyone who shared their time and experiences with me to make this research happen, as this thesis is a love letter to them and the bits of their lives they have shared with me.

# TABLE OF CONTENTS

PERMISSION TO USE	ii
ABSTRACT	iii
ACKNOWLEDGEMENTS	iv
TABLE OF CONTENTS	V
LIST OF TABLES	vii
List of TermsError! Bookman	rk not defined.
Gender:	2
Sexuality:	3
Chapter 1 – Introduction	4
1.1 – Vignette	4
1.2 - The Building Blocks of This Project	6
1.3 - Literature Review	12
1.4 - Theoretical Perspectives	15
1.5 - Research Question and Thesis Layout	20
Chapter 2 - Methods	22
2.1 - Unpredictable Changes – The Impact of COVID-19 on Methodology	22
2.2 - Recruitment	23
2.3 - Individual Interviews	23
2.4 - Analysis	25
2.5 - Participants	26
2.6 - Timeline	30
Chapter 3 - Queer Family Structures	31
3.1 – Introduction	31
3.2 - Exploding the Nuclear	33
3.3 - Pregnancy	39
3.4 - COVID-19	42
3.5 - Conclusion	48
Chapter 4 – Safety	50
4.1 Introduction	50

4.2 - Defining Safety	51
4.3 - Safety in Place - Queer Reproductive Safety Between Countries	54
4.4 - Safety in Place - Queer Reproductive Safety in Saskatoon	58
4.5 - Conclusion	63
Chapter 5 – Normativity	65
5.1 – Introduction	65
5.2 - Home-grown Norms in Saskatoon	66
5.3 - Norms From Brazil	72
5.4 - Conclusion	78
Chapter 6 – COVID-19 and Place	81
6.1 - Defining Place	81
6.2 - How COVID Has Changed Meanings of Place	84
6.3 - Creating New Meanings of the Space Between Places	90
6.4 - COVID-19 and Reproduction	93
6.5 - Conclusion	96
Chapter 7 - Conclusion	99
7.1 – Vignette	99
7.2 – Limitations	102
7.3 – Final Thoughts	103
BIBLIOGRAPHY	105
APPENDIX	142
A - Recruitment Advertisement	142
R - Interview Guide	143

# LIST OF TABLES

27
2

#### LIST OF TERMS

Author note: this is not intended to be a comprehensive compendium of 2SLGBTQIA+ terms. It is limited to the terms I discuss in this thesis. For a thorough (but not exhaustive) list of 2SLGBTQIA+ terminology, visit <a href="https://www.glaad.org/reference/terms">https://www.glaad.org/reference/terms</a>

**2SLGBTQIA+**: an inclusive but not exhaustive acronym used to describe people who fall outside of the gender and sexuality mainstream identities of cisgendered and heterosexual. This specific configuration of the acronyms is Canadian in origin, as other countries focus on different identity categories. Broken down, it represents:

- 2S = Two Spirited, an umbrella term for Turtle Island (North American) Indigenous gender identities that do not align with binary man/woman designations, but can sometimes be used to refer to all non-straight/non-cis Indigenous people (Filice 2020)
- L = lesbian (see below)
- G = gay (see below)
- B = bisexual (see below)
- T = transgender (see below)
- Q = queer and/or questioning as in questioning their gender/sexual orientation
- I = intersex, or people whose bodies fall outside of the binary state of male or female (Planned Parenthood n.d.)
- A = asexual (see below)
- += other gender or sexual identities not covered in this acronym. The world of gender and sexual identities is vast and everchanging, so no usable acronym will ever cover all identities but acknowledging their existence is important.

### Gender:

**Cis-gender**: often shortened to cis, cis-gender refers to someone who is not transgender (GLAAD 2022a). However, a more informational definition is someone whose identity aligns with the sex they were designated at birth (Planned Parenthood 2022).

Designated Female/Male At Birth (DFAB/DMAB): someone born with the anatomy generally associated with a binary biological sex who has therefore been designated a "female" or "male" at birth (Harrington 2016). This term is used to describe how people have been gendered in their childhood, which may or may not be related to how they are currently choosing to gender themselves. For example, I am a non-binary person who was designated female at birth (DFAB), and I use the DFAB acronym to explain how I was socialized and grew up thinking about my gender as important contextual information.

**Non-binary**: sometimes said to fall under the trans umbrella, refers to someone whose gender identity falls outside of the societal norm of day-to-day life adhering unchangingly to one of the binary genders, i.e. girls and boys, women and men (adapted from Matsuno and Budge 2017). They may have a singular identity that falls somewhere between or outside of the binary male and female identities, they may shift in their gender identity by experiencing being female or being male at separate times, or they may not have a gender identity at all.

**Trans-gender**: someone whose gender, in either identity or expression, differs from societal norms expected of others who share that person's designated sex at birth (King, Hughto, and Operario 2020). The most visible examples of trans-gender people are those who transition from one binary identity to the other, from female to male or from male to female. However, this is not how all trans people experience their gender and it can be an ongoing and shifting relationship with gender through the lifespan.

# **Sexuality:**

Asexual: often shortened to "ace", asexuality is defined by low or absent interest in sexual activity (Bogaert 2015). Asexuality does not inherently imply aromanticism – people who identify as asexual may still experience romantic attraction and seek out romantic relationships, so they may also identify as bisexual or lesbian or gay or straight. There are many different approaches to asexuality and have led to sub-groups within asexuality, including "grey-ace", which refers to the spectrum between sexuality and asexuality, where a person may experience sexual attraction on occasion or in specific situations, but still identifies as someone on the asexuality spectrum (Decker 2015). People who are not asexual, who experience sexual attraction, are allosexual.

**Bisexual**: someone who is attracted to people of both of the binary genders (Halperin 2009).

Grey-ace: see Asexual

**Gynesexual/gynephilic**: someone who is primarily attracted to femininity, which includes gender femininity (such as the way people dress, act, or speak) and physical aspects of femininity (including female-coded genitals or other female-coded body parts like breasts) (GLAAD 2022b).

**Gay**: men who are attracted to other men (again, referring to men of all orientations), though the word "gay" is also used more broadly to refer to anyone who is attracted to someone of the same gender as themselves (GLAAD 2022b). Unlike the word "lesbian" for women, there is not a separate term for men who are attracted to men, so often the word "gay" refers to men.

**Heterosexual**: also called "straight" or "hetero/het", heterosexuality is most commonly used to refer to people who identify within the gender binary of male/female and are attracted to people of the opposite binary gender as themselves. More broadly, "straight" can refer to someone who is attracted to someone of a gender or sex different than their own and does not need to be constrained by the gender binary (GLAAD 2022b).

**Lesbian**: women who are attracted to other women, a deceptively simple definition that requires me to clarify that I use the term to refer to women of all orientations (trans, cis, non-binary, etc.) (Tate 2012).

**Pansexual**: someone who has the potential to be attracted to people of all genders, regardless of the specificity of their gender identity or sexuality (Green 2019).

## Chapter 1 – Introduction

# 1.1 - Vignette

It is a hot June day, and I am in my bedroom, sitting on a kitchen chair in front of a makeshift desk that is actually a cutting board, a stack of my broadest books, and desperation. In my bedroom-cum-office, my laptop is precariously perched on this pile of pilfered plunder turned pulpit. It has been a year and four months since the COVID-19 pandemic began, and I am conducting interviews in as private a way as I can while sharing a small one-bedroom apartment with a partner who is also working from home. With my headphones in and a fan outside the door to reduce the sound carrying to the next room, I am talking to people who live in the same city but must remain distanced from me for their safety and mine. On this day, the person I am talking to is James. I watch them on my screen in their little Zoom box, moving a bit jerkily as our shared internet connection destabilizes now and again. As a self-professed "non-binary bisexual queer woman" in their early thirties who uses both they/them and she/her pronouns and lives in Saskatoon, James and I have much in common. To understand more about how James has made reproductive decisions throughout their life, they are telling me about their family and the context in which they experienced being a child.

Yeah, so my parents are both lovely, wonderful people. We've had a lot of healing. But they were not prepared to have kids when they had us. They wanted us, my mom wanted us, but that's different from being ready for us, right? And so there was that, there's a lot of intergenerational trauma in our family that I only found out about as an adult, which was like, "Oh! Many things make a lot more sense now." So there's that. My dad is a lot like me, which meant that he lived in our backyard mostly, so we didn't really see him even though he lived with us. And then my mother has had ongoing mental health struggles which she takes very seriously and takes care of very well and is very responsible about managing and is actually very inspirational about managing.

James is explaining to me the dynamics they experienced as normal in their life growing up, their absent-though-present father and their unprepared mother, as well as the intergenerational suffering passed on through family lines. Through hard work and love, they have since healed

<sup>&</sup>lt;sup>1</sup> For a definition of all queer terminology used in this thesis, see the list of terms.

the pains caused by these experiences. Still, when child James experienced these familial hardships, those experiences gave way to particular meanings about what role children play in one's life and what building a family means.

Though James was born with anatomy usually associated with females and was thus raised as a girl, as they became an adult, they realized that the parenting role with which they most closely associated was that of a father. Their own father took on the family position of "logistical support and breadwinner" as opposed to their mother's role of "caretaker and nurturer," and James felt they were far better equipped to provide that same logistical support and financial security. As James reckoned with their gender identity and came to identify as non-binary, the parental role of father became even more appealing because it allowed them to occupy a place in a family that was not dependent on the configuration of their body. As they explained, James eventually came to reject the notion that having particular anatomy meant they had to fit into a specific way of creating a family, leaving them free to explore an approach to family that fit them and their way of living in the world. That exploration continued to evolve as James matured and experienced more of the world, and they have now come to see family as something that does not have to include adults in the role of a parent.

As a result of these and a wide variety of other highly influential factors, James is now in a long-term committed relationship with a man who has a son, and they cohabitate together as a family. However, James has chosen a specific role for themselves within their partner's son's life – James is not a parent, and the family does not treat them like one. Instead, James is a trusted confidant, an adult whom the son can trust for advice, guidance, and non-parental love. In James' words, they became "the giver of information." James and their family do not see James as a stepparent, but that does not preclude them from having an important and loving place in their family. James also took their cue from James' father and built their own space in the backyard, a small building where they could work and sleep if they needed some space of their own. This blend of independence and non-traditional family formation arises from the specific meanings James has found within their own reproductive decisions and the decisions made by their partner, their partner's son, and the partner's son's mother.

The COVID-19 pandemic has also influenced James in their position as an informational and supportive figure instead of a parent within their family configuration. During the prevaccination period, when social distancing mandates and other preventative measures were

heaviest and most restrictive, James and their partner were taking care of their partner's son for longer stretches of time as they reduced the number of transfers between the son's father and his mother, who lived in a different household. These reduced transfers were intended to diminish the number of exposures between household bubbles for everyone's sake but, most importantly, for James's sake, as they have an autoimmune condition that could make a COVID-19 infection especially dangerous. To help the tween son navigate this change in social dynamics and to help protect him, James stepped up their information-giving role in his life. Instead of telling their partner's son what to do and how to do it, James offers advice and data for this young man to consider. James has explained how this position has been advantageous in the family dynamic because the lack of parental-style emotional attachment has created a sense of freedom and trust between James and their partner's son, allowing them to exchange information on subjects where the son would be less comfortable discussing them with his parents, such as dating and sex, gender and sexual identity, difficult peer relationships, and now the COVID-19 pandemic as well. As James puts it, "Yeah. I don't think he worries about disappointing me in the way that he worries about his parents." Even during a problematic pandemic, James' unique relationship with concepts of parenthood and their queerness has helped to strengthen their family dynamic while continuing to solidify James' identity of not being a parent.

I have been graciously and generously given a glimpse into James' life and how they see meaning in their reproductive decision-making processes as a queer person living in Saskatoon during the COVID-19 pandemic. Through a screen and speakers, they have shared their thoughts and opinions about the most critical aspects of how they see the idea of parenthood in this place, this time, and their particular context of gender and sexual identity. I am immensely grateful for this gift of time and insight because these questions surrounding how queer people make their reproductive decisions are near and dear to my heart as they are relevant and vital aspects of the various cultural and social dynamics within those decisions. So how did I get to talking to James about their family life?

## 1.2 - The Building Blocks of This Project

The experiences and meanings about reproductive decisions that James described exemplify why I felt drawn to exploring how queer people make their reproductive decisions.

My own family experiences align closely with those of James – unhealthy parents who did their

best but left me with pain surrounding my gender identity and reproduction. In my case, I have chosen not to have children because of my own experiences. After speaking to those who have shared their stories with me, I can trace the decisions I have made through my experiences and the meanings I have associated with those experiences. As a queer person, I am drawn to both the specific factors queer people must deal with in their reproductive journeys and the broader factors that affect every person in terms of reproduction. James' story perfectly highlights this blend of factors — everyone grows up with other people and can be impacted by those people, but navigating the gendered expectations of parenthood while not conforming to those genders is generally an experience specific to queer people. Due to the context in which I explored this curiosity around reproductive decisions, the scope of my research had to expand beyond the aspects of gender and sexual identity in relation to those decisions. And, as James' story demonstrates, much of reproductive decision-making relies on ideas of family and kin.

Anthropological considerations of reproductive decision-making must include a consideration of kinship. How do people conceive of the connections between themselves and others that constitute the relationships we call "kin" or "family"? Because of the cultural variations in who is considered related and understanding how kinship is built, there is no unifying approach or definition of the term. There are, however, some commonalities that should be reviewed. Kinship is not always reliant on the biological connections between kin and family, especially queer families, as they are often made up of people who share no genetic relation. This does not mean that biological connections can be completely ignored in understandings of kinship because reproduction must ultimately include the genetic processes that go into creating human beings (Scheffler 1973). It seems difficult to create an anthropological definition of kinship that completely avoids the biological aspect because, even if kinship is not about genetic relationships between people, it may involve shared biological processes such as eating together (Cucciari 1981).

However, kinship includes many other influences that create and reaffirm cultural perspectives on who is related. This is because kinship and the question of familial-style relations involves many other large questions. How do institutional definitions, such as political, legal, and educational, impact who is considered family (Ginsburg and Rapp 1991)? How do cultural understandings of gender and the roles gendered people play in kinship and relationality influence who builds families and how they are recognized (Dorow 2006)? How do kinship

structures and definitions respond to and incorporate large-scale events like climate change, disasters, and pandemics like COVID-19 (Faas and Barrios 2015; Hoffma and Barrios 2019)? Systems of relations that form family and kin, in their ever-evolving and highly culturally variable arrangements, play important roles in how we relate to the world and how the world relates to us (Strathern 1992). I have highlighted these particular aspects of kinship because they are most salient to the reproductive decision-making processes of the queer people in Saskatoon during the COVID-19 pandemic but it is by no means an exhaustive list of the important dimensions that should be considered in kinship and family formation.

It is also important that I delineate the terms I use throughout this thesis. I use the terms parenthood, family formation, and reproductive decision-making frequently because, while distinct, these terms are closely related. In the mainstream, the term "reproductive decision-making" may seem to be about making a decision about whether or not to reproduce in a biological sense with discussions of fertility rates often accompanying the term (Testa, Sobotka, and Morgan 2011), but in the practice of social science, the term can also be used to represent other aspects of family formation. Reproductive decision-making includes "the decision to have a child" (Klobas and Ajzen 2015), a decision which does not inherently require the people making the decision to be biologically producing the child themselves. The term therefore bleeds over into the idea of family formation or making families, a process that may involve decisions about adding family members without that family engaging in biological reproduction. These families can be made in many ways, including through adoption, surrogacy, or creating families through connections to non-related people like friends or community members, a process often called "found family" or "family of choice" (ten Have and Patrão Neves 2021; Schwartz 2016; Dewaele et al. 2011; Moore and Stambolis-Ruhstorfer 2013).

Family formation does not have to include parenthood, as families can be made without parental figures, but the context of reproductive decision-making means that, for this thesis, parenthood is an important aspect of those decisions. Parenthood is difficult to define because different institutions and approaches have a vested interest in defining parenthood their own way. For example, social definitions may differ from legal definitions, and legal definitions will differ from context to context and place to place (Meyer 2006; Nejaime 2017; Glazer 2014). I approach the term anthropologically and from the position of relatedness as discussed above which make a single definition of parenthood difficult, as the concept means different things to different people

again depending on their context. At its core, parenthood involves taking care of people with whom you have a relationship, but the specificities of that care and the nature of the relationship can be so broadly variable as to be difficult to define specifically (McKenzie 2022), and in even in the context of the Saskatonian interlocutors who shared their stories with me, parenthood meant many things. It is the interrelation between the three concepts of reproductive decision-making, family formation, and parenthood that leads me to use these concepts within this thesis. Anthropologically, I find it difficult to clearly delineate between these concepts when discussing a queer population. Reproductive decisions can be made on the basis of individual definitions of parenthood and in families that may not look like families to a non-queer observer, but the formation of these families and the decisions made within those families about how they are formed cannot be easily separated.

When I set out to research reproductive decision-making in Saskatoon, a city of 270,000 inhabitants on the Canadian prairies, I did not anticipate the highly global nature of the responses I would be receiving. This was perhaps an oversight on my part introduced by my bias as a White settler who has lived in Saskatoon all my life, as I failed to consider the global connections that are inherent to the multicultural makeup of the city. Immigration both past and present, as well as ongoing ties between Saskatonians and countries around the world, mean that it is impossible to consider Saskatoon in isolation from global connections (Saskatchewan Bureau of Statistics 2022; Dayal 2022). I was reminded of this fundamental fact by those who chose to share their stories with me. As my recruitment process continued, the people who participated in my research came from a wide variety of cultural backgrounds and experiences who came to Saskatoon for multiple purposes. I must thus include the questions of globalization and global citizenship in this thesis. Families and kinship are increasingly formed across the borders of countries, including transnational adoption, immigration and relocation, and digital communication that allows for the development of important kin relationships between people living in different cultural milieux (Dorow 2006; Ebaugh and Curry 2000; Shore, Goddard, and Llobera 2021; Matlock 2012; Hjorth 2022).

Part of the institutional definitions of family and kinship include who creates what kinds of families in order to develop the 'right' kinds of citizens for that particular institutional place, so these multinational/transnational approaches to kinship can complicate understandings and expectations of kin formation (Rabo 1997; Levitt 2018). Queer identity is also only one facet of

the identities that shape and inform experiences and definitions of family and reproduction. Like everyone else, queer peoples' identities are also intersectional. Queer people are sometimes immigrants who become ethnic minorities in their new location, or they sometimes become ethnic minorities when introduced to their hometowns through transnational adoption. Queer people who must also contend with other intersectional elements of class, gender and sexuality identity and many other factors besides, helps to highlight the intersections between family formations and global citizenship which are deeply complex (Mazzei and O'Brien 2009; Khader 2013; Mahler, Chaudhuri, and Patil 2015). Though this field is enormous and full of unanswered and perhaps unanswerable questions, my thesis will only address a narrow slice of kinship questions that apply to the people who shared their stories with me, especially as these questions grow more complex with the addition of other factors like global pandemics.

As I mentioned at the beginning of this vignette, this research occurred at a time when the COVID-19 pandemic changed the landscape of everyday life quite seriously. On a personal level, I was as burnt out on the pandemic as everyone else I knew when data collection started in this research project. But I had chosen to conduct anthropological research and, as Sherry Ortner mentions in her discussion of dark anthropology, "academic work, at least in the social sciences, cannot be detached from the conditions of the real world in which it takes place" (2016, 47). In my academic position of privilege and power, I have an ethical obligation to conduct (relatively) timely research on the issues impacting people who do not share my privilege and power. Whether it be SARS, Ebola, Zika, climate change, or other pressing and impactful situations, anthropologists modify their research or set out on new projects to respond to these emergent situations (Siu 2008; Abramowitz 2016; Gray and Mishtal 2019; Cartwright 2019). The timing of the COVID-19 pandemic meant that I could not ignore it or its impact on the reproductive decisions being made, so the scope of my research shifted to include the COVID-19 pandemic. For some, thesevere disruptions to daily life that emerged in response to the infectiousness of COVID-19, including quarantines, social distancing, vaccination requirements, and masking, just to name a few, were undeniably impactful to reproduction even in the earliest days of this specific pandemic (Taekema 2020; Ewing 2020). As I explain later, this was a fruitful area of discussion. Many people who shared their stories with me found the pandemic had either changed their reproductive decisions or deeply impacted how they were thinking about reproduction.

The last piece of the puzzle in my research was in deciding where my research would take place. My undergraduate experience in anthropology excited me with notions of conducting research in locales quite unlike my hometown of Saskatoon, with stories of researching traditional medicine and illness in Belize, the relationship between humour and violence for prostitutes in Costa Rica, or the connection between health and land in Timor Leste from respected scholars like Dr. James Waldram, Dr. Pamela Downe, and Dr. Susanna Barnes (Waldram and Hatala 2022; Downe 1999; Palmer and Barnes 2021). These fieldwork examples showed the breadth of knowledge a researcher could gain from removing themselves from their comfort zone in search of information and experiences that can emerge within the unfamiliar. Sitting in my stiflingly hot bedroom on that June day, more than a year into the COVID-19 pandemic and still largely isolated from the communities in which I was usually situated, I looked around this room that I had spent at least ten thousand hours occupying and realized, it too could be unfamiliar.

Research around reproductive decision-making is not entirely unfamiliar to me. My research is a subset of the Deferred Motherhood project, a SSHRC Insight study conducted by Drs. Karen Lawson and Pamela Downe. The larger project is aimed at understanding how and when women form intentions on whether or not to defer motherhood using a cognitive-social model and the theory of planned behaviour. The Deferred Motherhood project examines the personal, the interpersonal, and the structural influences on how fertility intentions are formed with the intent of supporting women's abilities to meet their fertility aspirations. For my undergraduate honours thesis I analyzed a subset of the interviews that provide much of the data for the Deferred Motherhood project to understand how responsibility, community support, and personal context factor into reproductive intentions for these Canadian women. The research conducted by Dr. Lawson, Dr. Downe, and other associated graduate students throughout the lifetime of this project has provided me with a firm grounding in the subject matter and important experiences that have rendered the landscape of reproductive decision-making familiar to me. As I have mentioned, anthropological approaches to understanding complex subjects often benefit from seeing things in an unfamiliar way, so I decided to move my own research into an area unfamiliar to the Deferred Motherhood project: queer reproduction in Saskatoon.

As a queer person who has lived in Saskatoon for more than thirty years, I have intimate knowledge of the context in which queer Saskatonians live their daily lives, but that knowledge

is colloquial or the happenstance result of my everyday life. When I began to look at the queer people in this city through the lens of research, I realized that the familiar turf that is the queerness of Saskatoon was instead the suddenly unfamiliar terrain of a fieldwork site (Caputo 1999). I could interrogate everything I thought I knew about queerness in this city and immerse myself in the accounts of other peoples' experiences and their meanings to move beyond merely critiquing the social structures I see as contributing to queer existence and, instead, critically examine my assumptions underlying those critiques (Mattingly 2019). To add to this intriguing lack of familiarity was the fact that, no matter what site I chose for this research, I would be restricted by the safety guidelines surrounding research during the COVID-19 pandemic. Thus, remote options, like my eventual Zoom calls, were the most likely way I would gather the data for this project. I began to think of Saskatoon as the right place to investigate queer reproductive decision-making. Saskatoon has a storied queer history and, like most places in the world, queer people have long played an integral role in the formation and evolution of the city (Korinek 2018). Yet research into contemporary queer experiences in this relatively conservative city is uncommon. I could not find any research documenting experiences with queer reproductive decision-making centred on Saskatoon, Saskatchewan, or the prairie provinces. The research I found about queer reproductive decision-making during COVID-19 played a vital role in this work, and I will now explore it by unpacking why this research is important.

### 1.3 - Literature Review

The question of why this research on queer reproductive decision-making is important has a many-faceted answer. Starting on a global scale, the continuously increasing global population, in combination with other global forces like climate change and unequal resource distribution, raises concerns about who is reproducing and why they are doing it (Battin 1997; Hartmann 1995; Bodin and Björklund 2022). In Canada specifically, the trends of note come largely from the difference between fertility rates and age regarding the country's demographics. Canadian fertility rates have been declining since 2009, and 2020 saw a fertility rate of just 1.4 children per Canadian woman (Statistics Canada 2021c). Canadian mothers are also trending towards older maternal ages, with the mean maternal age steadily rising from 27.8 years old in 1991 to 31.3 years old in 2020 (Statistics Canada 2021d). In addition, the Canadian-born population of Canada is aging – by 2030, 23% of Canadians will be over the age of 65, up from

15.6% in 2014 (Government of Canada 2016). An aging and declining population presents severe problems in terms of the labour force, availability of healthcare, and federal income and spending areas, so research into how reproductive decisions are made can have serious implications for the structural and cultural makeup of the country (Jackson, Clemens, and Palacios 2017; Gibbard 2018). The fertility and age variables must also contend with the influence of immigration, as the country's population increased by 5.2% from 2016 to 2021 due to the highest rates of immigration within the G7 countries (Government of Canada 2022, 7). Much of Canada's labour force is bolstered by immigrants to Canada, though it is not enough to fully offset the effects of the aging Canadian population (Statistics Canada 2022a).

None of the participants in this study self-identify as Indigenous. However, all newcomer and settler participants are living and working in Saskatchewan where demographic population trends for Indigenous and non-Indigenous people are quite distinct. Within Canada, the Indigenous population is increasing almost double the rate of non-Indigenous people, and the Indigenous population within Canada is also 8.2 years younger on average, with a higher fertility rate and a lower average maternal age than the non-Indigenous population (Statistics Canada 2016; 2022b; Boulet and Badets 2017; Sheppard et al. 2017). Statistics Canada is predicting that the Indigenous population within Canada could increase up to 56% more than its current number, growing from the current 1.8 million to up to 3.2 million Indigenous people within Canada by 2042 (Statistics Canada 2022b). The differences of the demographics and reproductive trends between Indigenous and non-Indigenous people in Canada could hold fascinating potential for the impact of Canada's declining and aging population, as well as presenting the possibility for some cultural shifts around kinship and reproduction. Interestingly, none of the participants spoke to this distinction directly, focusing instead on the trends that they assess to be more immediate in their daily lives.

To narrow my point of focus even further, my interest in reproductive decision-making was not piqued by whether or not someone decides to have children but by the meanings those decisions held for the people making them and the experiences that influenced the decisions being made. This is a relatively unexplored niche of Canadian reproductive research. However, previous studies have highlighted the importance of fertility education, changes in the meaning of reproductive decisions from experiencing illnesses like Multiple Sclerosis or infection of the Human Immunodeficiency Virus, and the impacts of intersectional identities of assigned or

assumed membership within ethnic groups (Williamson et al. 2014; Alwan et al. 2013; Leyva-Moral et al. 2018; Huertas-Zurriaga et al. 2022). My interest in queer reproductive decision-making further narrowed the field, especially in exploring queer people who choose not to have biologically related children or to have children living with them. Much of the extant research focuses on queer people who want children in these ways, with the subject of reproductive justice for queer people being common. Reproductive justice research looks at work done to guarantee reproductive rights for queer people and to move towards social equity in both freedom and safety regarding reproductive decisions for queer people (Cameron 2008; Ross and Solinger 2017). Existing research also focuses on gaining access to support and ways of making families that are often necessary for queer people to reproduce, and the interaction between the medicalization of queer bodies and the reproductive health system (Tam 2021; Harder 2021; Marvel 2016; Epstein 2018). Understanding the variables that have informed the decisions queer people have made about their reproduction, especially if they have chosen not to reproduce, highlights a poorly-represented segment of the queer population who may understand kinship differently than people who want to reproduce (Dahl 2014).

The last point about the importance of further exploring reproductive decision-making returns to the now-familiar bedfellow of COVID-19. Large-scale threats or disasters undeniably impact people's reproductive decisions (Wenham et al. 2020; Ariyabandu 2009; Freed et al. 2021). The COVID-19 pandemic has spurred a vast amount of research into COVID-19's effects on the institutions that surround peoples' reproductive decision-making, such as health services and schools (Stanton and Bateson 2021; Luttik et al. 2020; Cachón-Zagalaz et al. 2020), changes to reproductive health and justice access specifically (Ferreira-Filho et al. 2020; Hashem et al. 2021), and exploring the impacts of COVID-19 on existing families with much focus on negative variables like family violence and stress (Humphreys, Myint, and Zeanah 2020; Xu et al. 2020; Lebow 2020). Reproductive decisions are often considered within a wider conversation about the pandemic's impact on reproductive health or access concerns (Nanda et al. 2020; Ferreira-Filho et al. 2020; Senderowicz and Higgins 2022). Research on how COVID-19 has specifically impacted fertility intentions or childbearing intentions has emerged across the world and has focused on issues such as physical and mental health in Shanghai, the role of restricted access to contraception on short-term decision-making in Moldova, financial insecurity and mental health in Poland, the intersectional impact on Black and queer women in the United States leading to

higher levels of delayed fertility intentions, and a lack of change to reproductive decisions in Kenya (Zhu et al. 2020; Emery and Koops 2022; Sienicka et al. 2022; Malicka, Mynarska, and Świderska 2021; Naya, Saxbe, and Dunton 2021; Zimmerman et al. 2022). Some preliminary research has been done on fertility intentions in Canada as impacted by COVID-19, but it was described broad fertility demographics and predicted economic impacts that may arise from these pandemic-related fertility changes and did not address the social dynamics I have explored within this thesis (Fostik 2021).

Considering the extant literature and research surrounding queer reproductive decision-making during the COVID-19 pandemic in the city of Saskatoon in Canada, I feel that my research is well placed. There is a varied foundation of research examining the general categories I have reviewed, helping to prevent me from unnecessarily repeating already-answered questions and informing my thoughts and directions in my work. In addition, my research is unique and precise enough to ask questions that have not yet been asked through my focus on queer people in Saskatoon, my inclusion of the COVID-19 criteria, and my unique context that guides the work I have done. Thus, my thesis joins an important and diverse body of research about the social study of reproductive decision-making that also addresses a specific place and set of concerns that have not been previously explored.

# 1.4 - Theoretical Perspectives

The theories underpinning this research begin with my training and education as an anthropologist. After seven years of undergraduate and graduate training in medical anthropology and numerous social science experiences as a teaching and research assistant, I think in terms of social science and anthropological theory. I look for the cultural context in everything. I think in terms of cultural relativism alongside analytical judgments. I have been biased to think of people not in terms of their philosophical orientation, their financial position, or the health and balance of their body, but instead in terms of relational dynamics — how they relate to each other as individuals and as increasingly-larger social units (Dilley 2002; Brown 2008; Faier and Rofel 2014). I must stress that these orientations are not solely the purview of anthropology or even the social sciences (Rotman 2021). Rather than try to claim anthropological ownership of these ways of thinking, I bring them up only to illustrate that the theoretical orientation for this research goes beyond the specific theories I have chosen to

underpin this work and has instead emerged from the way I have been trained to see and hear and think.

Within the general anthropological theory, I have chosen a pastiche of two specific theoretical lenses through which I have viewed this research. The first and broadest is that of queer theory. As a theoretical orientation, queer theory is generally flexible, often vague, and therefore difficult to define. For use in this thesis, I appreciate Mark Graham's definition of queer theory, that it is "a critical inquiry into the alignments of sex, gender and desire that are in the service of normative forms of heterosexuality, the heteronormative, that saturates the social and cultural order" (2016). Graham has also borrowed from Steven Seidman's discussions of queer theory, who said that the theory is really the

study of those knowledges and social practices that organise "society" as a whole by sexualising – heterosexualising or homosexualising – bodies, desires, acts, identities, social relations, knowledges, culture, and social institutions" (1997).

I appreciate Graham's approach that labels queer theory a critical inquiry because Seidman's labelling of queer theory as a study does not go far enough to detail how queer theory can be used in an active and applied context. To take Graham's critical inquiry approach one step further, I also include some of Eliza Garwood's discussion of queer theory because it operationalizes my use of queer theory in this thesis, as I will use it to "deconstruct the normative notions of and established binaries surrounding 'male/female, masculine/feminine, heterosexual/homosexual', 'real families/pseudo families' to name a few" (Garwood 2016). Queer theory is thus a way to bring to light the powerful and frequently assumptive forces which code many social and cultural phenomena in binaries, like right or wrong, good or bad, and desirable or repulsive, through the lenses of gender and sexuality.

Within this research, I apply queer theory to the stories shared with me through the notion of queering, a shortened version of "queer reading," which is an analytical tool used to look beyond the surface of something to see and challenge implicit, and sometimes explicit, heteronormativity (Barker 2016). I am not alone in queering anthropology in general, and reproductive anthropology more precisely, because queering and queer theory are fantastic tools for revealing the taken-for-granteds, the assumptions, and the unspoken expectations that frequently guide cultural phenomena like reproduction (Hekma 2000; Boellstorff 2007; Wilson 2019; Craven 2019; Valdez and Deomampo 2019). Since anthropology is, in many ways,

focused on estranging the familiar and understanding the tacit forces that guide cultures, queering and queer theory are handy tools, and they have greatly informed the work in this thesis.

Garwood's emphasis on using queer theory to deconstruct normative notions is an especially salient aspect of this thesis. Three theoretical aspects of normativity are worth deconstructing in the light of queer reproductive decision-making. Understanding normativity is the first step, as norms are "cultural phenomena that prescribe and proscribe behaviour in specific circumstances" (Hechter and Opp 2001, xi). Everyone experiences the impacts of normativity in some way, but some forms of norms are focused on specific areas of behaviours.

Heteronormativity is the basis for the norms of note in this section, and heteronormativity is "a hegemonic system of norms, discourses, and practices that constructs heterosexuality as natural and superior to all other expressions of sexuality" (Robinson 2016, 1). Heteronormativity sets the expectation that heterosexuality is the default state of being and that general behaviours ought to align with (socially acceptable) expressions of heterosexuality. If these expectations are not met, there may be consequences – subpar or non-existent healthcare, restricted access to forms of bureaucratic support or help, ostracization, or even many different forms of violence, to name but a few (Beagan et al. 2022; Enson 2015; Portillo, Humphrey, and Bearfield 2022; Avramopoulou 2022; Gentry 2020; Javaid 2018). Heteronormativity can be ephemeral and difficult to attribute to any one source, but its ubiquity can mean that everyone, including queer people, may encounter pressure to adhere to hetero norms on a regular basis, including policing and surveilling their behaviours in an effort to ensure compliance (Cumming-Potvin and Martino 2014; Manning and Stern 2018).

Homonormativity is a different take on this norm, but one that comes from much the same place. Over the decades of the queer civil rights movements and as mainstream acceptance of queer people, or more accurately middle-upper-class gay White cis-gendered men, gained traction in North America, it became expected that non-straight people assimilate into heterosexual society by adopting much of the same priorities and expectations as heteronormativity. The term "homonormativity" was publicized by Lisa Duggan but the definition I am using comes from Matthew Archer, who said "homonormativity refers to an idealized performance of queerness that is congruent with as many privileged cisheterosexual norms as possible and which offers the least challenge to dominant narratives" (Archer 2021,

284). By sending out the message that gay rights have been established and that queer people should be happy productive citizens who focus on their individual domestic lives and perform them the same way as their acceptable straight peers, an expectation is created that queer people contribute to society in exactly the same way that straight people do. In application to the exploration of reproductive decision-making, homonormativity borrows reproductive goals from heteronormativity, and those goals are about creating families with children and two parents, ideally with some biological relation between at least one of the parents and the children (Garwood 2016). This leads us to the third important type of normativity for this thesis, that of reproduction.

Repronormativity is the term used to describe norms around reproduction that have arisen from heteronormativity and is defined by Anna Weismann as "legitimized, state-sanctioned heteronormative acts of reproduction specifically through the patriarchal heteronormative family" (2017, 279). Heteronormativity champions creating children through 'heterosexual' intercourse, or more technically, a penis introducing sperm to egg(s) within a uterus. While this type of reproduction is definitely the most common, holding it up as a moral and cultural standard because of its association with heterosexuality is repronormativity. Repronormativity is as much about gender identities and sexual identities as it is about the practicalities of creating a new human being, such as reinforcing ideas about acceptable masculinity or feminitity, about appropriate types of bodies for fertilizing or bearing children, and who is recognized as being appropriately female or appropriately male under institutional definitions (Mavuso 2022; Love 2022; Karaian 2013). Prioritizing and privileging repronormativity can negatively impact anyone, including straight people, but doing so makes navigating reproduction even more difficult for queer people because some of us cannot or will not engage with the types of reproduction that are considered normative. This thesis considers these three types of normativity (hetero-, homo-, and repro-normativity) in particular for two reasons. One is because contrasting and contesting normativity is a central aspect of queer theory, and the other is because my analysis showed that these concepts weigh heavily on the queer people with whom I spoke and the meanings they derived from the reproductive decisions with which they grappled.

The second theoretical approach is much more specific but it applies to this thesis just as broadly as the first, and that is Cheryl Mattingly's approach to critical phenomenology, what she calls "critical phenomenology 2.0" (Mattingly 2019). Critically examining the concepts,

categories, and classifications anthropologists regularly use to understand the phenomena they are seeing is at the heart of Mattingly's approach because critical phenomenology 2.0 is "an experience-near process of concept destabilization" (Mattingly 2019). As anthropology continues its tasks of understanding and illuminating, the profession grows and changes but also solidifies, taking for granted some of the foundational frameworks that have become key – frameworks such as relationships of power and the violence enacted through structures and institutions. To look at individual experiences and divorce them from these dominant frameworks provides a chance to defrost and destabilize the "canonical" critiques that have long underpinned anthropological works, such as those at the heart of what Sherry Ortner calls "dark anthropology" (Ortner 2016). Within this version of critical phenomenology, Mattingly hopes to "awaken [anthropology's] critical gaze" by disrupting established categories and frameworks to keep them "open to scrutiny" (Mattingly 2019). Letting experiences and meanings be strange, confusing, and difficult to categorize opens space for new and critical ideas to emerge within academic thinking and research.

In explaining how I have applied Mattingly's intriguing views on thinking and meaning to my thesis, I must first state that I have done so inexpertly. The depth and potential contained within Mattingly's theoretical approach is one that I feel I will wrestle with throughout my career, and only through practice can I hope to come close to applying the critical approach she calls for. Embracing this theoretical approach has helped me on my journey of analyzing my assumptions around how things work and looking closer at the individual meanings and thoughts that others share with me and that I experience as a researcher. I thus apply Mattingly's theory here not to claim any kind of academic triumph but merely to recognize the beginning of what must be a life-long journey of awareness and reflection on what I know and how I have come to know it.

Mattingly's critical phenomenology was extremely useful in my exploration of queer reproductive decision-making in Saskatoon during the COVID-19 pandemic. Queer people already defrost and destabilize what we think we know – the queer landscape of identities and terminology is constantly evolving, and queer peoples' experiences of queerness tend to shift and change over time (Kirsch 2000; S. D. Miller, Taylor, and Rupp 2016). Who and what is queer and how queerness exists within a broader cultural milieu can surprise, startle, and even cause upset because of how little queerness can conform to our expectations and assumptions. Queer is

weird, queer is different, queer is strange (Dator 2022; Valentine 2002; Alexander 2017). Also, depending on your viewpoint, queer is normal, queer is stable, and queer is average (Warner 2000; A. I. Green 2007; J. Ward 2016). Queerness often resists categorization as much as it embraces it, and queer identities and their relationships with the world are often sources of contention within the queer community as much as without (Peters 2001; Phillips 2009; Drushel 2019). Queerness is, therefore, a perfect area in which to practice critical phenomenology 2.0.

Both parenting and COVID-19 also hold potential for critical phenomenology. As Western definitions of parenthood move away from the norms of the nuclear family, the definition of a family and how it is put together actively resists the categories of kinship and relationality often employed by anthropological analysis (Carsten 2004; Herman 2009; Otto 2017). What family means and how it is experienced is a subject ripe for the critical gaze of Mattingly's phenomenology 2.0 and is an area I have tried to explore in this thesis. There is a similarity between the variable meanings and experiences of family and the broad scope of experiences and meanings that have emerged in the COVID-19 pandemic. Individual variation in the effects of the viruses that cause COVID-19, in how people make meaning of the social response to these viruses, and how important experiences are understood in the face of these responses add up to turn a "global pandemic" into a highly individualized, difficult-to-categorize, amorphous blob of stuff that we still refer to with the broad label of "COVID-19" (Pereira et al. 2021; Joy et al. 2020; Sandbakken and Moss 2021; Karataş and Tagay 2021). How, then, does anthropological research wrestle with analyzing parenthood and COVID-19 experiences without resorting to "sedimented" concepts that might misrepresent or misunderstand the subjects with which they work? This thesis is my attempt to do so within the particulars of my setting and research question.

### 1.5 - Research Question and Thesis Layout

I have thus outlined the important aspects of my research - queer reproductive decision-making during COVID-19 in Saskatoon. In addition, my theoretical approach hints at what specifically I am looking for within that particular setting. These factors add up to the research question that has driven my research for the past two years: What meanings do queer people in Saskatoon find in their reproductive decision-making processes, and how do those meanings influence those processes during the COVID-19 pandemic?

Chapter two outlines the methodology I used within this thesis, including recruitment, data collection, analysis, an outline of those who participated in this research, and a discussion about the impact of COVID-19 on my methodology. Chapter three examines queer family structures and how queer people make meaning in the creation of their families, with a focus on making families that work for them. This chapter dives into the social primacy of the nuclear family and how it both does and does not include queer families, how queer people view and wrestle with the gendered aspects of pregnancy, and the unique impacts COVID-19 has had on both encouraging and discouraging queer family formation. Chapter four is an examination of the role of safety in queer reproductive decision-making, starting with a critical definition of safety as related to reproductive decision-making and then diving into the disparities and similarities of safety as experienced in Saskatoon and as experienced in the places that people came from before moving to Saskatoon. Chapter five describes the impact of normativity on how queer people in Saskatoon make their reproductive decisions, including a critical definition of norms, exploring the norms that seem to originate in Saskatoon and discussing how norms from other countries have impacted the people who have since moved to Saskatoon and are making their reproductive decisions here. Chapter six focuses on the COVID-19 pandemic and how it has impacted 'place', starting with a critical definition of place, describing how COVID-19 has changed what meanings people associate with place, how the space between places has gained new meanings and then inspecting the impact that the COVID-19 pandemic has had on reproductive decision-making specifically. Finally, in chapter seven, I conclude by discussing this research's implications and suggest avenues of further inquiry.

The information I provide and consider in these chapters is intended to support my answer to the research question posed above, which is that queer people in Saskatoon are influenced in their reproductive decision-making by meanings they have created around: queer and reproductive safety; the family structures they feel are best and are supported in making; and how COVID has impacted meanings of place.

# <u>2.1 - Unpredictable Changes – The Impact of COVID-19 on Methodology</u>

Data collection for this research happened between May 4<sup>th</sup> and June 24<sup>th</sup>, 2021, with planning and proposals finalized by March of that same year. The COVID-19 pandemic had been in full swing for a year by the time the proposal process was completed, with the University of Saskatchewan and the University of Regina transitioning to full remote operation on March 13th, 2020 and the province of Saskatchewan declaring a state of emergency on March 18th (Fontaine 2020; Government of Saskatchewan 2020). Arguably, I had a full year of the pandemic to come to terms with the restrictions it would place on my research, but the constantly shifting variables of vaccinations, changing restrictions and gathering sizes, and the weather made planning the data collection process difficult at best. As a result, I abandoned the methodology I had hoped to use, that of traditional ethnography in the sense of participant observation and significant time spent in the contexts experienced by participants, and found myself instead engaging with the unfamiliar ethnographic possibilities found in remote interviewing and the observations and insights I could glean through a Zoom session. Though I was not conscious of this fact during my interviews, fitting this new version of ethnography into my research was, in some ways, an exercise in queering ethnography. I had had no prior experience with traditional ethnography but I had been prepared for it and had expectations set around what data collection within this anthropologic mainstay would be like.

This new type of ethnography, though limited in some of the ways I have mentioned, also provided new opportunities for exploring meaning and milieu and for experiencing the world through the borrowed perspective of those sharing their stories with me. In these ways, I was looking at my research "queerly", i.e. strangely and without a normative sense of familiarity, and I was not alone (Dasgupta 2021). I am not the only one who has experienced these methodological fluctuations and challenges in trying to understand how ethnography fits into an ever-changing world, with some calling these new approaches "patchwork ethnography" (Fine and Abramson 2020; Günel, Varma, and Watanabe 2020; Podjed 2021; Abad Espinoza 2022). At the time of data collection I was making the best of a research trajectory I had not fully anticipated and with which I was not familiar, but with the gift of hindsight I can appreciate the

opportunities for new ideas and analyses presented by my engagement with patchwork ethnography. In this chapter, I outline the steps I have taken throughout this patchwork research.

# 2.2 - Recruitment

As social distancing mandates and other restrictions persisted, and remote school and work remained prevalent, my recruitment methods were entirely digital and remote. I engaged in three main avenues of recruitment. First, I posted to the University of Saskatchewan's internal events feed on the PAWS Web Portal (see appendix A – Recruitment). Second, I posted a recruitment request in multiple places on Facebook. I posted it to my own personal Facebook page to reach existing queer contacts. From there, friends and peers circulated and shared it on their own Facebook pages. I also asked OUTSaskatoon, the city's largest queer support and service organization, to share my post to reach people beyond my social circle. Third, and though this was not intentional, I received many contacts through snowball recruitment as participants shared their experiences with their friends, who in turn reached out to me, and as people who had seen the PAWS announcement but had not gotten involved in my research spread the word to others who they thought might be interested.

One aspect of recruitment shifted during the research process. The initial age range of participants was 18-35 years old, but during recruitment, I found it difficult to speak to any queer participants in that age range who had already had children. To hear the perspectives of people who experienced parenthood and how that impacted their reproductive decision-making processes, I expanded the upper age range from 35 to 45 years old.

# 2.3 - Individual Interviews

My data collection consisted of nineteen semi-structured and person-centred interviews. All interviews were one-on-one and were held over Zoom. Interviews ranged between forty-five minutes to two hours. During these interviews, I asked about four main topics – their lives and their contextual lived experience (the "get to know you" questions), how they feel about and have experienced the province of Saskatchewan, how the COVID-19 pandemic has impacted their lives, and about their thoughts, feelings, and plans regarding parenthood (see Appendix B – Interview Guide).

The remote nature of these interviews presented both opportunities and challenges. Due to the high prevalence of people working and learning from home, rescheduling interviews due to scheduling difficulties was easy, as we did not have to consider commute times or the business hours of potential interviewing locations. It may have also helped establish an interviewing relationship between myself and each participant, as interviewing someone in their home, when done carefully and with respect to them and their space, can help to develop rapport, safety, and ease of self-disclosure (Borbasi et al. 2002). Remote interviews also allowed participants some choice in the interview location if we considered the location to be where the participants physically were when interviewing, not within the online location defined by the boundaries of Zoom's video windows. Their choices in interview locations may have imbued the interviewing process with more agency and less inequality than if we had met up in person (Herzog 2012).

The challenges of remote interviewing were also complex, partially because of my expectations of interviewing. I chose the field of anthropology due to my extroverted nature, love of talking to others, and deep and abiding interest in connecting to people as much as possible. Facing the possibility of the culmination of my life's work thus far, the interviews that would form the backbone of my Master's thesis, without the in-person dynamics that I had perceived as the hallmark of anthropological data collection, left me disappointed in the situation. This aligns with John Johnson and Timothy Rowlands' observations about how the interviewing process, and the deep understandings that emerge from in-depth interviews, can be impacted and biased by the interviewer's feelings and connections to the interview itself (Johnson and Rowlands 2012). Interviewing by Zoom fundamentally changed the social interaction of the interview, the interaction I had expected and longed for (Warren 2012). While Zoom interviews did not prevent interaction, they removed aspects of social communication that I had considered fundamental to understanding other people, such as revealing the full range of their body language, sometimes flattening tone of voice through the audio encoding and decoding process, changing the rapportbuilding and process of trust establishment by removing physical proximity, removing some sensory experiences like smell or touch. Zoom interviewing also introduced new challenges, with the most universal one perhaps being the interruptions and delays due to bandwidth and other technical difficulties. All of these experiences, so far removed from the in-person processes I had expected to encounter, rendered a familiar concept, that of a conversation, rather strange and alien (Oliffe et al. 2021).

A year after conducting these interviews, I can now appreciate the perspective I gained from having my expectations subverted by the necessities of interviewing during the peak of COVID-19. By removing my sense of comfort and ease around the interviews, I was able to "estrange the familiar" and gain insights I might otherwise have missed if I had done in-person interviews, which are a known entity to me. Being knocked off-balance, mentally speaking, was a great boon for the next step of my research – analysis.

# 2.4 - Analysis

All research data, including interview notes and field notes, were analyzed through the twin theoretical lenses of queer theory and Mattingly's critical phenomenology 2.0. I had originally intended to use grounded theory by continuously refining my analysis against the data as I collected it, simulating the "continual interplay between data collection and analysis to produce a theory" suggested by Glenn Bowen (2006, 13). While I did engage in preliminary analysis throughout the data collection process by asking different follow-up questions to explore themes I saw emerging in earlier interviews, my analysis process may not have met the threshold for grounded theory. At the very least, my analysis was inductive, meaning that "the patterns, themes, and categories of analysis come from the data; they emerge out of the data rather than being imposed on them prior to data collection and analysis" (Patton 1980, 306).

Pre-analysis began by immersing myself in my data, including field notes and interview transcripts, reading each through multiple times. Field notes played an especially important role in this part of the analysis process, as my field notes were essentially aide-mémoires about the non-verbal aspects of the data collection experience. I had audio-recorded the interviews I conducted but did not video-record them, so my field notes mentioned facial expressions, the visible physical surroundings of the interlocutors, non-verbal reactions, and notes to contextualize sounds I heard that were not made by the interlocutors. While little of these notes were carried directly into the analysis and subsequent thesis, they were essential for re-immersing myself into the interviewing experience during pre-analysis and to thus represent the words of my interlocutors in as contextually accurate a way as possible. After this pre-analysis stage, I engaged in phenomenological analysis, aiming to "explore in detail how participants are making sense of their personal and social world" by analyzing "the meanings particular

experiences, events, states hold for participants" (Smith and Shinebourne 2012, 53). I did this by reading through each transcript, highlighting specific areas of meaning.

I analyzed the data using a phenomenological framework borrowed from Linda Finlay (2014). As such, I attempted to "bracket," or put aside, what I take for granted about queer reproduction so that I could "dwell" on the data, thus making room so that "the phenomenon [could] reveal itself and speak its story into [my] understanding" (Finlay 2014, 126). After immersing myself in the data and the phenomenon of queer reproduction in Saskatchewan as told by the people I interviewed, I set out to distill the meaning contained within the phenomenon, or "discovering interrelated constituents of the phenomenon, then searching for connections across the data set and clustering any general or essential meanings that surface" (129). To do so, I followed the example of Paul Flowers and his colleagues (1997) by using thematic analysis to reveal phenomenological meaning within the emerging themes. This analytical approach aligns with the theoretical basis of critical phenomenology as espoused by Cheryl Mattingly, who sees the task of critical phenomenology as "an experience-near process of concept destabilization" (2019, 416). Finlay argues that phenomenological analysis includes a need to "break away from our own "natural attitude" and find a way to remain open to new understandings" (2014, 122).

I will note that I did not account for socioeconomic status (SES) in my analysis. Very few interlocutors gave me answers that related to their SES and, as my analysis was inductive, I did not find much relationship between SES and the themes I discuss in this thesis. The majority of interlocutors were university students and had thus attained a relatively similar baseline SES position, both due to their relatively young ages and to the financial necessities of attending university. This is not to imply that all interlocutors were financially privileged or had the same level of income, but rather that, through their status as university students, they occupied a similar SES category. This assertion does not cover the unique variables in each person's financial situations but very few of them shared their financial positions with me and I did not include any SES-specific questions.

## 2.5 - Participants

In total, I spoke to nineteen different people. Because Saskatoon is the largest city in Saskatchewan and is culturally distinct from the province's smaller cities, towns, and rural areas, I decided to focus my area of research on Saskatoon. All but one participant was located in

Saskatoon and so I excluded that one participant from my ongoing research and I therefore focused on a pool of eighteen Saskatonian interlocutors. Throughout this thesis I refer to interlocutors by the names they have chosen – some are their given names and others chose pseudonyms.

*Table 2.1 – Interlocutor demographics and parenthood intentions* 

Preferred	Age	Area of	Sexuality	Gender	Children
Name		origin			
Rowan (she	25	Sask.	Bisexual	Cis-	0
and they)				woman/female	
Lula (she	22	Sask.	Bisexual	Non-	0
and they)				binary/"meh"	
Thayara	28	Brazil	Lesbian	Cis-woman	0
(she)					
Claudia	27	Brazil	Bisexual	Non-binary	0
(she)					
Jason (he)	24	Sask.	Bisexual	Cis-man	0
Lisa (she)	21	Sask.	Bisexual	Cis-woman	0
Mariana	26	Mexico	Bisexual	Cis-woman	0
(she)					
Juan (he)	20	Colombia	Gay	Cis-man	0
Morgana	27	Brazil	Bisexual	Non-binary	0
(she)					
Ethan (he)	36	Quebec	Towards	Trans-	4
			women	masculine	
Jane (she)	29	Sask.	Lesbian	Non-binary	0
Elizabeth	22	Sask.	Lesbian	Cis-woman	0
(she)					
Nicole (she)	43	Sask.	Bisexual/queer	Cis-woman	1

James (she	31	Sask.	Bisexual	Non-binary	0
and they)					
Abi (she)	34	Philippines	Lesbian	Cis-woman	0
Carolina	24	Alberta/Brazil	Bisexual	Non-binary	0
(she and					
they)					
Cat (they)	27	Sask.	Pansexual/queer	Non-binary	0
Justin (he)	25	Sask.	Bisexual	Trans-man	0
			w/ grey-ace		
Preferred Na	me	Plans for Future	e Parenthood		
Rowan (she ar	(she and they) Wants to have children				
Lula (she and	(she and they) "it's dependent, it fluctuates, my feelings about it, depending on		pending on		
	the day"				
Thayara (she)		"Some days I do, some days I don't."			
Claudia (she)		Wants to have children			
Jason (he)		Wants to have children (two kids)			
Lisa (she)		Wants to have children			
Mariana (she)		Does not want to have children			
Juan (he)		"So I don't know." (Ambivalent)			
Morgana (she)	)	"I don't feel like this is something I can think of right now cause			
		there's so many things in life to be accomplished."			
Ethan (he)		Done having children			
Jane (she)		Wants to have children			
Elizabeth (she	)	Does not want to have children			
Nicole (she)		Done having children			
James (she and	d they)	Does not want to have children			
Abi (she)		Wants to have children			
Carolina (she	and they)	they) "Yeah, I still don't know where I land on whether that's a decision			
	that I will make in the future."				
		1			

Cat (they)	"I am interested in parenting, and I would love to parent, but I do
	not want my own children."
Justin (he)	"I think I would like to have kids if I knew that I'd be stable
	enough and the world would be stable enough to raise a kid in my
	30s, but I think right now, it's hard to say."

Participants ranged in age from 22-43 years old, with the average age being 27. Only four out of eighteen participants were over the age of 30. This age range encompasses the years that are often thought to be the prime years for reproductive decision-making.

The gender and sexual identities of the participants were quite varied. In describing these identities, I use queer terminology that may be new to the reader, so I have included a glossary of terms as at the beginning of this thesis. Seven participants identified as cis-gendered women, and two as cis-gendered men. Another seven participants identified as non-binary, and two as transmasculine. This variation continued in their sexual orientations; the most common identity was bisexual, as eleven participants claimed bisexuality, though one bisexual participant also mentioned they had grey-asexual tendencies. Four identified as lesbians, with an additional participant saying their sexual orientation was "primarily towards women," sometimes called gynesexuality/gynephilia. Finally, one participant stated he was gay, and one identified as pansexual/queer.

Another notable aspect of the participant demographics was the participants' places of origin. While all were now located in Saskatoon, only ten participants were born or were primarily raised in Saskatchewan. Two participants had grown up in other Canadian provinces, one had immigrated from Southeast Asia, and the remaining five participants were of Central American and South American origins. All participants were fluent in English and all interviews were conducted in English.

Finally, there was a wide spread of responses regarding their current reproductive decision-making. As of our interviews, six people were firmly in favour of becoming parents, while one wanted to be a parent but did not want to have children biologically. Two were already parents and were done having children. Three were not interested in becoming parents. The remaining six participants were ambivalent – some had fluctuating feelings, desiring children some days and being happy in their child-free state on others. Some were unsure because they

felt other priorities needed to be addressed before children became an option. Yet others felt that they might be interested in having children if the proper contexts presented themselves.

# 2.6 - Timeline

An approved ethics amendment to BEH REB-334 was approved in April of 2021. Recruitment for this project began that same month, and the first interview was conducted in early May. Interviews concluded at the end of June, and the analysis was completed by the end of December 2021. The writing and editing process of this thesis was completed between January 2022-March 2023.

# Chapter 3 - Queer Family Structures

# 3.1 – Introduction

Creating and maintaining kinship, family, or close relations with other people has been one of the main dimensions of human evolution and history (Bengtson 2001; Leonetti and Chabot-Hanowell 2011). The specifics of family and how they are built varies widely between cultures, generations, and individuals, but the importance of family in the social networks that keep people afloat is generally primary (Bott Spillius 2013). This chapter centers on the concept of "families that work", an approach that ties into current anthropological approaches to kinship centering on the idea of relatedness. Instead of focusing on kinship structures that outline the specific matrilineal, patrilineal, or bilateral tendencies within cultures that dominated 19<sup>th</sup> and 20<sup>th</sup> century anthropology, the idea of relatedness is much more about what the concept of family means to the people creating those families (Carsten 2000). In discussing relatedness, Janet Carsten points out how the domestic space, including sharing of resources and nurturing, is the most important aspect of family-building (2004). I agree but will expand on this concept, as intersectional approaches to the notion of the domestic space and the sharing of resources among queer people in Saskatoon highlight adaptable definitions of what exactly a domestic space is and how resources are identified and shared. Flexible networks of relatedness and how those networks are used to construct identity and explore the meaning of family or kin is not a unique idea (Holmes 2009). Within that framework, I set out to explore the specifics of how relatedness is envisioned by queer people making families in Saskatoon in their unique cultural contexts, and how that family-making relates to reproductive decision-making in Saskatoon.

Unfortunately, some groups have been restricted or prevented from creating families, or creating families in the ways that work for them, based on a host of intersectional reasons: Indigenous family-making in Canada was severely impacted by colonialism and the systemic implementation of colonial practices such as the Indian Act (Tam, Findlay, and Kohen 2017). Socioeconomic status also changes how and what kinds of families can be formed, as lower access to housing and unemployment are associated with lower levels of family formation or increased interpersonal problems within families (Boyle and Lipman 2002; Mulder 2006; Ariizumi, Hu, and Schirle 2015). In addition, identities that fall outside of what is depicted as the norm (straight White middle/upper-class people) are not equally supported in creating or

maintaining their families and thus must make different reproductive decisions than people who fall within that norm.

In my conversations with queer people in Saskatoon, I found that a 2SLGBTQIA+ identity is also associated with difficulty in creating families that work. This difficulty emerged in three distinct themes. The first is that many of the queer people with whom I spoke did not want to create nuclear families, or "a family unit comprising parents and their dependent children" (Fortunato 2017, 102), and instead discussed exploring family structures and dynamics that align with queer values and realities. The concept of the nuclear family is steeped in a history assuming heterosexual cis-gendered biological reproduction as a state of normalcy (often shortened to "heteronormative"), and while in recent years there has been a shift to include nonheteronormative families under the nuclear family umbrella, the concept still carries heteronormative baggage and is therefore not always embraced by queer people (Peterson 2013; Herz and Johansson 2015). Unfortunately, the people who shared their stories with me also described the significant lack of institutional and social support for creating non-nuclear families, or families that do not center on the two-parents-and-their-children structure often presented as the familial default. The second theme was pregnancy and specifically how many of the queer interlocutors do not want to experience their own pregnancy, though some were open to building families through their partner's pregnancy. Together, we explored how social expectations of gender and sexual identity performance are inherently tied to understandings and expectations of pregnancy and how queer people want to subvert, disrupt, or avoid those understandings and expectations in their reproductive decisions. The third theme dealt with the intersection of the impacts of COVID-19 on creating queer family structures, specifically focusing on how social distance and isolation allowed queer people to explore queer family structures without surveillance from others while also discussing the difficulties COVID-19 presents to marginalized communities like queer people in Saskatoon when it comes to support and family formation.

As part of the reproductive decision-making processes, queer people in Saskatoon are deciding how their families are formed. I have found that an important aspect of those processes is that most people want to make families that work for them. How queer people experience family can differ significantly from the experiences of non-queer people, including a reliance on families of choice and found families when families of origin are not accepting or supportive of

queer people. Queer peoples' experiences with family also include a different social relationship with the meanings of family and reproduction stemming from gender roles and stereotypes about sexual orientation (Bertone and Pallotta-Chiarolli 2014; Dixon 2018). Queer families are, therefore, more complex to make because of a lack of institutional or systemic support for making families that do not reflect the experiences or expectations of the cisgendered and heterosexual patriarchy. Ultimately, for the queer people in Saskatoon who shared their experiences with me, the reproductive decision-making process includes serious consideration of how to make families that work for queer people while lacking the same supports that cisgendered and heterosexual (or 'cishet') families receive.

# 3.2 - Exploding the Nuclear

Ethan is explaining to me the makeup of the family he and his fellow co-parents have built. There are four children in his family, five different parents in the mix, and three different levels of relationships that I am doing my best to follow – who is biologically related, who is maintaining interpersonal relationships, and who lives in which household. While I try to understand these intersections, Ethan reminds me that, regardless of these complexities, the kids are all siblings and are in a family full of love. For Ethan, the specifics of the makeup of their family structure are less important than ensuring that the family is supportive, functioning, and providing the best possible home for all four children. The multiple aspects of family that Ethan has shown me align with anthropological observations on the notion of relatedness which, as Pamela Downe points out, is multifaceted: "[r]elatedness, then, has social and emotional as well as material and behavioural dimensions that must be considered in discussions of family and kin networks" (2021, 28). Constructing family centers on the goal of the best possible home for these families and oftentimes for queer people what is best is not synonymous with the nuclear family. This sentiment is echoed by many participants and perhaps best summarized by Carolina, who says, "I've always preferred the concept of a larger family unit, whether that be my roommates or having an even wider net of support because one of my issues with parenting is I honestly feel like the nuclear family two-parent system just doesn't work."

Though the reasons each participant gave me about why the nuclear family system does not meet their needs were different, Carolina describes a difficult situation her little brother went through as an excellent example of why the nuclear family is not enough.

I think that's partly because I saw how [the nuclear family two-parent system] didn't really work with my family. When my brother was going through his stuff, I saw my parents could not have handled it by themselves, and I think sometimes I wish that they had already had a bit more of a net. In the moment, they really scrambled to create one, but if we'd already had that, I feel like maybe some of the things we went through would have been avoided.

Making decisions about reproduction means balancing a wide variety of variables, and for queer people considering these decisions during the COVID pandemic, some of the most important variables are those of support and family functioning (Shek 2021; Feinberg et al. 2022). These are extraordinarily difficult variables to navigate, as the role of support has changed due to the shifting and difficult experiences of this pandemic, the distance and shift to online methods of socializing, and the changing nature of childcare (Watson, Lupton, and Michael 2021; Stanley et al. 2021; Adams et al. 2021). Family functioning has also been impacted on a broad scale, as the pressure and uncertainty of the COVID-19 pandemic have led to decreased family mental health, increased conflict and dysfunction, leading to lower levels of family cohesion, and higher levels of stress and associated harmful coping mechanisms such as alcohol consumption (Gunawan, Juthamanee, and Aungsuroch 2020; Gadermann et al. 2021; Peltz et al. 2021; Greenglass, Chiacchia, and Fiskenbaum 2022). The COVID-19 pandemic has undoubtedly highlighted the need for community-oriented collective care, but that need has existed for much longer than this particular virus has been around (Evans et al. 2017; Sargsyan 2019; Downe 2021; Jones 2020).

For queer people, having been historically denied the chance to create the typical nuclear family through laws and social norms that long prohibited "gay" marriage, assistance for queer reproduction such as adoption, artificial insemination, or surrogacy, and restricting access to legal protection of family units, making public decisions about their reproduction around the concept of the nuclear family is a relatively new phenomenon (Cox 2005; Cameron 2008; Pettinicchio 2010; Macintosh, Reissing, and Andruff 2010; Harder 2021). Instead, queer families have long made their own connections, their own structures, and their own ways of doing things — in many ways, they have been pioneers of communities of belonging that can go beyond the cis-het idea of families and become supportive relationships based on constructed and deliberate relatedness rather than consanguinity and genetics (Clare 2019; Garwood 2022).

Families of choice have been a common way for queer people to create families that love, accept, and support them by building a family from friends, colleagues, and neighbours instead of blood relations (Weston 1991). As such, families of choice often do not follow the nuclear family concept, including the power dynamics inherent to mother/father/child relationships (Lehr 1999). To be clear, queer nuclear families do exist and are becoming more prevalent now that, in many countries, laws and norms are more open and supportive of queer parenthood (Folgerø 2008). However, some queer people see this shift as a negative rather than a positive, as the nuclear family is seen as an inherently heterosexual social construction that intends to suppress and marginalize lifestyles that do not conform to heteropatriarchal norms (Folgerø 2008; Garwood 2016). Many queer people see the construction of families outside of the nuclear as a matter of culture, carefully cultivated during decades and centuries of oppression and violence (Lehr 1999; Reimann 2001).

Interlocutor Lula provided what they consider to be a salient example of queer family-making and ends up tying it back to how they perceive reproductive decision-making in Saskatchewan:

It's the biggest example of queer straight people, it's one of my straight movies for gay people... Mamma Mia! is, to me, the quintessential heterosexual queer parenting situation. To me, if they were straight straights, they'd be like, oh, paternity test. [...] And we're also going to judge Meryl Streep cause she had sex with three men and she shouldn't have done that and is irresponsible. But in the movie, they're like, we don't care who the biological dad is! All of us are your dad, we all love you, each of us will be your father. [..] Everyone in that movie is theoretically heterosexual, but to me that's not a straight parenting situation cause they're like, yeah, you have three dads and we're chill with that. [..] That to me really encompasses what could be, but... I'm tying it back to the question, but that's not Saskatchewan. In Saskatchewan, it would be like, paternity test and we shame the mother.

In their description of the movie, Lula highlights how closely tied the notion of the nuclear family is to straight people by pointing out that this flexible approach is "not a straight parenting situation," even though the characters she describes are never shown in non-heterosexual situations. For many of the queer people I spoke with who were born or raised in Saskatchewan, this flexible parenting approach is seen as an ideal but is not terribly visible. By placing

themselves in Meryl Streep's shoes within the movie, Lula hypothesizes that if a similar parentage experience were to happen in Saskatchewan, the social and cultural dynamics of the province would not be open to creating the queer (i.e. strange, different) family whose dynamics are not based on biological relationships and the policing of the sexualities of people designated female at birth (DFAB), a term used to describe how people have been gendered in their childhood, which may or may not be related to how they are currently choosing to gender themselves. "Slut shaming," or the denigration of DFAB sexuality, is a prevalent concern in the province, and though Saskatchewan's systems and institutions are slowly becoming more open to families not based on genetic ties, the province still relies heavily on paternity testing and genetic relationships in determining family creation (Umereweneza et al. 2020; Wiegers 2010). While it is unlikely that the forces that make a Saskatchewanian *Mamma Mia!* an improbable prospect are intentionally aimed at preventing or shaming the creation of queer families, the outcome is still dissuading queer people from creating the families they think are best as part of their reproductive decision-making processes.

Another interlocutor also demonstrates the importance of flexible parenting practices to queer people in Saskatoon. Lisa, who grew up in a small town, only discovered the type of flexible parenting Lula pointed out when Lisa moved to Saskatoon for university. This knowledge changed Lisa's approach to reproduction. When she was younger, Lisa was sure she would never want children, but upon moving away from her hometown and attending post-secondary education, Lisa said,

Meeting so many diverse people and learning that there's so many ways to parent. [...] It doesn't look just how I thought it had to look, with this very nuclear heterosexual family where... Even though I knew that that's not what every family looked like, but it's different when you just know it but you still kind of have it internalized. But now it's definitely like, you can have kids and make that look any way that you want to, which I think is a lot more appealing.

Lisa's discussion of the internalization of the nuclear family norm highlights how deeply engrained these cultural messages about expectations of reproductive decisions are. Before she was exposed to ways of parenting that did not resemble the nuclear family model, she did not realize that there could be a different way of doing things. Through cultural narratives around reproduction, the nuclear family is often touted as the best, if not only, option for creating

families in many areas of the world (Fineman 1993; Pender 2007; Kradin 2009; Gachago, Clowes, and Condy 2018).

In Canada, even though the cultural narratives on queer reproduction are shifting, queer people like Lisa are sometimes not presented with the knowledge needed to make their families in any other way but nuclear, as many of the systems and institutions surrounding families are still based in the idea that the nuclear family is the norm (Ontario Human Rights Commission 2006; Harder 2011). Information about which reproductive decisions are possible, including the values, attitudes, and subjective norms to which a person is exposed, form core features of the reproductive decision-making process (Philipov, Liefbroer, and Klobas 2015). While queer people are almost always making the reproductive decisions that suit their current circumstances, it is important to note that many, like Lisa, may be making different decisions if they knew about the possibility of forming families that do not conform to the nuclear ideal.

As I have mentioned, the systemic and institutional influences on creating the nuclear family norm in Saskatchewan are strong. Interactions between the government, the provincial healthcare system, and the legal system all enforce the creation of specific types of families. The cultural orientation of appropriate types of Saskatchewan families, particularly White settler families, is partially controlled through forced and coerced sterilization of people deemed "unacceptable" parents, largely Indigenous and other marginalized Canadians (Clarke 2021; Akbari 2021). By making other peoples' reproductive decisions for them and stripping them of their agency, doctors and social support personnel dictate who is allowed to be parents and how many children they are supposed to have, done in an attempt "to control [Indigenous] bodies while criminalizing Indigenous health and reproductive practices" (Standing Senate Committee on Human Rights 2021, 22).

In addition, other marginalized people like "racialized persons, persons with disabilities, persons with HIV, institutionalized persons and persons whose sex and gender do not conform to gender-based expectations of society" are frequently targeted for sterilization and control over reproductive practices (Standing Senate Committee on Human Rights 2021, 24). Queer people are heavily present in many of these groups. Even though queer people are listed as their own subgroup within the Senate report that I reference, queer people are also more likely to be disabled than their cishet counterparts, are more likely to have been diagnosed with HIV, and are more likely to be institutionalized for mental health concerns (Fredriksen-Goldsen et al. 2017;

Centers for Disease Control and Prevention 2021; Veltman and Chaimowitz 2014). Even for those who are not queer, many of the groups mentioned are people who have had to build families of choice because their marginalization has either prevented them from engaging with typical nuclear family structures and/or because they have their own cultures and traditions surrounding family that does not match the social norms of nuclear families. Structural and institutional support in creating the right kinds of families for the people making them, whether or not that is a nuclear-style family, is an important aspect of reproductive decision-making in Saskatoon.

Structural family support also aligns with the primacy of nuclear families until it is actively challenged. Nicole shared with me her and her husband's struggle to overcome these structural assumptions about heterosexual nuclear families. Nicole's husband Jai is a transgender man, so when the couple decided to have children, Nicole became pregnant using the sperm of a known donor. As a result, when their daughter Alice was born, Jai was deemed ineligible to be listed as Alice's father on the birth certificate, instead requiring that the sperm donor be listed. Choosing to fight to have Jai legally recognized as Alice's father, the couple spent nearly \$20,000 and an untold number of hours working with their MLA to create the Children's Law Act 2019, also known as Alice's Law. As a result, parents no longer have to go to court to get a declaration of parentage, and verbiage in the law changed from "mother" and "father" to genderneutral "parents" (Djuric 2021; Docherty 2020). While it is commendable that the Saskatchewan government was responsive to the concerns brought forward by Nicole and Jai, the costs of time, effort, and money that were necessary to amend the government's definition of family demonstrate that the biological relationships inherent to nuclear families are still given primacy unless directly challenged. Alice's Law also broadens the definition of nuclear families to include more diversity in creating them, but it does not disrupt the concept of the nuclear family itself.

All over the world, people want to build families in the ways that work for them. Family and kin are built systems of relationships and connections that vary depending on place, time, cultural orientation, and a vast host of other factors (Levine 2008; Read, Fischer, and Hlaing) 2014; Roy and Thompson 2019). In Saskatoon, queer people want to build the families that work for them - families that may run counter to norms and expectations that emerge from Saskatchewan's cultural and religious orientations. As a result, some queer people struggle to

build their families without the social and institutional support offered to their cisheteronormative counterparts because they are not creating a nuclear family. Even though the nuclear family no longer represents the majority of Canadian families and may never have accurately depicted the family make-up in Canada, it remains a cultural and spiritual ideal supported by systemic and legal approaches to family and kinship in the country (Scoffield 2012; Campion-Smith 2012; Statistics Canada 2017). As a result, queer people in Saskatoon seek to explode the nuclear by establishing families that do not conform to these nuclear expectations.

## 3.3 - Pregnancy

Sixteen of the eighteen participants were designated female at birth (DFAB). Regardless of their current gender identities, they were born with anatomy commonly associated with females – vulvas, clitorises, ovaries, uteruses, and fallopian tubes, and someone recorded the child's sex as "female" on their birth certificate (Harrington 2016). These anatomical features are socially and culturally interpreted as gender-specific, so having these parts means that the person is generally seen as a girl or woman (Hubbard 1996). In heteropatriarchal mainstream North American cultural narratives, those anatomical features, and the "women" who have them, are to fill specific cultural roles – namely, sexual pleasure for men and reproduction through creating more humans (Welldon 1988; Musial 2014). The heterosexual assumptions surrounding the association between DFAB peoples' anatomy and reproduction are stark. As part of the heterosexual imaginary, people who have "female" reproductive organs are expected to become pregnant and reproduce (Ingraham 1997).

Regarding reproduction, the assignation of gender roles crystallizes for DFAB people as pregnancy is a "public symbol of femininity and also, however illusory, heterosexuality" (Goldberg 2010, 58). Pregnancy and motherhood are rendered highly visible and are thus objects of frequent and intensive surveillance by the societies in which pregnant and mothering people live (Neiterman 2013; Fox and Neiterman 2015; Johnson 2018). Queer people also experience surveillance, including surveillance of their bodies as gendered and sexual subjects through complex mechanisms of power (Dubrofsky and Magnet 2015; van der Meulen and Heynen 2016) because "non-normative gender and sexual identities disturb the norms in the dominant material, discursive, and symbolic logics of surveillance systems" (Ball et al. 2009, as cited in Kafer and Grinberg 2019, 592). The cumulative effects of being both queer and pregnant could

result in heightened surveillance (Hudak 2019) and was a significant consideration in the reproductive decisions made by the people who shared their experiences with me.

Lula, a DFAB person whose gender identity is "just vibes, I don't care" and who goes by she/they pronouns, already felt surveilled by the gaze of others due to their body being coded as female and saw pregnancy as something that would increase that surveillance. In considering the circumstances in which they would be comfortable being pregnant, they said:

If I could somehow grow a child without having a baby bump, I probably would be chill with it, but it's the visible... And I think that gets into the performance of gender... But being perceived that much as a woman would not vibe with me, I'd be like, mm, could you not perceive me? And particularly, don't perceive me as a woman.

This is because, they said, "we have termed pregnancy into one of the most womanly things you can do and it's so hyperfeminized." In these two statements, Lula is referring to the tension they feel between their individual body image, or the "collective and idiosyncratic representations an individual entertains about the body in its relationship to the environment," and how their body as an embodied social symbol, or using their body as "a cognitive map to represent other natural, supernatural, social, and even spatial relation" (Scheper-Hughes and Lock 1987, 16, 19). Lula's personal experience in their body is one of ambivalence towards gendered assignments based on shape and presence of body parts, saying, "I know that my gender identity is queer but I don't really care enough to label it or lock it down or anything. It is what it is – I don't really worry about it". Yet, by living within a culture that assigns gender based on those body parts, Lula's body and its potential to become visibly pregnant become part of the social body that turns a rounded abdomen on someone who is assigned a (sometimes unwanted) female gender identity into a symbol for "self-sacrifice, hyper femininity, and heterosexually defined roles" (Goldberg 2010, 58). In making reproductive decisions, Lula is hyper aware of the socially gendered impacts of a potential pregnancy and is turned off by the idea of pregnancy because of how they would be perceived by others.

Lula was not the only DFAB interlocutor to respond this way about pregnancy. Some 65% of the DFAB people with whom I spoke told me they did not want to be pregnant and disliked the idea of being pregnant. For many of them, it was for similar reasons that Lula described – being visibly pregnant and coded as feminine was an unacceptable outcome. For some, the idea caused feelings of gender dysphoria, which is the distress or discomfort someone experiences

when their physical body and their gender identity do not align (Atkinson and Russell 2015). For others, it was disgust for, or fear of, the idea of pregnancy, including the difficulties associated with birth and the physical changes a pregnant body will undergo, with some descriptions of these feelings verging on tokophobia or an unreasoning fear of childbirth (Jomeen et al. 2021). These terms have specific clinical meanings and are frequently used to diagnose people who suffer severe effects from them, but among the queer community, both have been adopted into common use and do not specifically refer to more limited clinical definitions of the term (Demšar et al. 2018; J. Moore 2021). Instead, they are used to communicate the queer-specific feelings these terms describe (discomfort, distress, fear, disgust) about the embodied aspects of reproduction and gender.

Caitlin, another DFAB non-binary person, described this relationship with their body, saying:

I am interested in parenting, and I would love to parent, but I do not want my own children. I do not want to be pregnant at all. That's not something that I don't think I would ever actually be able to do and maintain a healthy brain. It would just make me feel too weird, I already feel real weird about lots of the parts of my body and it not fitting with what I think it should be in my brain.

Caitlin's experiences align with the broader queer experience. Dysphoria is often at the heart of these "weird" feelings Caitlin describes, as the assignation of assumed gender and sexual orientation that often surrounds pregnancy and the institutions involved in pregnancy can trigger similar feelings of disquiet and discomfort in people who do not conform to their designated gender at birth (Fischer 2021; Croll, Sanapo, and Bourjeily 2022). Dysphoria among queer people is also associated with lower levels of mental health and is often associated with mental health problems like depression and anxiety (Dhejne et al. 2016; Byne et al. 2018). Aside from dysphoria, queer people are more likely to experience a fear of childbirth in part because of a correlation between queer pregnancy and mental health concerns, because of a medical system that often traumatizes queer people because of heteronormative assumptions and cultures in birthing practices, and because systemic and institutional supports are often inappropriate for queer people (Ross et al. 2008; 2012; Abelsohn, Epstein, and Ross 2013; Carpenter 2021; Hallström et al. 2022). With these intersecting and complex factors, fear of or aversion to

pregnancy among queer people seriously impacts their reproductive decision-making in Saskatoon.

These aversions to pregnancy do not represent the experiences of all those who shared their stories with me. Three people had either already experienced pregnancy and were happy to have their resulting children, or were open to the possibility of pregnancy in the future, but they were decidedly in the minority among those who spoke with me. However, because pregnancy and its complications emerged as an important topic from my analysis and were not the focus of my questions, I did not directly ask the participants about their pregnancy experiences or expectations and they did not share with me stories about their positive feelings about pregnancy. The two participants who have already had children were clearly happy with their families and loved their children, but neither told me much about their pregnancy experiences or the meanings those experiences had. Both had had at least some of their children through conception by donated sperm, so their pregnancies were intentional and sought after but I did not receive much information about their pregnancy experiences beyond that. The pressing issues these parents shared with me were not related to pregnancy but instead surrounded the legal and structural difficulties that they experienced with having their families appropriately recognized by the institutions that govern aspects of those relationships. Thus it seems that pregnancy can be a difficult concept for many queer people to navigate but, when pregnancy is desired and sought after, the complexities surrounding queer parenthood then shift to parental relationships with institutions instead.

#### 3.4 - COVID-19

Because Canada's settler society generally depicts families as mom + dad + kids living in a house through advertising, news media, social programs, and entertainment media, there is an implied social norm that "everyone participates in identical sorts of kinship relations and subscribes to one universally agreed-upon definition of family" (Weston 1991, 22). For those who do not fit into that definition, creating families that work, often based on families of choice, is an equally acceptable way to reproduce and love the people around them. And creating families that work has never been easy for queer people, but it is significantly more difficult with the COVID-19 pandemic. The social disruption caused by social distancing and other public health protective measures has deeply impacted whole family systems and the individual

relationships between family members, presenting challenges for pre-existing families (Prime, Wade, and Browne 2020). Yet these measures have also provided the fodder for a new take on families of choice with the suggestions of "bubbles," or the small group of individuals who can safely socialize with each other because they have reduced their contact with others (Trnka and Graham Davies 2021).

For Carolina, the COVID-19 pandemic played a substantial role in creating their family of choice, impacting their thoughts about reproductive decision-making. They had already considered their best friend/roommate to be their family, saying:

We've been living together for six years, we raised our cats together, it's very much one of those situations where... I was telling a friend recently that I think people confuse us for being in a romantic relationship a lot but we just have a really strong platonic relationship.

This happy and supportive relationship evolved with the addition of a third person, the romantic partner of Carolina's best friend/roommate, as a result of the first wave of COVID-19 restrictions.

[Best friend/roommate's] partner ended up moving to Saskatoon in November of 2019, October maybe, so there were three or four months there where my friend and their partner were really close, obviously, and [...] every once in a while I'd hang out with them, and then COVID hit. And my friend's partner was living across the street from us, and was living alone. So we kind of just ended up creating this three-person household and I think, because of the pressures of social distancing from everyone else, it became really easy for us to create this sort of family unit when everything else about the pressure of society and the way everything else is supposed to work was kind of just removed all of a sudden. So it was really easy to be like, we're just gonna create this little family.

As Carolina pointed out, the social disruptions inherent to COVID-19 responses changed existing systems of sociality and relationality in Saskatoon, paving the way for the quick formation of new family units within "bubbles." This three-person family has become so central to Carolina's concept of the future that they play a massive role in how they think about their reproductive decisions, saying, "So I think if I imagine what parenting would be like in the future, it would probably be... They [roommates] would definitely be involved, I would be the

more central figure or the person putting down my name on the documents." But for Carolina, this particular family of choice is just the centre of a more comprehensive network of support that would be important in having a child, as they went on to say, "And I don't think it would be just the three of us, either. I would still want to feel like there was a bit more support, yeah." This wider support network is important to Carolina because "I don't think anyone should have sole or even divided between two responsibilities for another human life. That's a lot, that's scary!" The COVID-19 restriction measures certainly reinforced Carolina's perspective, as the removal of childcare options outside the home demonstrated that it is difficult for households to cover all of their needs, such as childcare, without the help of outside resources, leading to significantly increased burdens for many families (Cheng et al. 2021).

For others, reproductive decision-making changes the meaning of family structures concerning the state of the world, as highlighted by the COVID-19 pandemic. For many of the people who shared their stories with me, the struggles that all families faced during COVID-19, including the shutdown of childcare systems, the sudden transition to remote school and work, and the isolation of families and its impacts on mental health, brought a sharp focus to the less-than-glamorous realities of having children. Moreover, knowing that pandemic restrictions had such a sharp impact on everyone, especially families, led some people to wonder if they would want to have children because of the certainty of future pandemics and the possibility of necessarily harsh restrictions.

For some participants, the uncertainty surrounding how functional family structures can be made during the COVID-19 pandemic centered on the provincial government's response to the COVID-19 crisis. For Justin, who goes by he/him pronouns, the government's seeming focus on profits over people made him wary of how much the government would prioritize his family's health, leaving him to wonder if having children in this province would be a good idea. When I asked about how COVID-19 in Saskatoon had affected his thoughts on parenthood, Justin replied,

We kind of have a government where they're like, "Yeah, we should take the vaccine, we want to open up as fast as possible, we want to have this economic prosperity, we want to bring back the revenues", but we're not always thinking about the kind of people who are going to be interacting with each other in the public who are potentially getting themselves sick. So you've also got the idea of, does the government actually

care about my wellbeing, my family's wellbeing, and our health? I think that kind of worries me a bit when it comes to the SK's government's policies. [...] What if I have to send my kid to school with someone who doesn't vaccinate their kids or somebody who doesn't want their kids to wear a mask, or whatever. If something like this happens again and, for whatever reason, so-and-so isn't masking their kids and there are no policies to say that unvaccinated kids can't come to school, then you've got a problem.

In this discussion, the meaning of COVID-19 in relation to parenthood centred around aspects of the pandemic experience over which Justin would have little control but would seriously affect his life. The uncertainty he feels about how his family would fare in similar pandemic conditions under a similar government profoundly resonates with his uncertainty about having children, which rests on similar variables. Before we discussed COVID-19, I asked Justin how he felt about the possibility of parenthood, and he said, "I think I would like to have kids if I knew that I'd be stable enough and the world would be stable enough to raise a kid in my 30s, but I think right now, it's hard to say." Stability and certainty are key elements of family planning for people of many orientations and backgrounds, but family stability can be lacking for queer families, often due to queer-harming laws, policies, and social attitudes (Younkin, Firmin, and Zurlinden 2022; Sullivan 2004; Sullivan 2013).

Family stability in general faced further deterioration during the COVID-19 pandemic, and people in Saskatchewan have since expressed their displeasure with the Saskatchewan government's pandemic response, with half of Saskatchewanians believing that Saskatchewan premier Scott Moe did "a bad job" handling COVID-19 (Jadoo 2020; Angus Reid Institute 2022). In making reproductive decisions during the COVID-19 pandemic, the impact of the systemic pandemic response was, for many of those who shared their experiences with me, to reduce certainty in their plans for building families and to destabilize the structures around which people were making their decisions, such as childcare, schools, and jobs. Like Justin, they were left feeling unsure that they would be able to build the families that work for them because they were not sure of the government's support during this time of crisis and because the nature of pandemic responses (i.e. social distancing, quarantines, and isolation periods) eroded social stability by disrupting the social mechanisms on which people have come to rely (Maestripieri 2021; Schormans et al. 2021). If some people cannot trust that the families they make will not

remain safe and supported during a crisis like a pandemic, they may not trust that they can build the families that work for them.

In addition to the direct impacts of COVID-19, the pandemic also served to highlight the important global issues that some interlocutors were also considering when making reproductive decisions, including climate change and the carbon footprint of bringing new people into the world, global population trends and their interaction with housing, food, and climate crises, and issues of equity and equality in who can access healthcare, support, and services both daily and in crisis situations like the height of the COVID-19 pandemic. As a result of experiencing the current pandemic, some of those who shared their stories with me had to deeply consider their values and ethics as weighed in the desire for family vs. the impact of making that family. Mariana has given the ethics of her reproductive decisions a great deal of thought because of the confluence between the pandemic and her academic program, which includes a heavy focus on environmentalism. As she put it,

In the COVID context, I think it makes sense, when you're part of [her program] and all you talk about is climate change, you don't see the future as very bright. So that's something that worries me. That's a reason why I wouldn't put that future into someone else's life.

Mariana's experiences within her academic and professional context have attributed a specific meaning to the events of both climate change and the COVID-19 pandemic, which contribute to a future that she may not want for her family. It is not only the separate impacts of climate change and COVID-19, but the syndemic-like way they contribute to each other, increase the impact of each other, and, possibly, create each other, following the trail to COVID-19's likely origins in the bushmeat market (Di Ciaula et al. 2021; Maier 2021; Platto et al. 2021). By building a family, does she want to potentially contribute to the inequity and hardship likely to result from the effects of climate change and exacerbated by the responses to and consequences of pandemics (Di Marco et al. 2020; Sharif and Dey 2021)? Does she want her potential family to live with the various types of instability caused by climate change that also make pandemics more likely and their outcomes more severe (Lawson et al., 2019; Cooper and Nagel 2021)? Mariana is thus questioning the ethics of her reproductive decisions and considering how her family might both impact and be impacted by the world in which they are living, and in this, she

is not alone (Schneider-Mayerson 2022; Schneider-Mayerson and Leong 2020; Lin et al. 2021; Wenham et al. 2020).

To Mariana and many others, the relationship between COVID-19 and climate change seems obvious, as both are serious crises that need systemic responses and buy-in from the population to enact effective management strategies (Manzanedo and Manning 2020; Zang et al. 2021). In addition, COVID-19 and climate change seriously impact individual lives and the social and institutional structures that affect individuals. Both have produced significant anxiety and emotional turmoil among the population (Grundmann 2021; Marazziti et al. 2021). In considering these similarities and their effects on reproductive decision-making on a broad scale, it is important to remember the intersectional impacts, specifically the gendered impacts, of both COVID-19 and climate change. People identified as women faced increased domestic burden caused by COVID-19 restrictions, causing them to leave jobs or reduce their careers in greater numbers than people identified as men (Power 2020; Carli 2020). In addition, people identified as women are bearing the brunt of the impacts of climate change in their day-to-day lives, such as travelling greater distances to complete domestic chores, increased gender-based violence, asset insecurity, and deepening poverty (Sultana 2021).

This is not to say that both COVID-19 and climate change have not also affected people identified as men, but that the impacts on men, especially in rich developed countries, are less material than the impacts on women. Instead, men are more likely to struggle with impacts to their sense of masculinity, to their self-identities, and to the way they position themselves in relation to their employment as it is affected by the dual threats of COVID-19 and climate change (Hultman and Anshelm 2017; Nelson 2020; Mellström 2020). The intersection between COVID-19 and climate change, and between the loss of livelihood and increased stress facing women and the strong identity challenges faced by men, have led to reproductive decisions being made in an atmosphere that is, as Mariana describes it, not "very bright." Especially for queer people, seeing a future for the families they build, sometimes called "futurity," can be an essential aspect of reproductive decision-making (Sullivan and Davidmann 2015). The state of the world, including the very real contextual considerations of both COVID-19 and climate change, significantly impacts and possibly restricts the reproductive decisions of queer people in Saskatoon.

The COVID-19 pandemic and the social, institutional, and personal responses to the pandemic have been a serious disruption to some peoples' everyday lives, a new and challenging state of the world with which most people must contend when making their reproductive decisions. Social distancing, work and school from home, isolation, and vaccine requirements have changed how our lives and our worlds look. For many of the people who shared their stories with me, building families that work when childcare, employment, support, and health are all in a state of flux is a difficult task, one that led some interlocutors to wonder if it is possible to build families that work for them under these conditions. For others, the interruption to regular social conventions and the monitoring of queer peoples' family creation that came with them was a welcome reprieve in which they could explore making non-normative families. Beyond the direct impacts of COVID-19 in the daily lives of interlocutors, this pandemic also serves as a metaphor, a currently salient example of what some people see as the larger "state of the world" or the ever-shifting and sometimes-frightening context in which they must make their decisions. In addition to COVID-19, more significant concerns about the state of the world include the current political climate, such as the growth of extreme far-right political parties across the globe, the ethics of producing children when global resource consumption is at all-time highs, and as previously mentioned, the very real effects of climate change (Shead 2021; Williams 2021). For queer people in Saskatoon, these elements of life and their meanings have led some people to ask serious questions about the feasibility of creating families that work.

## 3.5 - Conclusion

Everyone faces challenges when making families that work, but queer people must overcome additional barriers that cishet people do not often confront. I have highlighted the three barriers that were most salient in my conversations with queer people in Saskatoon – the problem of the nuclear family, the difficulties with pregnancy, and the effects of the COVID-19 pandemic. In Saskatoon, the social primacy of the nuclear family construct is increasingly inappropriate when applied to families broadly, but it has never been a useful way of talking about queer families. As a result, the people who have shared their stories with me discussed how the nuclear family construct led them to think that families could only be built in the nuclear fashion of two married parents having biologically related children, how they feel that the culture of Saskatoon (and Saskatchewan more broadly) would not welcome non-nuclear approaches to

parenting, and also how they have tried to subvert or challenge the nuclear family construct on both personal and institutional levels.

Another challenge of pregnancy comes with a host of gendered expectations that can be a struggle for any person considering pregnancy but are especially difficult for queer people who may not fit the gender assigned to their anatomy. In addition, the surveillance and norms that queer people experience during pregnancy can make physical reproduction challenging for some but turn it into an impossible obstacle for others because of problems with dysphoria and social reactions. The third challenge is that the COVID-19 pandemic plays a nebulous role in building families that work. For some, the isolation and social distancing measures created freedom to explore queer family structures outside the usual social surveillance they have experienced. For others, the insecurity of social support surrounding families that was demonstrated by the COVID-19 control responses made people unsure if they would want to reproduce, knowing that they too might face a lack of childcare and combined work/school from home situations. In addition, the worrisome "state of the world," as seen through the dual lenses of COVID-19 and climate change, can reduce hope for the future and introduce uncertainty as to whether bringing a child into this situation is really the best decision for them and for the potential parents. By exploring the reproductive decision-making processes among queer people in Saskatoon, I hope to highlight how important it is for them to make families that work within the social and institutional structures that are designed to support cisgendered heterosexual families.

## Chapter 4 – Safety

#### 4.1 - Introduction

In queer circles, safety is usually a primary concern, regardless of the topic at hand. The queer collective consciousness remembers the violence many of us have faced: the brutal police raid of the Stonewall Inn and the resulting riots that led to the beginning of Pride celebrations (Pitman 2019), the horrific death of Matthew Shepard, who was beaten and left to die hanging on a fence post in Wyoming (Noelle 2002), the mass murder at Pulse, a queer nightclub in Florida where 49 people died and 53 were injured when a shooter opened fire during Latin night (Stults et al. 2017), the one-time prevalence of "queer-bashing," or public assaults on people who "look" queer (Petersen 1991; Lunny 2013), and the often unseen and un-reported killings of queer and transgender sex workers, frequently racialized people, who have been marginalized by society and generally die in ways that go without investigation or systemic reaction (Stardust et al. 2021). This list does not include the deeply ingrained systemic violence aimed at queer people, such as the governmental and healthcare response to the HIV/AIDS epidemic in the 80s and 90s that left many gay men to die while other members of the queer community had to care for them because they were denied treatment elsewhere (Bell 2020). While the current cultural zeitgeist in Canada is one of respect and acceptance of queer people, we know that our safety is never guaranteed.

The precarity of queer safety is especially true when discussing reproductive decision-making. While queer parents are now staples of mainstream media, we remember that it was not long ago that queer parents were considered unfit, that children raised by queer parents would be damaged or tainted, and queer people were, and still are, legally prevented from forming recognized families in many places (Hicks 2005; Riggs 2012; Kazyak et al. 2018). Queer people also face safety difficulties in accessing options to prevent having children – queer people can have problems accessing birth control and abortions in a safe and timely manner and may even face violence in their journey to seeking this medical care (Hudak 2021; Mollon 2012; Carpenter 2021). When I set out to hear about queer peoples' reproductive decision-making in Saskatoon, I did not ask about safety, but because of my understanding of the perilous safety experiences of being queer, I was not surprised when it repeatedly emerged in the stories people have shared with me.

There was a distinct contrast in safety experiences between people born in other countries but who have chosen to move to Saskatoon, often focused on aspects of physical safety, and the experiences of people born and raised in Saskatoon, who generally talked about elements of social safety. This contrast highlights the experiental, contextual nature of safety and how meanings of safety are constructed (Haukelid 2008; Bubandt 2005). For both groups, the formative experiences in their countries of origin have highlighted the most significant threats they feel to their safety and how their reproductive decision-making processes have been impacted by those threats. The difference emerges in the nature of those threats: for those not born in Canada and who were born in cultures lacking Canada's relative acceptance of queer people, threats to their bodies and their lives were the most poignant. For people born in Canada who have grown up with the experiences of relative physical safety, the focus shifted to feelings of belonging, of acceptance, and of freedom to live their lives how they choose, a concept I refer to as social safety. In combination, these experiences highlight the critical role of safety in queer reproductive decision-making in Saskatoon.

# 4.2 - Defining Safety

Before I explore how safety experiences were a key aspect of reproductive decision-making for the people who shared their stories with me, I first want to establish what I mean by safety. Constructing a critical definition of safety is difficult, as there are many different ideas about what safety can mean. Being safe is being "free from harm or risk; not able or likely to be hurt or harmed in any way: not in danger" (Merriam-Webster 2021b). The dictionary definition begins on the footing of "risk," a term central to notions of safety. Risk is the "possibility of loss or injury" (Merriam-Webster 2021a). Safety is often defined and considered by relationships of risk (Erikson 2012; Sjöberg and Giritli Nygren 2021), to the point where discussions of safety are sometimes discussions of risk (Giritli Nygren and Olofsson 2020; Giritli Nygren, Öhman, and Olofsson 2017; Nygren, Öhman, and Olofsson 2016; Olofsson et al. 2014; Chapman 2006). However, the academic literature reveals no agreed-upon definition of "risk." The definition of risk in the social sciences that finds some consensus is "uncertainty about and severity of the consequences (or outcomes) of an activity with respect to something that humans value" (Aven and Renn 2009). If safety is being free from uncertainty about the outcomes of an activity we value, safety must have a quality of certainty. Certainty is a cognitive and affective quality – you

"feel" certain about something; you think about the potential certainty of outcomes of a particular action. Risk, in this definition, is also centred around the results of activity – something must be occurring to introduce risk and to remove safety. If it is based in risk, safety is experiential, in that we experience both safety and risk. Safety based on risk is a complex phenomenon that covers many aspects of being a person (cognitive, affective, experiential) and is a widely-used approach to safety.

Risk can be in multiple aspects of safety; some definitions can be exclusionary, as risk has broader implications for people than mere uncertainty. The affective qualities of safety can be paired with aspects of social justice such as "the right to feel safe" (Sjöberg and Giritli Nygren 2021, 1251), and feeling safe has been "naturalized as a fundamental need" (Hutta 2009, 251). Safety is therefore personal, as each person wants to feel individually safe, but it is also a social construct, as the notions of rights and fundamental needs speak to "deeply entangled [and] complex relationships of power in the public arena" (Sjöberg and Giritli Nygren 2021, 1253). This leads us back towards the idea of risk, as Anna Olofsson and colleagues note that social problems, frequently the result of power differentials in the Foucauldian sense, are now framed in terms of risk and "thus, 'social problems', often associated with particular categories of 'vulnerable' people, are redefined, reframed, and ultimately managed as 'risks'" (Olofsson et al. 2014, 417). The difference is that earlier definition of risk focused on the certainty of outcomes from activities that humans value, while safety in relation to social justice has a quality of embodiment in that it defines people, not outcomes, as risks. Both constructs of risk and safety must therefore be considered through an intersectional lens to understand how these constructs impact groups outside of the majority. This includes the safety and risk experienced by queer people.

Safety for queer people, just like all safety, is multifaceted. Safety (or a lack thereof) can emerge from discourses by and around queer people (Hackford-Peer 2010), can be defined by spaces like school, work, or home (Collins 2013; Moran 2002; Kentlyn 2008; K. Browne, Bakshi, and Lim 2011; Weems 2019), and is a part of queer social membership more generally (Giritli Nygren, Öhman, and Olofsson 2017; Nygren, Öhman, and Olofsson 2016; Ben Daniel and Berwick 2020). A common theme that unites these aspects of safety is that, by being publicly queer through performing their genders and sexualities, queer peoples' safety is inherently less certain than those of cis-gendered heterosexual people, a concept Judith Butler

refers to as "precarity" (Butler 2009). This intersectional precarity is created and influenced by the social and political institutions that seek to reduce precarity for the entire population and, in so doing, end up introducing risk and reducing safety for minority groups like queer people through insufficient social and economic support. Precarity can be seen when trans people accessing assisted reproductive technology providers and services often end up with negative experiences or even iatrogenic harm, damaging their reproductive safety through services meant to support that safety (James-Abra et al., 2015). Precarity can be seen when governments and institutions refuse to acknowledge the children of queer people because their vital statistics systems are set up to support heteronormative approaches to reproduction, reducing familial safety by damaging queer families and preventing them from accessing necessary institutional supports (Leibetseder and Griffin 2018). Precarity can be seen when reproductive loss support groups exclude queer parents, especially non-gestational or non-biological parents, when the complexities of queer reproduction mean that queer people are even more likely to face reproductive loss than their cisgender heterosexual peers, putting their social and mental safety at risk (Craven and Peel 2014). No one is guaranteed safety, but queer people's safety concerning their reproduction is especially precarious.

Resulting from these factors, I arrive at the definition of safety I am using for this chapter. First, safety includes assessments of risk, especially as risk relates to uncertainty within experiences. Safety is experienced at many different levels of social abstraction, from the intimately personal safety of whether someone may feel like becoming a parent is right for them to the global-social level of feeling it is not a safe time to have children because of the growing effects of global warming and population growth. Safety is intersectional, resulting in queer people's safety being precarious and sometimes damaged by those forces created to help protect safety.

Lastly, safety can have embodied qualities, such as labelling particular people or groups of people safe or unsafe, like the restrictions on men who have sex with men donating blood. Between 1977 and 2022, Canadian men who had had sex with men (MSM) were restricted from donating blood through the Canadian Red Cross and its successor, Canadian Blood Services (Jubran et al. 2016; Health Canada 2022). This restriction began as a lifetime ban on donating blood that lasted from 1977 to 2012 (Canadian Blood Services 2023). These measures were communicated to be a response to growing rates of HIV which could be passed on through blood

transmissions, a disease that was growing prominent among MSM (Caffrey et al. 2022). The restriction was intended to reduce the potential risk of passing HIV along through donated blood, which had happened 1200 times between the late 1970s and early 1980s as the Canadian Red Cross was slow to implement HIV testing (Harvey 1996). Insufficient testing resulted in what is now called the Canadian Tainted Blood Scandal, when it was found that, between 1980 and 1995, 43% of Canadian hemophiliacs who had received donated blood were infected with HIV due to "tainted blood" (Dean 2002). Blood donation services focused on MSM because anal sex is more likely to pass along HIV infections than other sexual intercourse methods and , but instead of focusing on sexual behaviours, the message of the restriction was that MSM as people were inherently more risky and less safe than people with different sexual practices (Baggaley, White, and Boily 2010; Wainberg et al. 2010; Grace et al. 2019).

The restrictions on MSM donating blood have changed quite rapidly over the last decade. In 2013, Canadian Blood Services removed the lifetime restriction and instead said that MSM could not donate for five years after their last MSM experience, then 2016 saw that restriction reduced to one year, and in 2019 it was reduced to three months (Canadian Blood Services 2023). 2022 saw the MSM restriction removed in favour of restricting donations for three months after any person's last experience with anal sex regardless of their gender presentation and the gender of their sexual partners, though some queer people argue that this change in rules can still stigmatize queer people by restricting their donations instead of testing drawn blood (Lam 2022; Kislenko 2022). While institutional attitudes about the embodied aspects of queer safety are changing, the social dialogue about how queer people embody safety in relation to blood donations and risk may be slower to evolve, leaving queer safety a complex topic to navigate (Grace et al. 2019; 2020; 2021) Safety, then, is a multifaceted, multi-experiential, intersectional consideration of how certain a person is that they will not be harmed by their experiences, with a broad scope of what could be considered harmful.

## 4.3 - Safety in Place - Queer Reproductive Safety Between Countries

In considering safety, context shapes the meanings of reproductive decision-making that were found by the people who shared their stories with me. Notably, the cultural environs and the norms associated with those places had clear influence on notions of reproductive safety. The experiences of participants who immigrated to Canada focus directly on physical safety, whereas

the experiences of Canadian-born participants focus on social safety. Interlocutors in the former category hailed from Brazil, Mexico, Colombia, and the Philippines. Though their specific experiences differed, the meanings they associated with safety and reproductive decision-making carried a common thread. Similarly, people who were born and raised in Saskatchewan had a wide variety of lived experiences and exposures to cultural nuance, but there were strong similarities in their descriptions of their experiences.

For people who have immigrated to Saskatoon from other countries, physical safety was key to their considerations of safety within reproductive decision-making. I define physical safety as the assessment of the perceived threat (or lack thereof) to a person's body. Many of these immigrant interlocutors were seeking reprieve from the levels of violence and threat they felt in their home countries. For Claudia, remaining in Brazil to have children was unacceptably dangerous. When I asked her what aspects of living in Saskatoon were important, she replied with a story.

We had some friend that went through something really terrible. He was one of my husband's best friends and he was walking with his fiancée back home, and they were... A van stopped right by them, and the guys were high and were trying to mug them, and he was like getting his backpack off to give it up. [...] And they just shot him in the head, right in front of his fiancee. He died in the spot. That was the kind of stuff you don't want to see happen to your friends and family, and it was getting more common where we lived, so we really wanted to go somewhere safer to live in. Well, eventually,

I really want to have kids, so yeah, I dunno, I need a safe place to raise them, I guess. Living in Saskatoon represents safety in that the likelihood of Claudia's friends, her family, her partner, or her children being violently injured or killed seems lower than if she had remained in her home country. Just being there was unsafe enough, enough of a risk, that she made decisions to live her reproductive future in a different country. Mariana echoed these sentiments when answering the same question, "Well, I feel way more safe here than I do in Mexico, for example. So that's something that I value."

For some other immigrating interlocutors, that lack of physical safety was more intersectional. Thayara brought her queerness into the conversation when I asked her a similar question about the importance of living in Saskatoon. She replied somewhat flippantly and more succinctly: "Well, I'm not afraid of dying because I'm a lesbian." Her concerns are well-

founded, as her home country of Brazil sees high rates of violence in general, and especially against queer people (Francisco and Muggah 2020). The risk of direct or implied physical violence, and therefore a lack of physical safety, directly impacted the reproductive decisions some queer people make in immigrating to Saskatoon. Sometimes, even the potential threat to their physical safety without directly worrying about violence can be enough. This was the case for Abi, moving from the Philippines to Saskatoon. She described the potential problems that she and her partner would face when visiting family in the Philippines, saying:

What I'm concerned about is [...] if I marry my girlfriend and we have our kids and then I bring them back to the Philippines, people there are conservative and I personally don't know how I would react if people were going to be homophobic against us and what they're going to say about my kids. [...] As much as the Philippine media would say, 'Oh the Philippines is very tolerant', actual situation on the ground might be hit or miss. Some people might be progressive but a lot of people are not.

While Abi is not directly concerned with the potential for physical violence, the distinct uncertainty about how people would react to a visibly queer family in the Philippines contains a possible threat to their safety, and while Abi did not mention it directly, the homophobia she fears has the potential to threaten her and her family's physical safety. Violent homophobia and anti-queer sentiment is an ongoing struggle in the Philippines, so while Abi and her future family may not fear a direct threat to their bodies, the uncertainty about the potential for things to turn physical when there is a problem with anti-queer violence in the country is definitely enough to leave queer people feeling unsafe (Dakis and International Gay and Lesbian Human Rights Commission 2012; Human Rights Watch 2017; Redfern 2021). For Abi, there were many other reasons to move to Saskatoon and make her reproductive decisions here, but the ability to fully embrace her reproductive future as a queer person also raises the importance of considering the potential physical safety of her, her partner, and their future children.

When considering physical safety, others highlighted aspects of safety regarding their physical bodies and their health. Juan saw physical safety in the form of equal access to healthcare as a queer person. In describing his experiences in his home country of Colombia, he said, "Access to healthcare and all of that other stuff, it's related to who you are and what you can get to or how much money you have, so it's not very easy to get into, like... Yeah if you want to go to a medical check up, you always have to pay, it's not that easy." The uncertainty of

reliably accessing healthcare, especially if one is experiencing economic instability, is a form of structural violence (Farmer 2004; Garcia-Subirats et al. 2014; Ruiz-Sánchez 2021). This violence can lead to uncertainty about remaining in good health and being able to address urgent health concerns promptly, all of which can degrade a person's feeling of safety. This experience is magnified for queer people, many of whom face anti-queer treatment when seeking medical care and deal with higher rates of mental health problems than their cishet counterparts (Choi et al. 2019). Juan pointed this out when he compared Saskatoon's medical system to that of Colombia, continuing,

And like also, doctors are more professional here as well. They understand if you're gay or LGBT or whatever. They also know how to understand that position and say hey, if you need to do this test or if you need access to something, feel free to reach out. Whereas that doesn't happen in my country either because there's not that many policies for LGBTQ people as well, just kinda... Sad.

Gaining a feeling of certainty that the medical professionals helping you understand your needs and who you are as a person, essentially validating you and your needs, can greatly increase patient confidence in healthcare providers and improve the patient-provider relationship, leading to greater feelings of certainty and safety (Epstein et al. 2007; Linton et al. 2017; Heyhoe et al. 2016). This is especially important for people with mental health concerns, which queer people are more likely to have (Russell and Fish 2016). While this particular aspect of physical safety is not directly related to reproductive decision-making, it does influence the context in which people like Juan are making their reproductive decisions, making it an important aspect of safety to consider.

For this group of interlocutors, coming to Saskatoon was an important part of the meaning they find in their reproductive decisions because Saskatoon represents a reprieve from these worries concerning their physical safety. Of course, no interlocutor thought of Saskatoon as a perfect place, but it was a place where they found increased freedom to make their reproductive decisions because, in Saskatoon, there is no undercurrent of fear about facing random violence in the street or being the target of other types of violence due to their queerness. This is not to say that Saskatoon has no history of violence against queer people, as queer people in Saskatoon have been and continue to be targeted for violence (Modjeski 2016; Atter 2021). But rates of violence in Saskatoon for the relatively privileged group of university students who made up the

majority of these interlocutors are fairly low, and open "queer-bashing" is far less prevalent than it used to be (Moreau and Statistics Canada 2021; Statistics Canada 2021b). Thus these interlocutors have found that Saskatoon carries the meaning of a safer place in which to consider their reproductive decision-making and potential families in comparison to their countries of origin.

#### 4.4 - Safety in Place - Queer Reproductive Safety in Saskatoon

Compared to the experiences of people who have moved to Saskatoon from other countries, interlocutors who were born and raised in Saskatchewan were also concerned about safety, but they were focused on what I am calling social safety. Borrowing from Jolanda Maas and colleagues, I define social safety as a subjective feeling of safety (or lack thereof) resulting from interactions between people within a social setting, whether public or private (Maas et al. 2009). Most often, feelings about social safety center on concerns of sociality; how does a queer person maintain feelings of safety and security within their sociality, the "dynamic and interactive relational matrix through which human beings come to know the world they live in and to find their purpose and meaning within it" (Long and Moore 2013, 2)? In Saskatoon, certain mainstream aspects of this relational matrix are biased against queer people as anti-queer sentiment is still prevalent (Edkins et al. 2015; Leroux 2021). Uncertainty and risk within mainstream sociality jeopardize queer people's safety within a social context. It is less likely that they will encounter similar physical violence that Eugene Grosh, who was attacked at a bar for his queerness, and others have endured (Warick 2016). Still, severe potential remains for receiving stigmatizing, marginalizing, and/or discriminatory treatment, not having a "place at the table" (Bawer 2008) because they are queer. As Long and Moore point out, they are trying to know the world they live in, and most applicably to this thesis, they are trying to find meaning within that world. In this context, born-and-raised Saskatchewanians shared their concerns about safety within their reproductive decision-making.

Sometimes the meaning of social safety comes at the intersection of multiple avenues, such as religion, culture, and politics. Interlocutor Lula speaks to this complexity by situating their perception of their social safety in Saskatchewan at the nexus of the Catholic education system of Saskatchewan and the province's recent history of political conservativism. For them,

Saskatchewan's separate but publicly funded Catholic education system creates space for antiqueer sentiment, thus reducing their experience of social safety. Lula explains,

[G]oing through the Catholic school system in SK where it's so... Because of the way it's set up in SK, it's really shoved down your throat, how unique it is to SK, and you learn the history of the legal battle to get the right to have a Catholic education. It's so enshrined and tied into being a part of SK, and I think that leads to a lot of sheltered prejudice.

Lula sees a lack of social safety arising from the prejudice introduced against queer people by the Catholic education system as a key part of who they became as a queer person. To Lula and others who shared similar sentiments, the current and ongoing Catholic anti-queer views (Loughlin 2018; Crary and Henao 2021) that influenced them and their peers made room for homophobia in the lives of the people who have attended these schools, an argument they are not alone in making (Callaghan 2016; 2019). The "enshrined" institution of separate Catholic education, and Lula's experiences there, contribute to their sense of feeling less than safe in their sociality in Saskatchewan. Lula then discusses how these identity-formative Catholic school experiences contributed to their political values and how they interact with the province's political values. Directly following their previous statement, Lula continued:

And I think too, I've only ever really known a conservative SK. I'm well aware that we're the birthplace of Medicare and we were one of the most progressive provinces. But as I was gaining my political awareness as a kid, it was right around the time when that flipped. So I've only known a SK that is very conservative and has only become more and more conservative as I've gotten older.

The tension Lula experiences in Saskatchewan's political identity results from the tension between the province's socialist past and its conservative present. Between the Cooperative Commonwealth Federation and the NDP, the left led the legislature for 43 of the 59 years between 1944-2003 (de Vlieger 2007; Bingaman 2013), but that changed with the current fifteen-year conservative leadership of the Saskatchewan Party.

The reigning conservative political power ties into Lula's perception of social safety within the province of Saskatchewan (Frew 2020; OUTSaskatoon 2020; Cameron 2021). Social reproductive safety is of special concern for queer people, especially in medicalized and politicized contexts. Such was the case when Rowan told me about her experience with abortion.

Rowan's birth control failed and she had a non-viable pregnancy<sup>2</sup> that she chose to terminate. Rowan's experience with abortion is not uncommon, as Canada has averaged 92,633 abortions per year since 2007, with an average of 1,932 originating in Saskatchewan (Abortion Rights Coalition of Canada 2022; Canadian Institute for Health Information 2022). And while I am unable to find any information regarding rates of abortion among queer people in Canada, abortion itself is a difficult subject for queer people, as they are more likely to have unplanned pregnancies that end in abortion than cisgender or heterosexual women, they are less likely to be able to access healthcare services for abortions, and when they do, they are more likely to experience negative or harmful treatment during appointments (Hartnett, Lindley, and Walsemann 2017; Carpenter 2021; Wingo, Ingraham, and Roberts 2018).

While Rowan did not share any difficulties in accessing abortion services or care, she did explain the intensity and significance of her experience to me:

And [abortion] is something that I have always been very strongly for, and at the time, it was the choice that I wanted. I was not ready, we were not ready, and I don't think would have worked out even if I was. But, I mean, definitely had to think about where I stood on all of those things when that happened. And it's definitely something that I continue to think about because of that. A lot of post-stress from it, it's something I'm still working through, but that kind of really made me have to start thinking about [reproduction] in a more serious light.

In describing her experiences, Rowan focuses on aspects of meaning – what does abortion mean to her? What does it mean to others? What meaning does reproduction hold for her? What meanings are present in the feelings she has after her experience with abortion? Together, these phenomenological questions add up to serious consideration about the social aspects of abortion or those aspects of the "dynamic and interactive relational matrix" that is human sociality regarding abortion. (Long and Moore 2013, 2). Part of those meanings are concerned with Rowan's social safety in accessing abortion services – the safety of her relationship with her partner ("we were not ready"), her emotional safety ("a lot of post-stress from it"), and positioning herself within the broader sociality surrounding abortion ("had to think about where I stood on all of those things"). For Rowan and other queer people accessing abortion services,

<sup>&</sup>lt;sup>2</sup> A blanket term for a pregnancy that cannot result in a live baby being born. There are many potential causes for non-viable pregnancies (Doubilet et al. 2013).

their social and reproductive safety can be impacted by their individual experiences, how those experiences are situated within the broader context of their sociality, and how the meanings of those experiences interact with that sociality.

These diverse aspects of safety and sociality were of concern in June 2021 at the time of our interview, but now, writing in November of 2022, it is even more serious. On June 24 2022, the American Supreme Court overturned Roe v. Wade, the iconic case supporting abortion rights, and thus walks back much of those rights in the United States (Gerstein and Ward 2022; Housman 2022). The current justices of the Supreme Court are majority conservative, and conservativism is linked with anti-abortion sentiments within the USA and Canada (Oriana Gonzalez 2022; Tasker 2022; Shivaram 2022). The shockwaves of the overturn of Roe v. Wade in Canada could stir up similar anti-abortion sentiment in Canada that would likely disproportionately impact the queer community (Chung 2022; Lavery 2022). In Saskatchewan specifically, the province already has problems providing abortion services to many who need it, so concerns about reductions in access and rights coming from the conservative provincial government may be well-founded (Rasche 2022a; Vescera 2022; Kurz 2022). As of November 2022, the abortion debate has not been reopened in Saskatchewan's political and legal institutions, but the visibility of the issue has raised heated voices with multiple viewpoints (Ghania 2022; Hunter 2022; Rasche 2022b) Like Rowan, many queer people could have been harmed or killed if their pregnancies had continued, so these queer people would have had their physical, reproductive, and social safety threatened by reductions to Canadian abortion rights.

Social safety surrounding queer reproductive decisions can also result from interactions between personal relationships, institutions, and governing bodies. In my conversation with Ethan, a Canadian trans-masculine person raised in Quebec who moved to Saskatoon at the age of 12, considerations of safety revolved around his legal status in his children's lives. Ethan, his wife, his wife's ex-partner, and the sperm donor who helped them create their family form the parents in a queer family with their four children, creating a legally tricky situation for Ethan. He is not biologically related to his children, which made being listed as a parent on the birth certificate issue a challenge. Nicole, a different participant who shared her story with me, worked with her partner and their MLA in 2017 to legally recognize queer parents who may not be biologically related to their children. However, these laws were not in place when Ethan's family had their children. It is in that context that Ethan shared his story with me.

When we had our child, we asked our pediatrician and the pediatrician shrugged and said, "Sperm donors don't have to go through this, so you shouldn't have a problem." So we didn't bother. When we had the child in the hospital and it was time to sign the birth certificate, we asked the nurse, "What do we write down for the father?" and the nurse said, "Sperm donors don't have any problem, and people write down the wrong name all the time. Just write down your name." So I did. Technically that's a felony.

[...] If that ever comes back to bite me in the ass, I am facing jail time. [long pause]

Because... This is not a small deal. But there were no systems in place for us and our family. What we would have had to do, legally, is write down the sperm donor as father and then he would have to give up his legal rights as a father, I would have to adopt my own kid. And that's why Nicole and [her husband] went to court to fight. They spent I don't know how many thousands of dollars, last time I heard it was over \$14k. We never had that money. We are very low income, like I have chronic illness, my wife has bipolar, we already had a child. That was not happening.

In creating their family, Ethan's social safety was compromised. He committed a crime by signing his child's birth certificate as the father because he did not contribute genetic material to the creation of that child. While it is unlikely that he would ever be punished for that crime, the experience invalidates his role as one of his children's parents and can compromise his feeling of safety within his family and the society in which his family lives, an experience shared with other queer parents who are not biologically related to their children (Chbat and Côté 2022). The illegality of his actions due to an inflexible legal institution also puts his family's feelings of social safety at risk.

Ethan and his family felt the injustice of not having any systems that worked with how they created their family. They knew this was not fair, but due to their intersectional position within Saskatoon's social matrix (low income, chronic illnesses, multiple children to support), they could not pursue a legal resolution the way that Nicole's family did. Queer people are more vulnerable to poverty than cisgender heterosexual people, and they are more likely to face issues with chronic illness and mental health, with non-birth parents facing different issues than birth parents (Badgett, Durso, and Schneebaum 2013; Kia et al. 2021; Abelsohn, Epstein, and Ross 2013; Grant et al. 2014; Rothblum 2020). These issues contribute to difficulty in accessing legal aid and additional barriers to seeking social justice for the challenges queer people face in

general, but especially in making their reproductive decisions. The difficulties presented to Ethan's family are not likely to be replicated, given the advocacy of people like Nicole, but how Saskatchewan's political and legal institutions influence their feelings of social safety definitely impacts the reproductive decisions queer people are making here.

#### 4.5 - Conclusion

Canada enjoys a reputation as one of the safest places in the world to be a queer person, with 85% of the total population and 92% of the population under 30 years of age favouring the social acceptance of queer people (Ferguson and Ferguson 2022; Spartacus 2021; Poushter and Kent 2020). It was the third country in the world to formalize marriage equality by legally allowing same-sex marriage in 2005, with only the Netherlands (2001) and Belgium (2003) preceding it (Macintosh, Reissing, and Andruff 2010; CBS News 2016; McCann 2018). As such, queer people in Canada have had more time than most global citizens to construct legally recognized marriages, engaging in kinship structures of relatedness in an atmosphere of relative security. In comparison to many countries throughout the world, queer people in Canada have a different relationship to their sense of safety, based in the related legal and institutional protections they are afforded and the atmosphere of social acceptance from the majority of the Canadian population.

As a result of this relative acceptance of queer people in Canada, there is a marked difference in perceptions of safety between people who were born and raised here and people who were born elsewhere but have since relocated to Canada. The cultural norms related to reproductive decision-making were related to but distinct from those described by people originally from Saskatoon. For those coming from Central America, South America, and Southeast Asia, the meaning of safety they apply to their Canadian lives has its roots in their experiences with violence and risk against queer people in their countries of origin. Not having to be constantly vigilant about expressing their gender and sexuality in socially correct ways, as well as not having to constantly look over their shoulder to protect their sense of physical safety, removes a mental and emotional burden from queer people as well as physical damage that comes from violent events, allowing them to live more securely (D. Meyer 2015; Stoneman et al. 2020). Physical safety is essential for everyone, so much so that Canada has codified it as a basic

human right, and is crucial for queer people in the context of making reproductive decisions (Government of Canada 2017; Fixmer-Oraiz and Yam 2021).

For queer people born and raised in Saskatchewan who shared their stories with me, being raised in a culture that is accepting and offers relatively high amounts of institutional protections meant that their discussions of safety assumed their physical safety and it was never raised as a point of concern. Instead, they looked to their experiences of social safety to explain how, even though Canada is a queer-friendly place, mainstream society in Saskatchewan, and Saskatoon, in particular, is frequently less than safe for queer people. Important yet lacking elements of social safety for queer people in Saskatoon include legal safety surrounding their reproductive decisions and family creation, the political atmosphere of the province, and the complex intersection between the province's history with education and religion and how that intersection creates a social norm that is not inclusive of queer people. When added together, the impacts of these aspects of social safety make queer people feel less like they belong to their society than cishet people (Dryden and Lenon 2015). As many people noted, feeling supported by an accepting and loving community is an important aspect of queer reproductive decision-making among those who shared their stories with me, so feeling like they belong to the society they live in can have a huge impact on how queer people build their families (Freeman 2007).

Ultimately, the contrast between the focus on physical safety for queer people who had moved to Saskatoon, and the emphasis on social safety for queer people born and raised in Saskatoon, demonstrate the importance of context in considering safety among queer reproductive decision-makers who spoke with me. Coming from a cultural context where your experiences of physical safety are unsure at best because you are queer means that moving to a place where you no longer must be afraid for your physical safety feels so much safer. Whereas being raised in the context of assured physical safety means you focus on the aspects of your safety that are still not guaranteed, and in Saskatoon, those aspects are social in nature. The experiences queer people have had regarding their safety have enormous implications for their reproductive decision-making, and the uniqueness of those experiences means that what it means to be safe is a highly variable and contextual thing.

# Chapter 5 – Normativity

### 5.1 – Introduction

Whether or not we are aware of them, norms shape many of our experiences during our daily lives. Social norms play an especially important role in influencing how we interact with others and the expectations we have for them, and provide a sometimes-invisible set of rules and guidelines for how social experiences should go, all informed by specific cultural contexts (Horne 2001). For people who transgress these norms or actively resist them, there are generally consequences. Depending on the contexts and circumstances, these consequences can range from a raised eyebrow or shake of the head, to social shunning or disrupted relationships, to losing livelihoods or living spaces, to death (McDonald 2013; Huxley, Clarke, and Halliwell 2014; Lilja 2013). Norms, and the consequences for breaking them, pose problems for queer people, as implied by the word I have chosen to describe the 2SLGBTQIA+ community; the term 'queer' has its origins in describing something odd or strange, a concept that explicitly opposes and problematizes norms (Barnhart and Steinmetz 1999; Jagose 2013). As a result of being queer, these people live in a state of antagonism to the surrounding normative society. To be clear, not all queer people set out to consciously defy and disrupt norms, though active resistance to harmful normativity is a central aspect of North American queer culture. However, all queer people in dominant settler societies are considered defiant and disruptive through their very existence (Da Costa 2020; Getsy 2017). As such, many (though not all) queer people are conscious of the norms that are expected of them but to which they do not inherently conform (McDermott, Hughes, and Rawlings 2018; Idier 2020). For queer people, norms and their influence are an incredibly important aspect of day-to-day living.

The importance of norms is especially true for queer people making reproductive decisions, as there is a whole host of normative expectations surrounding reproduction that do not usually make space for queer experiences. There is a wide range of institutional, systemic, and social norms surrounding reproduction that exclude or alienate queer people, whether they are seeking to be parents or seeking to remain childless (Roulston 2021; Klesse 2019; Wilkinson 2020; V. Clarke et al. 2018). Norms surrounding reproduction are also culturally and contextually modulated, making an exploration of norms contingent on personal lived experience appropriately situated within the suitable framework (Ginsburg and Rapp 1991; Martin 2001). In exploring the influence of norms on experiences of reproductive decision-making in Saskatoon, I

found it became an exploration of norms deriving from multiple cultural backgrounds intersecting with the institutional influences of religion, politics, education, and colonialism. Queer experiences of reproductive decision-making in Saskatoon must contend with the variable and influential norms that seek to guide gender, sexual orientation, and questions of reproduction. For an in-depth discussion of the definitions of norms explored within this thesis, please refer to my discussion of theoretical perspectives in Chapter 1.4, but I will be examining the related concepts of heteronormativity, homonormativity, and repronormativity as they relate to queer reproductive decision-making among those who have shared their stories with me.

# 5.2 - Home-grown Norms in Saskatoon

Lula is a nonbinary person who grew up designated female at birth (DFAB), and though their family was not especially religious, Lula grew up attending a Catholic school. At school, Lula was exposed to specific flavours of heteronormativity combined with repronormativity, "I will say too that this probably a lot of my Catholic school trauma, you are a woman, your job is to bring life." This perception was not just aimed at people with uteruses and vaginas, as demonstrated by Lula's later observation about the seemingly inherent heterosexuality of people in Saskatchewan having children:

And maybe it's my Catholic education background, but it feels like part of the heterosexual rite of passage. 'I'm a man so I must have a son so he can take over the farm' and it just feels very... Nuclear, still, to be like, well, that's what we do! You know? We're straight, then we have children, then our children- it just feels very, it doesn't feel like, oh you have co-parenting situations. It all just feels very much like, children here are straight peoples' children.

Lula's experience within the public Catholic education system in Saskatchewan instilled within them a sense of what kind of reproduction is correct - heterosexual lineage-creating nuclear patriarchal families who will go on to create other heterosexual nuclear patriarchal families to continue the process. These religiously-influenced norms exclude other parenting approaches ("it doesn't feel like, oh you have co-parenting situations") even though the two-parent nuclear family is a decreasing way of creating families within Canada (Statistics Canada 2012; Zimonjic 2022). The provincial government of Saskatchewan also supports Christian schools grounded in anti-queer traditions and approaches, in addition to focusing on patriarchal gender and sexual

norms (Anton 2022). Saskatchewan has had a Catholic school system since before the formation of the province (Saskatchewan Catholic School Boards Association n.d.) and the availability of a separate school system is so engrained in the country's culture that denominational schooling for Catholics is a part of the 1867 Constitution Act (Government of Canada 2020) and has been subsequently upheld and strengthened by later laws.

Many Saskatchewanians are educated within the Catholic school system – in Saskatoon, of the 45,000 children enrolled in the main two school boards, 43% of them attended Catholic schools (Dayal 2021). And while not all children attending Catholic schools are Catholic, it does mean they are presented with Catholic education and worldview. To people like Lula, who went through the Catholic system, the Catholic influence on Saskatchewan's approach to reproduction is clear:

It's protected. There's a little more you can get away with as a teacher because it's enshrined within this institution that's viewed as part of the important part of SK. It definitely leaves a little more wiggle room for prejudice than you would have if we didn't have the worst sex ed curriculum in the country.

The perceived protection of educational institutions that deliver anti-queer messages alongside poor sexual education has had a lasting impact on the reproductive decisions made by Lula and other Catholic-educated queer people (Risch and Lawler 2003; Callaghan 2016; Montero-Ossandón et al. 2017). These influences lead to queer people like Lula coming away from their elementary and secondary education with expectations for reproduction to happen in ways that may be at odds, or even harmful, to the reproductive decisions many queer people are making.

Tying these heteronormative notions back to repronormativity, Lula extends her commentary on the heteronormative expectations in Saskatoon, saying, "Very much feels like breeder culture, I'll just say it." In queer slang, 'breeder' is a derogatory term, originally applied to heterosexual people who were heavily focused on their children to the exclusion of all else in their lives, but which eventually broadened to be a reference to straight people in general. The term implies they breed like livestock within an animal husbandry program, much like how Jonathan Swift used the term in *A Modest Proposal* (Mamo 2007; Swift 1729). Here, Lula uses the word to accent their assessment of how people in Saskatchewan are influenced by the prominent public Catholic education system and the culturally Christian nature of the province. Lula is also commenting on how these people have come to reproduce because it is how they

demonstrate their straightness and because it is, as Lula notes, a rite of passage to be considered an adult. Thus, they are expected to reproduce in heterosexual, state-sanctioned ways because it is a norm within the province.

Heteronormativity has also influenced Lula's views on reproduction through its interaction with notions of identity and performativity. In the following excerpt, Lula deconstructs the types and expectations of heteronormativity that they see as being culturally mainstream in Saskatchewan and how heteronormativity goes on to create homonormativity and its expectations for queer people and their reproduction.

Sometimes it feels like there are straight people that are straight people, then straight people that just happen to be straight. In Saskatchewan it feels like a lot more of the very straight people and less of the straight people that happen to be straight but are pretty chill. [...]It feels like, here there's a lot of straight-straight people. My best friend, I love her very much, but she married her high-school sweetheart, going to have kids right after the wedding, very straight straight, as opposed to more queer in the broad sense experiences of family and parenthood. [...] And also, an element of comp-het [author note: compensating heterosexual], where you're het but you're like, I need to be extra het. Not comp-het in terms of like, "I'll ignore my attractions", but there's a script where, like, if you're heterosexual, you have to follow this script. And buying into that entirely, being like, yes, I am hetero, so that means I will marry my high school sweetheart and we will have two kids. [...] To have the white picket fence and to have the children, like, of course that's what I would do. That's what heterosexual people do! ... [T]here are some people that are like, that's what you do, they can't imagine a world in which people don't want to have kids. But I'm like... Yeah, but there are other goals in life than having a big queer family. [Homonormativity is] one of the few acceptable ways to be queer, is like yes, you still have to have the goals we've associated with heterosexual people. You still want that life course, cause, we're okay with you if you follow exactly what we would like to do. And I think that, you're not gonna... I think that won't carry as much. I think a lot of younger people are like, no thank you, but it's become kind of normalized in its own way to have the gay nuclear family, essentially.

For queer people like Lula, these heteronormative expectations are not directly related to issues of sexuality and procreation but instead speak to the social scripts that shape how people are

supposed to publicly perform particular sexualities and lifestyles associated with their heterosexuality, similar to Judith Butler's take on the performativity of gender (Butler 2006). These heteronormative constructs that seem to be part of the majority culture of Saskatchewan put pressure on queer people to take on heteronormative roles and perform acceptable heteronormativity in their daily lives.

Rowan also described this heteronormative pressure but hers emerged in her family's expectations of how and when she would reproduce in a heteronormative way.

It never occurred to me, no matter who I was with, that that was something that, you know, if we both wanted it and when we were ready, that would be something that we would do. My mom has had this really annoying joke for years where, she had me when she was 27, and both of my cousins had their first kids at 27, so that I should also have my first kid at 27, and I've been saying since basically I was old enough for her to start making the joke, that that would not be happening. [laughs] It's a joke, but it's one of those things where it's like, I'm not doing it because that's the timeframe that people start having kids, I'm going to have kids when I want to and when I'm ready.

Rowan's mother's framing of her expectations as a joke may have given her heteronormative expectations a veneer of deniability by implying a lack of seriousness, but repetitively hearing these messages over time sends a message that there are specific timelines and ways of reproducing that are normal and therefore acceptable. The way humans communicate with each other can be packed with pressures of normativity, even if they are not as blatant as the jokes Rowan grew up hearing (Goldberg 2020). And while Rowan articulated this expectation of normativity most succinctly, she was not the only one to describe expectations of heteronormative reproduction emerging from the systems with which queer people are often surrounded, whether it be through schools, family, the workplace, the media, and institutions of many varieties. The people who shared their stories with me shared many instances of pressure to reproduce, whether implicit or explicit, because having children is considered 'normal'. For some interlocutors, they continued to receive these expectations and pressures even after revealing that they were queer.

While the recent cultural shifts have led to a more tolerant and accepting zeitgeist for queer people that has reduced some of the pressure of heteronormativity, they have instead introduced a new source of pressure by funnelling those previously hetero expectations into

homonormativity. This is at least in part because, as Brad Van Eeden-Moorefield and colleagues assert, government action and policy around accepting queer people is often done based on heteronormative frameworks and approaches (2011). They continue,

by positioning gay and lesbian couples and families as *just like* heterosexual couples and families, the ways in which these couples may have experiences different than those of heterosexual couples are made invisible and unexamined" (Van Eeden-Moorefield et al. 2011, 563, emphasis in original).

From the perspective of critical phenomenology, the irreplaceable and irreplicable experiences that queer people have when making their reproductive decisions are being erased by assumptions that, because queer people are more widely socially accepted now, they have now achieved the state of "normality" that is usually defined by heterosexuality.

I find parallels between the phenomenological perspective and the experiences of other marginalized people, such as the idea that Black people are just "normal" citizens and should be treated like everyone else after the Civil Rights movement in the United States, even though the centuries of anti-Black laws and views in the country continue to have marginalizing impacts on the lives of Black people today (Rashawn and Perry 2020). There are also parallels with public perception of Indigenous people in Canada, with the idea that since Indigenous people living on Turtle Island are now "regular" Canadians because many of the most visible colonizing institutions have been dismantled, that they should not get additional governmental, institutional, or social support because they are just like us and should therefore be treated the same as "us," i.e., White settler Canadians. This perspective ignores the particular challenges Indigenous people must face because of those centuries of colonialism and genocide, even if you ignore the fact that colonialism continues today in different guises, with the associated devastating impacts on even modern-day Indigenous people and their livelihoods and wellbeing (Yazzie 2000; Battell Lowman and Barker 2015). Assigning normative standards to marginalized people just because the current circumstances of their marginalization look different is an erasure of the important distinctions between the majority group who hold power to create norms and the marginalized groups now expected to adhere to those norms. Moreover, it ignores the unique experiences of marginalized groups and the benefits of those experiences.

Lula has noticed that the assumption that reproductive decisions for queer people are expected to mirror those of their straight counterparts has led to the invisibility of queer families, saying,

It feels like all the parents in Saskatchewan are straight, and I know they're not all, and I know, out there somewhere... I actually know some kids that have a co-parenting situation, but it feels like every child here is the child of a straight couple, where it might not feel like that as much in places we romanticize as the queer haven. You know? Like, "oh, but in New York, all the moms are lesbians," right? And I know that's just a representation thing, you don't see those stories told about SK.

Homonormativity in Saskatoon thus poses a dilemma – queer people are expected to engage in the "white picket fence" type of family creation that has been established as a norm by the cisgender heterosexual White patriarchy who hold much of the bureaucratic and political power in the city, but because those norms are so similar to those of heterosexual families, it erases the visibility of queer families. To Lula, places like New York make space for queer families to be visibly queer to the point where it almost becomes an assumption about reproducing ("in New York, all the moms are lesbians"). Like any other cultural group, queer people have long fought for the space to have their ways of life, so comparing the visible success of queer reproduction in New York to the relative invisibility in Saskatoon is disheartening for people like Lula.

Some participants experienced a difficult relationship between resisting the expectations of homo- and repro-normativity and determining how they truly wanted to make their reproductive decisions, but others have used the tension from that difficulty as a guide to making what they see as better decisions. Caitlin told me about the tough childhood they experienced due to tension between the person their parents expected them to be and the person they felt themselves to be, exacerbated by the restrictive norms they received from growing up in a conservative rural area. While their relationship with their parents and the norms to which they were exposed has improved in recent years, Caitlin has decided to use those negative experiences and the meanings they have elicited from them through reflection, self-work, and the letting go of expectations as a lens to guide their future reproductive decisions.

While I had a lot of resentment towards my parents and my rural area upbringing when I was a fresh adult, time, therapy, and perspective has allowed me to understand the importance of family – blood or chosen or however else family is defined – and that a

large part of it is the effort and attempts that people put into the relationships. My parents and I don't always see eye to eye, but my parents have mostly let go of who they expected me to be, and I've let go of who I thought my parents were. [...] [Now] I want to be an example to potential queer adopted kids and to show them that they can be who they are and not worry about it, and they can live their life and be successful and be fulfilled and be content with who they are by just being themselves, and I want to show them adult examples of that. I want to be an example not just queer kids, but all kids. I want to be an example that queer and non-binary people can live full, content, and thriving lives. I also want to be able to interact with other parents and maybe change a few minds while I'm at it. I just want to be a part of raising conscientious, caring, empathetic, thoughtful children.

Caitlin is tying their understanding of relatedness and how they want to build family regardless of the norms that dictate family structures to the meaning they attach to reproduction and the role they can play in the lives of others. By eschewing their own notions of normativity and the expectations that come with them, Caitlin has focused on the meaning they see in reproducing, that of being a positive role model for children by being a good and fulfilled person regardless of their intersectional identity.

It is arguable that the goal of raising "conscientious, caring, empathetic, thoughtful children" is a culturally mediated one arising from the settler-North-American values in which Caitlin has been born and raised and is thus still attributable to North American norms. Other cultures may have different values of how children should be raised and who they ought to become, though the individual variation in these values make for inaccurate generalizations (Tudge et al. 2000; Tamis-LeMonda et al. 2002; Wang and Tamis-Lemonda 2003). Yet as Caitlin and other Canadian-born interlocutors demonstrate, the variable and dynamic relationship between queer people in Saskatoon and the normativity surrounding their reproductive decisions can impact those decisions in serious and complex ways.

### 5.3 - Norms From Brazil

For people coming to Saskatoon after being born in other countries, their idea of norms will be influenced by the experiences, norms, and meanings they found within their originating cultures. To explore how these differences interact with their current reproductive decision-

making processes in Saskatoon, I am focusing on Morgana's story. Morgana is a DFAB non-binary person who uses both they/them and she/her pronouns, is in her mid-twenties, and she originally hails from Brazil. In our discussion about her family life and how she came to the reproductive decisions she is now making, Morgana narrates how she ran head-first into tensions around repronormativity early in life. Her life as a female child, and the assumptions that were made about her reproductive future, highlight her experiences of repronormativity.

But since I was 8 years old, some crazy teacher showed us a video of a woman giving birth. And who does that, right?? It's crazy! And I was so traumatized! I was like, what the...?! You know? Why would someone do that? And that's a crazy thing to have, so since I was 8 years old, I remember being at the family, the big Sunday family reunion, and being a kid and saying, 'I don't wanna have kids cause it's gross!' And my family, my grandma and grandparents and everybody, really, they're very religious except for my nuclear family, the rest of my family is extremely Christian, crazy Christian, almost borderline fanatics, and they were all shocked.

Exposing young children to medically graphic images of childbirth, seemingly without adequately preparing them for the experience, aligns with heteronormative and repronormative assumptions that children will grow up to be straight, will want to have children, and especially that female children are already interested in babies as they relate to their own reproduction (Martin 2009; Martin and Luke 2010).

Morgana draws a direct correlation between her exposure to this birthing video and her anti-children stance from a young age, finding that the idea of having kids was "gross" to her, disrupting the norms surrounding female-identified people and their interest in reproduction. In addition, her family's religious character influences the norms she experiences regarding her reproductive decisions, as she highlights their nature as "extreme," "crazy," and "almost borderline fanatic" Christians as the principal reason they were shocked by her disinterest in children. The colonial religion of Christianity has long been associated with reproduction and its complex place in life in many areas of the world but especially in Brazil, as has been highlighted by many eminent anthropological works like Nancy Scheper-Hughes' Death Without Weeping (1993). While Scheper-Hughes situates Christianity as a prominent aspect of the expected pregnancies and subsequent losses that women see as a part of their daily lives, Morgana resists that same Christian influence when she declares that having kids is "gross." Her family and

education system expected Morgana to be happy about the idea of giving birth and having children, so it shocked them all when she announced her clear intention not to engage with these expectations. These experiences gave Morgana the impetus to critically evaluate her understanding of the meanings behind reproductive decisions for people from many walks of life, essentially queering the lens through which she saw the "legitimized, state-sanctioned heteronormative acts of reproduction specifically through the patriarchal heteronormative family" that Weissman mentioned (2017).

As Morgana is from Brazil, her relationship with Brazilian colonialism and the complex interplay between forces of heterosexuality, Indigeneity, and family ties together to deconstruct the meanings behind reproduction in her home country. This deconstruction took many forms during our discussion. The first was noting the complex relationship between heterosexual fatherhood and familial obligations she observed within Brazil and her own family, saying, "it's very common in Brazil for dads to divorce the woman, the partner, and divorce the kids too. So when [my bio-dad]'s relationship with my mom ended, his relationship with us kind of ended too." She later described the impact of this phenomenon on her reproductive decision-making, saying,

But I do feel like adopting, because that would be, that makes sense for me. Like, in this crazy world where so many straight parents are just, you know, giving up their child, like my biological dad did a little bit, and most men in Brazil do, there are so many kids that need to be loved!

The experiences Morgana had with her biological father changed the meaning she associates with reproductive decisions, making adoption seem like a viable option because heterosexual families are not guaranteed to be loving or intact families, thus leaving a surplus of children who need to be loved, possibly by adoptive parents.

Morgana also deconstructs the meanings behind her reproductive decisions in Brazil in relation to her self-declared Whiteness by comparing her lack of desire for biological children to the reasons why Indigenous Brazilians may choose to bear children. She said,

Indigenous women want to have biological kids because the Indigenous population has been killed and it's a long-lasting genocide, so for them, there's this other side to giving birth, right? So it's about culture. But for me, as a White woman, you know, it doesn't make sense.

Both hetero- and homo-normativity within Saskatchewan (and, according to Morgana and other Brazilian interlocutors who shared their experiences with me, within Brazil) create an expectation of White reproduction, and some queer people like Morgana critically examine whether they want to proliferate Whiteness in the world while recognizing the decolonizing meanings of reproduction among Indigenous reproductive decision-makers.

But Morgana takes this critique of colonial family construction within the meanings she finds in her own reproductive decisions one step further in discussing her views on what ties families together.

So I feel like adopting for me is the most logical way to do this, and also just... Believe that family can be made from other ties than biological – my family, they think of blood, like my extended family, or my biological fathers. What are the connections, how are we connected, really, except by blood, you know? So for me, blood is just a Christian colonial idea of family.

The association between blood ties and family is common throughout the colonized world (Newman 2018), and the assumption that blood ties inherently make family was, at one point, the basis for kinship studies in anthropology, with seminal but now much-critiqued works like Lewis Henry Morgan's Systems of Consanguinity and Affinity of the Human Family (1871). This is partially because, in colonial contexts, acceptable blood ties created intergenerational legalities about owning people and property that kept ownership in the hands of the colonizers who were generally men (Newman 2018). As world-dominating cultures grew out of colonial roots, scholars and laypeople began to assume that the ties between blood relationships and social relationships were the universal basis for family-building, thus continuing the belief that, as Morgana says, the connections that make us family are in the blood (Feeley-Harnik 1999). In deconstructing and dismissing these colonial approaches to the creation of family, Morgana actively resists participating in colonialism through reproduction and, in turn, resists the related concepts of repro-, hetero-, and homo-normativity.

In deconstructing these approaches to normativity by discussing the meanings behind her reproductive decisions, Morgana is resisting the norms that are at the heart of the related concepts repro-hetero-homo-normativity. Queer people, queer movements, and queerness have often been cited as sources of resistance to the socially dominant norm (Grindstaff 2014), and part of that resistance has been the power that comes from organizing as a collective identity of

"homosexuals", "gay and lesbian", or "LGBT+" groups (Armstrong 2002). The meaning Morgana has found within her reproductive decisions aligns with familiar themes of resistance among queer groups – repronormativity, Indigeneity, and colonialism (Gosine 2010; Jackman and Upadhyay 2014; Hunt and Holmes 2015; Rowe 2017). These common themes, along with Morgana's self-identified belonging to queer collective identities (in her words: "I have a bunch of identities, I guess I could just list them! I am Brazilian, I am White, I am bisexual and non-binary, cat mom [laughs]... I don't know!"), denotes some significance to her belonging to the collectives or groups of queer people. The formation of these group identities and their meanings align with critical medical anthropology's focus on critiquing the dynamics of power and the cultural dialogues power can produce.

Shifting to the lens of critical phenomenology, Morgana's position regarding the influence of normative reproductive standards on the meanings she finds within her reproductive decision-making highlights her individuality and the necessity of critiquing and deconstructing the categories and constructs I have just used to describe her experiences. This is, in part, because my own experiences have helped me to highlight these particular statements and meanings she shared with me – as a queer person with a similar family experience to Morgana's and as a White settler who benefits from the way my country has treated Indigenous people but who wants to contribute to decolonization, it seems clear to me that these aspects of Morgana's experiences and the meanings she attributes to them are important and salient aspects of her reproductive decisions. That is because my own experiences, and the judgments I make of others, as a result, come with the dangers of what Hannah Arendt calls the insidious "veneer of certainty" (2003, 168). It is also because, as Cheryl Mattingly notes, "experience is always excessive, unruly; it has a singularity—especially if we pay close attention—that cannot be captured by explaining it in terms of underlying causes" (2019, 423). So, a critical examination of normative influences on reproduction, and my own assignation of Morgana's experiences and meanings to the categories and concepts laid out above, will hopefully help to capture my uncertainty about the connections I have assigned to the singularity of the experiences shared with me that have informed these arguments.

Morgana's experiences with normativity surrounding her reproductive decisions and the meanings she assigns them are representative of her experiences, her specific and unique context, and the way she, as a human, interprets her experiences and turns them into meaning. This is the

singularity to which Mattingly refers. Phenomenologically, Morgana's experiences may fit her into categories of identity and shared experience, but no one else will encounter her specific blend of experiences that have resulted in her reproductive decisions and the meanings they hold for her. For example, other Brazilian interlocutors in this study did not share experiences of their biological fathers leaving their families, even though academic literature suggests this is a relatively common phenomenon (Bastos et al., 2013; Vieira et al., 2013).

Indigeneity was a shared concern among interlocutors, but not from other Brazilians and not from the perspectives of decolonization. And while Christianity was a primary source of decision-making and meaning for some interlocutors, the discussion of blood making family in the Christian context was unique to Morgana. She shares many traits and experiences in common with the others who shared their stories with me, but Morgana's reproductive decisions and the meanings she finds within them are a singularity, unable to be sufficiently captured within the scope of a Master's thesis. There is an irony to discussing norms within the framework of Mattingly's approach to critical phenomenology, as norms assume group adherence and agreement to "what is good and what is desirable, [...] determining how people in the culture will behave" (Boyd and Richerson 1994). In contrast, critical phenomenology stresses the uniqueness and singularity of people, their standards of goodness and desirability, and the behaviours that are part of their experiences.

Morgana was not the only interlocutor to express these deeply complex and multifaceted relationships with the norms they experienced in their culture of origin and how they have interacted with the different norms they are now experiencing in living in Saskatoon. It is, however, difficult to summarize the experiences of those who were not born in Canada because they came from a variety of cultural backgrounds and differing experiences with their homegrown norms. Connecting threads that ran through these unique individual experiences include many facets I have highlighted from Morgana's experiences. Others discussed open experiences of homophobia, overt pressure to conform to heteronormative family structures and approaches to reproductive decision-making, and colonial pressure to adopt settler-oriented approaches to relatedness and reproduction including the privileging of blood relations and the primacy of biologically-related children.

These experiences varied based on intersectional identity, with DFAB and DMAB interlocutors sharing different experiences of hetero-, homo-, and repro-normativity, differences

along class lines and the financial expectations surrounding reproduction, and the impact of their relationships with normative expectations among their own ethnic groups and in contrast to other ethnic groups living in their countries of origin. Like Morgana, the pressures of these and further aspects of normativity fueled their desire to immigrate to Saskatoon in order to make their reproductive decisions in a climate with different and, to them, less stringent normative expectations. When considering culture within the norms that influence reproductive decision-making amongst queer people living in Saskatoon, there cannot be a singular cultural definition of Saskatoon – Morgana's experiences in Brazil are just as influential over her reproductive decisions in Saskatoon as the experiences of others like Lula, for whom Saskatoon is their culture of origin.

### 5.4 - Conclusion

When discussing a group that is in the minority, the norms that are created by, upheld by, or conformed to by a majority will generally be of interest because, in some ways, minority groups are defined by not adhering to, or having a voice in creating, those norms (Gardikiotis 2011). In Canada, queer people represent approximately 4% of the population, making the people who shared their stories with me undeniable minorities through their gender and sexual identities (Statistics Canada 2021a). How much a queer person conforms to the normative expectations surrounding their gender, sexuality, and all other aspects of their lives can have profound implications for their health, wellness, quality of life, and, for this thesis, their reproductive decisions (Browne 2019). Within my conversations with queer people in Saskatoon, three categories of norms were especially relevant – heteronormativity, homonormativity, and repronormativity. These three constructs intersected with each other and the specific contexts in which interlocutors have lived their lives to influence their reproductive decisions. Since a significant portion of interlocutors in Saskatoon were originally from other places, a division emerged in my analysis – the cultural orientation of norms in Saskatoon, and the norms found in the cultures from which interlocutors originally hailed.

Normative experiences originating in Saskatoon that influence reproductive decisionmaking include the relatively widespread background of Christianity in the province, such as the Saskatchewan government's close ties to Christianity and our publicly funded Catholic school system, as well as the institutional foundation of norms surrounding gender, sexuality, and family formation found within areas like the provincial government, the province's legal system, and the construction of cultural norms that happens within the province's educational system. The norms in Saskatchewan stemming from Christianity have a strong legacy in the provincial government's legislature, as a Christian prayer is included at the beginning of every sitting of the Legaslative Assembly, though occasionally prayers from other religions are included (White-Crumney 2018). In addition to Saskatchewan's long history of creating and supporting Christian schools, the provincial government's funding and softening of regulations of independent religious schools through a series of educational policies and amendments have contributed to a growth of heavily Christian education with relatively little institutional oversight outsight of the SaskParty by deciding who qualifies as an eligible educator (Minister of Education 2012; 2018; 2022; Watkinson 2022; Langager 2022). Saskatchewan's government and associated institutions continue to maintain strong ties with Christianity and Saskatchewan-born interlocutors have noted the impact those ties have on the province's norms surrounding queer reproduction.

In comparison, the normative influences in other cultures are non-exhaustively illustrated by Morgana's experiences in Brazil. The cultural norms related to her reproductive decision-making were different but distinct from those described by people originally from Saskatoon. Morgana's notable experiences with norms and the meanings she finds within them include assumptions about femininity and the desire for children as informed by Christianity, the far-reaching influences of colonialism on who is considered family through relationships of blood, and expectations of masculinity and the norms of fatherhood in the Brazilian family.

It is also essential to keep my bias as a White Canadian person born and raised in Saskatoon at the forefront of my analysis of norms described in different cultural contexts with which I have no direct experience. The limitation in my analysis perfectly describes the inescapable and highly formative role of norms and their influence on how we experience the world and the meanings we associate with those experiences. The norms, or "cultural phenomena that prescribe and proscribe behaviour in specific circumstances," that inform the meaning I identify in Morgana's interview responses act similarly on residents of Saskatoon and the contexts in which they make their reproductive decisions. Just as I have specific ideas about what is interesting and meaningful in the experiences shared with me, so too do Saskatonians have particular ideas and behaviours about what constitutes normal reproductive decision-making. Those norms emerged as important talking points during our interviews. Finally, a wide array of

intersecting norms that derive from multiple cultural backgrounds then intersect with the institutional influences of religion, politics, education, and colonialism go on to seriously inform the reproductive decisions being made by queer people living in Saskatoon.

### Chapter 6 – COVID-19 and Place

## 6.1 - Defining Place

The idea of place is tricky. While every place consists of location, locale, and sense of place (Cresswell 2014), the vast combinations in which these elements can be combined, and the specific meanings each element holds to each individual, make understanding place a life's work. But to narrow that immense scope for the purposes of this thesis, it is important that I consider place as it coexists and influences reproductive decision-making processes, especially as they relate to the COVID-19 pandemic.

How does a location, or "an absolute point in space with a specific set of coordinates and measurable distances from other location" (Cresswell 2009, 1), change experiences of COVID-19 and associated reproductive decisions? In broad terms, locations come with specific environmental climates — my location in Saskatoon, for example, leaves me more susceptible to COVID-19 in the winter months because our long, cold winters mean that people congregate indoors and thus introduce additional vectors of infection (del Rio, Omer, and Malani 2022; Djuric 2022). Knowing that childcare and schooling during those months present opportunities for infection is one small example demonstrating the specific concern surrounding location and the intersection of COVID-19, reproductive decision-making, and place.

Locale, or the physical shape that a place takes (Cresswell 2009), can be exemplified in the growing concerns about ventilation and its role in preventing the spread of COVID-19. The importance of adequate indoor ventilation in reducing the transmission of COVID-19 is high, especially in a world that is becoming primarily urban, meaning that most people spend their time indoors, often in fairly dense living situations (Bhagat et al. 2020; Connolly, Ali, and Keil 2020). If the locales in which a person spends the majority of their time do not provide adequate ventilation to reduce the spread of COVID-19, then that may be a place the person does not want to spend much time, a factor which may include their decisions about bringing a child into that same locale. Apartment buildings, schools, daycare centres, and stores all become locales in Saskatoon that could factor into the reproductive decision-making process.

Last, but perhaps most important, is the sense of place – essentially, the phenomenology of a place. What meaning does a location have for someone, what are their attachments to it, and what is their relationship like with that place? These are broad questions and, though narrower when considered in relation to experiences with the place of Saskatoon during the COVID-19

pandemic and its interaction with reproductive decision-making, remain large enough that they were at the centre of many of the participants who shared their stories with me. Yet, as the breadth of the question suggests, I did not receive a uniform answer about how their sense of place in Saskatoon and the impacts of COVID-19 were factored into reproductive decision-making processes. Ultimately, the meaning we attach to a location or space define it as a place, and this chapter includes an exploration of the individual meanings, and thus places, that are important to those people who shared their reproductive decision-making processes with me.

Having broken "place" into its constituent parts of location, locale, and sense of place makes it seem as though these are distinct and separate parts of place, but as Margaret Rodman points out, an anthropological approach to the idea of place involves unifying these aspects as essential to fully understanding the meaning of place (1992). Her observations on the significance of place and its meanings fit into Mattingly's critical phenomenology 2.0, as Rodman says that places are "local and multiple. For each inhabitant, a place has a unique reality, one in which meaning is shared with other people and places. The links in these chains of experiences places are forged of culture and history" (1992, 643). The meaning of place is therefore unique to each person living within that place, but those meanings are connected to the meanings others find within and attribute to both that place and others, shared and influenced across time and across experiences. These meanings and experiences can be understood more deeply by looking at them through the lenses of location, locale, and sense of place, but ultimately they must be understood by looking at these and other aspects as a whole.

Places are created by the people that live in those places or, as Nancy Munn states, in the "ongoing formation [of place] in certain experiential syntheses that actors create in practices, and the events that transpire in their terms" (1990, 2). The people who have experiences and events within a location, locale, or even a sense of place are the people who turn that location, that locale, or even that sense into what we recognize as a place. Those experiences and events may be individual, but it is the connections between them that create a place. Place is a concept that belongs to and is created by those who are creating place by living it (Williams 1977). Understanding who makes place should give way to who defines and controls it because place, as with many other aspects of critical anthropology, is indelibly linked to questions of power and agency. As I have discussed in this thesis, power and agency are vital aspects of queer reproductive decision-making in Saskatoon, such as the power to legally and institutionally

define families and who is allowed to reproduce based on the place those people live and the agency to move between countries to fulfill reproductive intentions.

Within contemporary anthropological conceptions of place, movement between places, and thus the expansion of what place means, is also a key concern. Understanding place as it relates to movement is important because, as Sally Ward argues, movement "is the norm rather than the exception of human experience in the world today" if the term 'movement' is understood to include not just physically relocating but also the effects of "electronic communication, satellite technology, economic and environmental catastrophe and political turmoil" (2003, 81, 93). Movement contributes to the phenomelogical nature of place because, no matter where an individual goes, they are in a place because their existence creates the place as it relates to the meanings that person applies to and creates from their experiences with that place. Anthropologically speaking, movement may always be an aspect of place because, as Edward Casey notes, "there is nothing like a completely static place, a place involving no movement, no change, no transiency" (1993, 289). Moving within places, moving between places, and even movement of the places themselves will always be an inherent aspect of the human, and cultural, experiences of place.

We create the meanings of a place based on how we move within it. Consider the cultural meanings attached to public transportation in pedestrian-heavy places like Tokyo in comparison to the car-centric cultural meanings of places like rural Saskatchewan. I have already discussed some of the meanings in moving between places, such as moving away from a lack of physical safety in Brazil to what people perceive to be as the safer location of Canada. And the places we are in are always moving and that movement changes what places mean to us. For example, as the planet rotates and produces seasons, the meanings people associate with the city of Saskatoon change from summer to winter, as I discuss later in this chapter. Saskatoon may stay in the same spot on the planet, but its movement in space and in relation to the sun changes what the city means to the people who create the place of Saskatoon by living it. These meanings and the resultant definitions of place are also deeply rooted in the cultures of and produced by the people living and creating these places – a Saskatonian in Tokyo may assign a completely different meaning to their public transit system, Brazilians may define their safety in relation to place differently than Canadians do, and the existence of snowbirds, or people who choose to spend their winters in warmer climates rather than stay in snowy Saskatoon, demonstrates a very

different meaning of place based on how Saskatoon is moving throughout the year than that held by people who enjoy and revel in the winter aspects of life in Saskatoon.

Place can be understood by breaking it down into aspects that benefit from further scrutiny, such as location, locale, and sense of place, but from an anthropological standpoint, these aspects must be unified to truly understand what place can be. Place is created by those living it and it is a phenomenological thing because the meaning of a place cannot be divorced from the place itself. Movement, a central concern of this thesis, is also an intrinsic part of place. The role of culture must also be highlighted as it interacts with individual experiences and the meanings of place produced by those experiences, then linked to how others have also experienced and created meaning about their places. Place, like most anthropological considerations, is more than the sum of its parts. Rodman points out that "the meaning of place often seems to go without saying" (1992, 640) but instead I find that, because of its complex and holistic nature, there is a neverending amount of things to say about place. For the purposes of brevity and this thesis, I will keep my exploration of place limited to queer reproductive decision-making in Saskatoon during the COVID-19 pandemic.

## 6.2 - How COVID Has Changed Meanings of Place

To understand how COVID-19 has changed the meanings of place in relation to reproductive decisions, I first consider how place ties to notions of family, especially in the queer context. I return to Ethan's family and how they have built their family in relation to ideas of place. Ethan's family has more than two parents involved because he and his wife have chosen to co-parent with the sperm donor who helped to create most of their children, and thus their children live in multiple households within the city. Normative perceptions of families, especially immediate family such as parents and children, often locate the family as residing within the same household and a common definition of family is the "household," relegating family to one specific location, a shared house, and assigning particular meanings to "family" that are dependent on the place of that one shared house (Canetto 1996). Families spread across multiple households take on a different relationship with space and place, as outlined by Klas Borell, who said,

a common feature for contemporary family life is not just a more open and more flexible concept of the family, but also a significant spatial flexibility. A household *no* 

*longer* necessarily embraces what people with mutual commitments define as 'their family'. Family ties have a tendency to extend beyond the boundaries of a household which leads to the creation of multi-household families (2003, 468, emphasis in original).

As Borell points out, while many families live in one household, the norm that families are the same as one household has not represented reality for a long time, or perhaps ever. For example, divorce and the separate households that often result from divorce have existed in Canada since at least the late 1800s (Library and Archives Canada 2021). Indigenous Peoples living on Turtle Island also have families that skew this idea of households, with multiple families living within one household or having immediate family members spread out across multiple households (Tam, Findlay, and Kohen 2017). And queer people have long been making families that do not share one household, due in part to the legal barriers we face in creating a familial household and partially because our experiences have led to developing our own ways of doing and being that align with queer values and experiences (Kelly 2013; Epstein 2005). The assignation of the family to one unified household is a restrictive way of thinking about all families, including queer ones, because it oversimplifies complex inter- and intra-personal dynamics (Sharma 2013). Queer people must navigate these assumptions around place and family in their reproductive decision-making, as they have the potential to seriously impact their lives and their families.

An example of how the relationship between place and family, and assumptions about that relationship, can impact queer families is seen in the COVID-19 pandemic. The restrictions enacted in Saskatoon to reduce the spread of COVID-19 changed the meaning of a multi-household family for Ethan, his fellow co-parents, and his children because the idea of family being synonymous with household is problematized by interactions with the social response to COVID-19. The public health measures put in place in Canada to deal with COVID-19 did not accommodate the spatial distribution of households, instead creating blanket rules for a perceived majority of families that embody family as one household in one locale (Public Health Agency of Canada 2022). This is not necessarily unreasonable. Wide-scale emergency health measures cannot always be finely nuanced, but the restrictions here were some of the strictest in the world and marginalized groups paid an unequal cost for that stringency (Razak et al. 2022; Mykhalovskiy et al. 2020; Spadafora et al. 2022). Public health measures, such as the restrictions accompanying the COVID-19 pandemic, added burden to families spread across multiple

households and locales, especially impacting queer people and their families (Brennan et al. 2020; Kay 2020; Reid and Ritholtz 2020). This was the case for Ethan and his multi-household family, as he said, "There's been a lot of restricted visiting. I've only seen my oldest daughter a handful of times in the last year." Ethan and his family have chosen to create multiple households that co-parent together, with all of the children in those households being regarded as siblings. When COVID-19 restrictions and distancing measures came into effect, Ethan and his family had to choose who would be in each household-oriented safety bubble, thus being distanced from the children who were not living in Ethan's household.

Another aspect of the household-family association presented was that, when COVID-19 began, Ethan's wife was a care worker and their sperm donor co-parent worked for the city in a public facility, so both would have to quarantine quite regularly due to COVID-19 exposures in the workplace. These quarantines disrupted the ability to function as a multi-household family by further separating the children and co-parents for the duration of the quarantines, which, as Ethan mentioned, "affected which children could be in which homes and how we could visit appropriately." Parents being separated from their children during COVID-19 quarantine and social distancing had detrimental effects on mental health for those families and can even have long-term impacts on their children's attachment style and other aspects of psychological and relational development (Singh et al. 2020; Fore 2020; Jiao et al. 2020). Also, siblings being separated from each other can contribute to distress and familial instability (Drapeau et al. 2000).

In this way, the COVID-19 pandemic magnified the distance between some family members – for Ethan and his kids, they lived in the same city but the distance between the places they were became so serious as to be impossible to traverse safely due to the risks it would introduce to the health, well-being, and financial safety of their family. While the distance never changed in the sense of metres, the meaning of that distance changed drastically, impacting the sense of place and relationship between the multiple households that make up Ethan's family. The multi-household family and the unique meaning of place that Ethan and his co-parents have developed demonstrates how the COVID-19 pandemic influenced some families and their decisions in Saskatoon. Because of how the city of Saskatoon defined families in association with place, i.e., as a single household, non-mainstream family creation and maintenance was interrupted for some by pandemic protection procedures. Deciding whether or not to create a family, or add new members to a family, must take into account the potential for such

interruptions to happen again in the future. For the queer people who shared their experiences with me, the ability to maintain connections between multi-household families was an important variable in making reproductive decisions.

The existence of the virtual world, and the increasing availability and ubiquity of remote connection software like Zoom, WebEx and Facetime arising from the COVID-19 pandemic, have impacted conceptions of place though, as Sally Ward pointed out, these electronic 'movements' are not new to concerns surrounding place (2003). These types of software compress both space and time, allowing people to meet from across the world and jump between groups much faster than possible in the physical world, and they facilitate interconnection between people or groups who might not otherwise connect (Hay 2020). While this software is not new, as Skype launched in 2003 and Apple's FaceTime began in 2010, their popularity and necessity skyrocketed due to worldwide restrictions and distancing measures enacted in response to the COVID-19 pandemic. With a world suddenly reliant on virtual connection technology, formerly insurmountable distances have been significantly reduced, making relationship-building between some people in different areas, cities, provinces, and countries far more sustainable. For Claudia, this reduced distance has helped her stay connected and engaged with friends and family from her home country of Brazil. Hopping onto a Zoom call with her friends to play games gave Claudia an accessible way to relax with friends and stave off the isolation many people felt during the COVID-19 pandemic. Claudia's complicated relationship with her biological family means that her friends are one of the most important components of her chosen family that help her feel prepared to have children. When I asked her what a perfect world would look like for her to have kids, she told me how important it was to have her family and community around her and that, "the thing is, my friends, I know some people who think of friends as just, oh they're people that I love a lot but not family. I mean, they are literally my family". Virtual connection software and its ability to minimize distance between places to keep Claudia and her family of friends connected provides key support in her reproductive decisionmaking.

Lisa also spoke about the ubiquity of these types of technologies and their positive impact on the way families are forming and keeping in touch with each other, saying,

I feel like that might be a positive change, after it's no longer serious and they can see the people close to them in person and stuff, but it might be family that's far away, more of a possibility. And people know that there is options for them to have that relationship with kids and stuff long-distance.

By collapsing the physical distance between a person and their important family members, people like Lisa may now consider having children when they would have otherwise been opposed to it because the importance of the children knowing their extended family and the support that family can provide was an essential aspect of their reproductive decision-making. The proximity to family, whether in person or through digital connections, is an important aspect of Lisa's deliberations about her reproductive decisions. She described her desire to have children who know the members of their family.

I definitely would think that I would be grateful to have my family so close by. And yeah, it's definitely a large support system, I have lots of extended family in SK. And my family is very close, even the extended family and stuff, so that would be huge. For the support and also for the fact that my kids would get to know their families.

Now that potential children can grow up knowing their relatives in far-away places with similar ease of knowing those who are local, the load of responsibility that comes with having children can be somewhat mitigated through at least the emotional and relational support offered by relatives who can connect via calling technology (Karki, Dhonju, and Kunwar 2020; Hall, Pennington, and Holmstrom 2021).

The collapsing of place and time provided by remote call technology also presented drawbacks for a few people. As COVID-19 forced schooling for young people to rapidly shift online, parents were suddenly finding themselves in a classroom, office, and living space compressed into one house or, in some cases, one room. Negotiating this new meaning of place, with people like colleagues and fellow students, with whom interactions usually happen in public or work-related spaces, suddenly being invited into private at-home spaces through remote calling technology, and with those spaces now having multiple functions that they were not originally intended to have, privacy and autonomy at home became a difficult issue for many people (Katsabian 2020; Fukumura et al. 2021; Aznar et al. 2021). In addition to these concerns, parents were now having to hold multiple roles that they were not prepared to hold – teaching assistant, IT support, and caregiver roles were suddenly happening in the same time and space as their pre-existing work or family roles (Michelson et al. 2021). The chaos and stress presented by this collapsing of the work-home-school bubble into one place was mentioned by most of the

people who shared their stories with me as a reason they were hesitant about having children. Jane, a lesbian woman who is planning to adopt children with her partner, spoke to this perspective by observing her friends who have children.

Seeing my friends who have small children and they're trying to juggle working from home while their kids aren't in school or their daycares are closed, they have to try to be a full-time employee and a full-time parent, and even with a supportive workplace...

You can't tell a 3-year-old to go entertain themselves for 8 hours.

Knowing now how institutions responded to emergency pandemic situations and that other pandemics of this scale are possible or even likely in the future, some people are unsure if they would want to take on the burden with which they saw peers and family members struggling. For people like Mariana, that burden is especially daunting because, as a DFAB person, she is more likely to take on additional work in domestic spaces, is more likely to have had to leave jobs and have her career trajectory impacted, and is more likely to have been balancing front-line jobs with higher levels of stress and COVID-19 exposure potential with the increased work found in the household (Carli 2020; Zamarro, Perez-Arce, and Jose Prados 2020; Burki 2020). All of these issues that are related to the collapsing of many places into the domestic one are part of the reason why Mariana said, "parenting during this pandemic sounds terrifying, working from home and all." As someone who greatly values their independence as a person, a scholar, and a feminist, Mariana considered the place made by COVID-19 and the use of remote call technology to be a significant factor in her reproductive decision-making.

The combination of many digital places into one physical place, the home, was not a problem for everyone, though no one mentioned any positive benefits directly related to their reproductive decision-making. Jason, a bisexual man in a serious relationship with a woman, explained some of the benefits he found in this new conception of place as influenced by the COVID-19 pandemic. He found it easier to manage his workload because removing the constraints of office hours, in-person class times, and travel time to campus and the office allowed him helpful freedom and flexibility in conducting his work.

The online system was difficult for a lot of people, but I think that I actually excelled with the online format. I think that my experience would have been better in person, but because it was a lot easier to set my own hours to work on stuff because it was all asynchronous, so I was able to factor in my work schedule a little bit easier, as well as...

[pause] I'm trying to get better, but I'm a big night owl, so I would do a lot of my stuff at night, and instead of being tired in the morning for class like I would have been on campus, I wasn't.

By collapsing the places of both school and home into the same location, Jason no longer had to abide by the in-person strictures of his usual university experiences and could instead capitalize on this newfound flexibility work when and where best suited him.

This was not the only unexpected bonus to how pandemic responses changed Jason's relationship with place. He also found that working and schooling from home meant they were in much closer proximity to his partner for much more time than usual, which allowed them to grow closer and deepen their relationship.

I feel like I've grown a lot closer with my partner. I feel like that's something that is a big change. I definitely went from just kind of like, you know, kinda... [pause] Thinking about our future and an engagement-type deal, to actively now wanting to get that done because we have never spent as much time as we have together as during this, because our schedules are very similar, er, sorry, yeah, schedules very similar. I'm home around the same time as she is, which never used to happen cause we'd have opposite class schedules. We've done classes in the same room so we've gotten a lot closer.

Spending more time together because most of their lives were being carried out in the place of home changed the context of their relationship, allowing Jason to see his relationship priorities in a different light. These benefits could be tangentially impactful on some peoples' reproductive decision-making, as flexible working arrangements and increased positive romantic relationships both carry the possibility of either changing or reinforcing reproductive decision-making, but no one who shared their stories with me tied these aspects to their reproductive decisions.

## 6.3 - Creating New Meanings of the Space Between Places

It is not only the places that have changed meanings during the COVID-19 pandemic but also the more ephemeral space between places. Philosophically, this is slightly trickier to discuss as every space between places is its own place, as Lisa Findley notes when describing paths, saying,

When we move from one place to another we do so along a path. In this sense a path is simply a route that we follow, whether it's visual, kinesthetic, or psychological

character. However a 'path' as either an idea or a physical space, can be a profound experience of movement and place (1993, 72).

For the participants who have shared their experiences with me, the spaces, or paths, in between the places they occupy play important roles in creating and explaining meaning within their reproductive decision-making processes. How the COVID-19 pandemic has impacted the meanings of those spaces is also significant. As previously mentioned, many people who shared their stories with me have moved here from other countries. For these people, the paths they have taken from their countries of origin to Saskatoon, including their experiences on those paths, are an influential part of their reproductive decision-making. For Juan, moving to Saskatoon complicated the path to his family. After leaving his home country of Colombia, Juan first lived in the United States of America for less than two years to attend university there. While in the USA, Juan's relationship with his family was easier to navigate – COVID had not yet come on the scene, and he had an easier time seeing his family as they could easily meet in Miami, which was halfway between where he lived and where they lived. Now, living in Saskatoon, Juan points to the ten-hour flight he or his family would have to take to visit each other, as well as the at-the-time strict restrictions on travelling due to COVID, as a part of why he felt removed from his family. In considering his reproductive decisions, Juan mentioned that an important aspect of those decisions was the support of his family, so the physical distance between them made him feel less confident about the possibility of having a child.

The physical path separating Juan and his family became arduous to overcome and impacted his reproductive decisions, but it was not the only aspect of the path between him and his family to be impactful. As a gay man, Juan found Colombia to be an imperfect place to live, as homophobia and marginalization of queer people were still prevalent in his experiences. After leaving Colombia, he further explored his sexuality in post-secondary schools, first in the United States and then in Saskatoon. His relationship with his sexuality has developed thus, and he named his boyfriend one of the most important people in his life. His sexual orientation came up again when I asked him about important factors in his reproductive decision-making and he replied, "I would say probably having my family supporting as who I am, because right now I haven't come out to my family yet because I was holding on to it. Having a family that supports me and my partner as well, and my kid, of course." He went on to explain that "the ways that my parents raised me, [...] they were never very supportive of the LGBT community either." The

importance Juan places on the acceptance and support of his family is complicated by their views on queer people and their lack of knowledge about Juan's sexual orientation. This complexity is part of the path Juan is still on between Colombia and Saskatoon, a path he has physically traversed but is still mentally and emotionally navigating.

At a physical distance, Juan can live in a place that is more supportive of his identity and his choices, including the possibility of starting a family with another man. He misses the closeness with his family but enjoys the freedom to explore himself. The emotional distance that exists between their lack of knowledge about Juan's sexual orientation and his desire for their support and acceptance of his potential family is more nebulous and less easily traversed. The distance between these places is mirrored in the distance between Juan's current state of uncertainty and his future reproductive decisions. He is unsure whether he wants to have a family, weighing the pros and cons, partly because he is still walking the path between his family in Colombia and his future as a gay man. Where that path will take him is not clear, but what is clear is that the meaning of his journey on the path plays an important and formative role in his reproductive decision-making. While Juan was not the only person to discuss this ill-defined path between their country of origin and their reproductive present in Saskatoon, his story illustrates a shared thread among many of those who shared their story with me – traversing physical distances is often mirrored by the phenomenological journey included in moving from one place to another, one approach to queerness to another, one type of relationship to another.

For those who moved away from places that are less friendly to queers, the path includes self-exploration and a renegotiation of relationships with people who are still at a distance. For those who moved to attend university and gain new financial opportunities, the path contains the difference between institutional systems, standards of living, and the life one could possibly live as they make their reproductive decisions. For many queer people experiencing the COVID-19 pandemic, a broad range of variables have and continue to change the distance between their place of origin and their current and future reproductive decisions in Saskatoon. These variables can be physical, such as the interruption of travel and the closure of borders, and they can be phenomenological, like the distance between relationships with queer people "back home" and those found in Saskatoon. Regardless of their origin, it is undeniable that some of the queer people who shared their experiences with me discovered that COVID-19 changed the space between places and thus impacted their reproductive decision-making.

## 6.4 - COVID-19 and Reproduction

In some cases, the restrictions, quarantines, and other measures of dealing with COVID-19 have highlighted an important aspect of place as being the sense of place. This aligns with Mattingly's critical phenomenology because sense of place is intimately associated with what a place means and how individuals are experiencing and interpreting their relationships with the places they create. In this section, I explore how the same place, Saskatoon, is experienced and interpreted by those who have shared their stories with me. This exploration is an illustration of the way that place, and specifically the phenomenology of place, is culturally, contextually, and individually constructed through perspective and experience. This complicated relationship with place, shaped by the COVID-19 pandemic and human actions surrounding it, also has important implications for reproductive decision-making.

More than half of the participants shared how the COVID-19 experience changed their relationships with place through a shift in the experience of both location and locale, in many cases relocating places to the home, and emphasizing the sense of place associated with their living quarters. Carolina's story illustrates the complex interconnections between multiple aspects of life, sense of place, COVID-19, and reproductive decision-making. For Carolina, the social response to COVID-19 interrupted her sense of place as it relates to her locale and location within Saskatoon. Originally from Calgary, Carolina explained how the COVID-19 responses turned Saskatoon into just any other place.

But then with COVID, it's mostly just being at home now, so I feel like where I am doesn't really matter as much right now, just because where I am is in a house and there is very little connection to the outside city. [...] And the places that I go to now, when I leave the house, are generic. Like, I go to the grocery store, I go to the pet store, I don't really go to places that I feel are the heart of Saskatoon. Especially with the summer, the one summer that I actually got to spend in Saskatoon, I went to the street festivals all the time and being outside and by the river and all of that, and felt very much in the city. And with those not going on, it doesn't feel like there's as much of a connection to the things that make Saskatoon different from anywhere else. [...] I just think about this past year and I'm like, I don't think it would have made much of a difference if I'd been in Calgary instead.

Without the events and cultural activities Carolina sees as the heart of Saskatoon, her location is interchangeable with comparable locations (in this instance, Calgary and Saskatoon) and comes to matter little because Carolina had "little connection to the outside city." Instead, the focus shifts to her locale ("where I am is in a house") and her sense of place (for her, Saskatoon is heavily associated with street festivals and being by the river). Carolina's experiences align with the experiences of many others worldwide – as mobility became heavily restricted, so too did our sense of place become restricted.

Human geography scholar David Bissell describes an experience in Australia that contains many of the same themes as Carolina's in his essay reflecting on sense of place and the COVID-19 pandemic.

This interconnected sense of place is so true when we think of what makes many of the places we value as vibrant and energising places to be. Walking along a street in the inner north of Melbourne, you might glimpse a billboard advertising a gig for a band based in Los Angeles and an orchestra visiting from Taiwan. A businesswoman just flown in from interstate gets out of a taxi, pulling a small suitcase. [...] You wait to cross the road, and you notice a passing mini bus that may be taking tourists to the Great Ocean Road. You decide to pop into your usual coffee shop, and the barista you have known for a few years updates you on his mother's health in Ahmedabad in India. Little wonder that, for Doreen Massey (1991), place is not a static and unchanging object, but a process formed through all kinds of globally interrelated movements coming together (Bissell 2021, 151; Massey 1991).

Bissell's description of the multicultural and multispatial mobility that makes a place feel "vibrant and energising" overlaps with Carolina's description of what makes Saskatoon feel like Saskatoon and not just another place for a house to be situated. Festivals alongside the South Saskatchewan River dominate much of Saskatoon's social life in the summertime; The Saskatchewan Jazz Festival, the Saskatoon Fringe Festival, FolkFest, Taste of Saskatchewan, Pets in the Park, the Children's Festival, Shakespeare on the Saskatchewan, and many others give the city its distinct social flavour based on the social experiences people have there. The mixing and interacting with people from all over the province, country, and sometimes the globe all contribute to a specific sense of place and sociality within Saskatoon – they "bring the city together" (Barrow 2022) and contribute to a sense of normalcy in the city (Dahlseide 2022).

Festivals like these are experiential, and the memories of those experiences can be important to people's sense of place (Katczynski, Stratford, and Marsh 2022).

When these events were paused or seriously reduced in 2020 and 2021, it impacted the sense of place some Saskatoon residents had developed. Without these distinctive events that give Saskatoon its social flavour, alongside the difficulty of seeing friends and family amid restrictions and social bubbles, Carolina's sense of the city was reduced to the daily humdrum of subsistence shopping. As Bissell points out, daily activities like shopping can be full of opportunities for sociality through mobility between shoppers, store employees, and the social dynamics that connect them, but COVID-19 prevention efforts curtailed social interaction so heavily as to remove the little Saskatonian flavour left in those interactions. Without mobility, particularly social mobility, people like Carolina experienced a very different sense of place that left Saskatoon feeling like it could be anywhere else.

The changed sense of place in Saskatoon because of the reduced mobility imposed by COVID-19 restrictions played a formative role in Carolina's reproductive decision-making. As discussed in the Queer Family Structures chapter, Carolina has made a family with her best friend and her best friend's partner. Carolina and her best friend had already made the move from Calgary to Saskatoon together, choosing to attend post-secondary in the same place. Her best friend and their partner began dating just a few months before the COVID-19 pandemic began.

Their partner ended up moving to Saskatoon in November of 2019, October maybe, so there were three or four months there where my friend and their partner were really close, obviously, and I was kinda just like, every once in a while I'd hang out with them, and then COVID hit. And my friend's partner was living across the street from us, and was living alone. So we kind of just ended up creating this three-person household and I think, because of the pressures of social distancing from everyone else, it became really easy for us to create this sort of family unit when everything else about the pressure of society and the way everything else is supposed to work was kind of just removed all of a sudden. So it was really easy to be like, we're just gonna create this little family.

A lack of mobility, especially social mobility, during the height of the COVID-19 restrictions, allowed the familial relationship between Carolina, her best friend, and her best friend's partner

to blossom and come to fruition for multiple reasons. A lack of in-person socializing with others, and a lack of ability to leave the house much, allowed the three of them to focus their relational energy on each other to grow their family. A lack of mobility also meant that the three of them faced less of the potential scrutiny that their atypical family might otherwise have invited as they were able to grow their family outside of the watchful eyes and judging comments of people with different approaches to family creation.

With the disconnect from location (the coordinates at which a place exists) and sense of place that was precipitated by the COVID-19 restrictions came a disconnect from social surveillance and the pressures of performing gender and sexual orientation in cisheteronormative ways, providing space for queer people to explore themselves and their relationship with the world in relative privacy and with relative ease (Quathamer and Joy 2022). The resulting family that Carolina has built with these two people forms an important foundation for her reproductive decision-making. In June of 2022, Carolina, her best friend and her best friend's partner, relocated to Toronto to continue their post-secondary education and further solidify the familial bonds from which Carolina contemplates her reproductive future. As I detailed in the Queer Family Structures chapter, while Carolina is currently unsure about whether she will herself have children, she looks forward to the potential of adding children to their family in whatever configuration makes the most sense for them. This important family and the security Carolina enjoys in her relationship with them may not have happened were it not for the changed sense of place and reduced mobility they experienced during the COVID-19 pandemic. The queer people in Saskatoon who shared their experiences with me are making their reproductive decisions at the intersection of queerness, sense of place, COVID-19, and the possibility of reproducing.

#### 6.5 – Conclusion

The global COVID-19 pandemic has impacted the meanings of many things in life for many people, and meaning-making has been a strong focus of much academic research on the pandemic (Walsh 2020; de Jong, Ziegler, and Schippers 2020; Karataş and Tagay 2021). This chapter has focused specifically on the association between place and COVID-19 as it impacts queer reproductive decision-making in Saskatoon because place deeply affects our experiences and meanings of illness in general and pandemics in particular (Fletcher 2006; Årestedt et al. 2016; Devine-Wright et al. 2020; Cowden, Counted, and Ramkissoon 2021). What exactly place

means and how people navigate it is ephemeral and individually defined, so the meanings people associate with it and draw from it are equally unique and transitory. Yet these phenomena are also social and cultural, connecting the meanings and experiences of many people in the creation and understandings of place and its meaning.

In this chapter, I have begun to address how some queer peoples' relationships with place and this pandemic have impacted their reproductive decisions, but I have found that those relationships are often based on how the place these people are in impacts their relationships with other people. Familiar places suddenly becoming distant or strange due to the social response to the pandemic, including restrictions and social distancing, can change the meanings of support and connection queer people feel with their families and community. Since support and connection are key aspects of reproductive decision-making, these changed relationships with place and meaning have impacted how some queer people view reproduction (Sodik and Nzilibili 2017; Alvarez 2018; Schaffnit and Sear 2017). In addition, the space between places, including travel and the virtual world, were key aspects of the COVID-19 experience for many people and changed the meaning many who shared their stories with me found in their reproductive decisions (Fatmi 2020; Fatmi, Thirkell, and Hossain 2021; Anifandis et al. 2020; Orlando et al. 2021). Finally, the way in which the COVID-19 pandemic changed how people could interact with others had variable impacts on how queer people could explore and build their queerness in relation to family, with some experiencing isolation while others described freedom from heteronormative judgment in creating their ideal families (Pohshna 2020; Anderson and Knee 2021).

Ultimately, relationships between place, the pandemic, and queer reproductive decision-making proved to be important enough that they emerged in many of the discussions I had with some those who have chosen to share their stories with me. I was surprised at this revelation as nothing in my research preparation had signalled that this would be a significant aspect of the decision-making process, but through the inductive research process I have followed, I now see why it is so important. COVID-19 has unsettled many aspects of some peoples' lives by reifying others, and the relationship between place and queer reproductive decision-making is no different. Those who have shared their experiences with me navigated the change (and non-changes) they experienced in the COVID-19 pandemic by making and discovering meaning in their relationship with place and reproduction as the pandemic simultaneously pushed some

people apart and brought them closer together. By examining those meanings, I hoped to highlight how queer people make their reproductive decisions and what those decisions may mean for them, their places, and the world beyond.

### Chapter 7 - Conclusion

## 7.1 – Vignette

Justin is sitting in front of a metal rack filled with plants, a white grow light bouncing highlights off their glossy green leaves. Justin is leaning forward in his chair, towards the webcam, almost as if he is trying to lean through it and out of my monitor so we can chat in person. He is telling me about the things that are important to him in making his reproductive decisions, and now in the conclusion of this thesis, they are familiar themes. He talks about the "pre-apocalyptical" state of the world, saying:

I'm terrified of the idea of not being stable or well enough, or having enough assets to make sure my child would have a good life, or also the greater condition the world's going to be in... Like, is having kids in ten or fifteen years even a good place to bring somebody up when there's going to be all these additional stressors, how global affairs are going to be going...

Within this brief description, Justin outlines two broad categories of concerns impacting his reproductive decisions. One is what I have broadly termed "the state of the world," referring to the combined factors of climate change, global inequality, systemic racism, misogyny, homophobia, and other facets of unrest, a concept I have explored in chapter three, section four. The other broad category ties to the first in some intricate ways but is also worth considering separately. This category is his own personal context, his experiences, and the meanings he has derived from those experiences. To understand these aspects, it is important to understand a little more about him. His concerns about global affairs are tied directly to concerns about his parental capabilities, informed by his childhood experiences.

Justin grew up as one of the first Chinese adoptees to Saskatoon, brought over when he was ten months old by a loving family. He describes turbulent teen years marked with struggles with his mental wellness, partially due to the relative isolation he experienced as the sole Asian person in his school and partially due to realizing his trans-ness after puberty. Therefore, trying to answer the question I asked him, how he feels about being a parent, is a negotiation for Justin. On the one hand, he wonders, "should I take on that responsibility when there's going to be that additional stress in my life and everybody's lives?" On the other, he tells me that if he were to overcome these concerns and have children, he would adopt because of the difficulties he would

experience in trying to become pregnant because of the physical and hormonal changes he has experienced in his transgender journey. Even then, having been adopted himself adds another layer of complexity to his thoughts. As he explains to me:

Sometimes I have these nice ideas of, 'Oh if I had a baby it could have my hair' or something like that, but I really don't care about that. I always wondered what my parents look like, it was always kind of nice to think about having somebody else who looked like you, but in the grand scheme of things it's really not that important to me. It's moreso the almost fetishized idea about what it's like to have perfect children who look like you and act like you and behave well all the time.

Negotiating the relationship between a family of origin, Justin's birth parents in China and his adoptive family in Saskatoon is a negotiation with himself. What aspects of who he is, both socially and biologically, take precedent in making his reproductive decisions? He recognizes the social forces at play behind some of his thinking, pointing out the socially "fetishized" idea about children being copies of their parents, yet also sometimes feels the appeal of having children who share some of his biological traits, connections to a personal history in China of which he has little information.

How then does Justin envision a queer family structure that includes reproducing as he must as a trans man who identifies as grey-ace? How does he envision reproducing his culture and family identity within the framework of adoption rather than living with the family that birthed him? Those decisions must be made in conjunction with how he feels about his place in a global society, one where the very real and pressing issues of climate change, systemic inequality, and global pandemics like COVID-19 have very real but very different impacts on people depending on their place within society as a whole. He is considering how safe he feels within this global society, how safe he feels to reproduce, and how safe he assumes his possible children would be. He is also considering his relationship with place, as the distance between who he is because of the bio-geographical nature of his origins and who he is as a result of his parents bringing him to live in Saskatoon, a place with a relatively small Asian population and, at the time of his upbringing, relatively little racial diversity within his peer groups. I asked him how he feels about being a parent, and after negotiating through these complex and challenging aspects of the question, he answers that the 2021 version of Justin, who is 25 years old, is not yet ready to make that decision: "I think I would like to have kids if I knew that I'd be stable enough

and the world would be stable enough to raise a kid in my 30s, but I think right now, it's hard to say." Given the complexities he has described, his apprehension in deciding about his parenthood status is more than understandable. Surrounded by greenery, Justin is living with his reproductive decisions, which is currently the state of not having made a decision.

While Justin's experiences fit into the areas of concern I have described in this thesis – the queer family structures, the concerns about safety, and the question of the relationship between COVID-19 and place – the reasons for those experiences and his context are incredibly unique. The nexus between his queerness as a transman and grey-ace, his ethnicity and adoptive status in relation to living in Saskatoon, and his concerns about the state of the world, mean that Justin's position within his reproductive decision-making does not easily map to anthropological theories of kinship, of queerness, and of global citizenship. I believe this is because no individual experience can fit neatly into theories intended to explore broader patterns and group experiences that tend to form concepts of culture. Individual lived experiences of kinship are often both mundane and complex, unique and consistent, and central to cultures and removed from cultural experiences. Kinship experiences like Justin's can be about, among other things, how families are *formed*, what family *is*, what families *do*, what families *mean*, and how families *feel* (Carsten 2004; Bourdieu 1990; Yan 2001; Peletz 2001; Bryant 2002; Faubion 2001; Stone 2002).

While considering his answers in conjunction with those of others who have shared their stories with me leads to patterns and themes emerging, the individuality of his experiences and what they mean to him map overlap with Mattingly's critical phenomenology 2.0 by challenging my assumptions about reproduction among queer people and how they experience reproduction in Saskatoon. Such is the case with those who spoke with me – while connections between them might exist, each person's experiences and the meanings they find in those experiences are always unique, as Jan Patrick Heiss and Albert Piette point out when discussing the work of Gottfried Wilhelm Leibniz - "the emphasis on human beings as products of various liaisons is tempered by a conception of human beings as unique products that are irreplaceable in the play of connections" (2015, 14). And while I see this uniqueness as a strength of Mattingly's critical phenomenology and my work, it is also a part of the limitations of my research.

### 7.2 – Limitations

The main limitation of my research is the voices I have not included. Missing from the pool of people who decided to share their stories with me was important representation of other demographic groups. Most notably, I did not speak to any Indigenous people from Saskatoon or the rest of Canada. Given the unique cultural and historical influences on Indigenous reproduction and the specific dynamics of Indigenous queer people in the face of colonial practices, this is an important absence (Devries and Free 2011; Dyck and Lux 2016; Stevenson 2020; Jackman and Upadhyay 2014; Greensmith and Giwa 2013). Making space in research for Indigenous perspectives and voices is an important aspect of decolonizing academics and research, though, in application, it is often less simple than that (Moreton-Robinson 2004; Paine et al. 2013). I also heard very few culturally masculine voices in this research, as out of 19 participants, only two were DMAB and a further two were identified somewhere on the masculine gender spectrum. Recruiting men in general for research, and especially for reproductive research, is often quite challenging for a number of reasons, and less representation from masculine perspectives means that we are missing their viewpoints (de Lacey 2014; Law 2019). In retrospect, my recruitment practices could have been tweaked to seek out interlocutors from these groups to ensure their voices were heard and represented within this research.

Another limitation was that of my data collection methodology. Due to the safety precautions I had to follow because of the COVID-19 pandemic, I was unable to collect a wide variety of types of data often present in the staple anthropological methodologies of participant observation and ethnography and had to rely solely on interviews. I was limited to what I could see in the restricted window offered by webcams and hear through the digital encoding of Zoom audio, and in some cases, the limitations of bandwidth and the problems with computer operation meant I could not even see those sharing their stories with me. The lack of experiential data, of direct observations of locale and context, of smell and touch and impressions and insight, restricted what I can say about how people fully engage with their reproductive decision-making processes (Boyle 1994; Angrosino 2006; Spradley 2016; Pool 2017). The results of this thesis are thus limited to what people were able to tell me when the human experience goes far beyond words and the thoughts that go into communicating them.

There is much work that must still be done. The area of queer reproductive decision-making is ripe for further qualitative, especially anthropological, research. Specifically, I believe

an ethnographic approach to queer reproductive decision-making in Saskatoon would bring light to the broader interconnections among the institutional, interpersonal, and individual influences that inform the decisions being made. Revisiting specific interlocutors over multiple points in time, speaking with them in the places where they make many of their decisions or find meaning, and observing their interactions with other people involved in the decision-making process would likely reveal a web of interconnected influences on reproductive decision-making that I have not even glimpsed. Such an effort would be unlikely to find the same meanings regarding COVID-19 as the scale, scope, and, most importantly, the meaning of the pandemic has shifted significantly in the year between my interviews and the writing of this thesis. I cannot imagine capturing the same types of reactions I have found in my work at an even later date, but such is the temporal nature of qualitative research (Bear 2016; Gell 2021). No research is perfect, and in outlining the ways my research could be improved, I hope to add substance and guidance to the continued discussion of queer reproductive decision-making.

# 7.3 – Final Thoughts

Making reproductive decisions will never be a simple task. There is always a complex network of decisions and factors within the broader question of "should I have a child?" (Miller 1994). The complexity is increased when you are considering a minority population who may be making decisions that do not align neatly with cultural narratives about "normal" reproduction, like some of the queer people I interviewed for this research (Mamo and Alston-Stepnitz 2015). These and other factors make the exploration of why those decisions are made and the factors that influence making those decisions a serious undertaking in even the most ideal of conditions a significant undertaking. This thesis presents the participants' experiences and my interpretation of them through an understanding of reproductive anthropology, queer theory, and the critical deconstruction possible within phenomenology. I began with the question: what meanings do queer people in Saskatoon find in their reproductive decision-making processes, and how do those meanings influence those processes during the COVID-19 pandemic? The generous gift of eighteen stories graciously given by queer people living in Saskatoon during the early years of the COVID-19 pandemic has provided helpful insight into the experiences and meanings of queer reproductive decision-making. Yet I feel I have only scratched the surface of the voluminous myriad of influences experienced and described by those who have shared with me.

I hope that this thesis has honoured those eighteen voices and the wider communities they represent with my answer to that research question: queer people in Saskatoon are influenced in their reproductive decision-making by meanings they have created around queer and reproductive safety, the family structures they feel are best and are supported in making, and how COVID has impacted meanings of place.

#### BIBLIOGRAPHY

- Abad Espinoza, Luis Gregorio. 2022. "The Ethnographic Quest in the Midst of COVID-19." *International Journal of Qualitative Methods* 21 (October): 16094069221135968. https://doi.org/10.1177/16094069221135967.
- Abelsohn, Kira A., Rachel Epstein, and Lori E. Ross. 2013. "Celebrating the 'Other' Parent: Mental Health and Wellness of Expecting Lesbian, Bisexual, and Queer Non-Birth Parents." *Journal of Gay & Lesbian Mental Health* 17 (4): 387–405. https://doi.org/10.1080/19359705.2013.771808.
- Abortion Rights Coalition of Canada. 2022. "Statistics Abortion in Canada." https://www.arcc-cdac.ca/wp-content/uploads/2020/07/statistics-abortion-in-canada.pdf.
- Abramowitz, Sharon. 2016. "Epidemics (Especially Ebola)." *Annual Review of Anthropology* 46 (1): 421.
- Adams, Britni L., Jennifer W. Applebaum, Michelle N. Eliasson, Shelby E. McDonald, and Barbara A. Zsembik. 2021. "Child and Pet Care-Planning During COVID-19: Considerations for the Evolving Family Unit." *Family Relations* 70 (3): 705–16. https://doi.org/10.1111/fare.12542.
- Akbari, Mirabel. 2021. "Forced Sterilization of Indigenous Women: : An Act of Genocide or Policing Women's Bodies?" *York University Criminological Review* 3 (1). https://csri.journals.yorku.ca/index.php/default/article/view/113.
- Alexander, Bryant Keith. 2017. "Queer/Quare Theory: Worldmaking and Methodologies." In *The SAGE Handbook of Qualitative Research*, edited by Norman K. Denzin and Yvonna S. Lincoln, 275–307. SAGE Publications.
- Alvarez, Bruna. 2018. "Reproductive Decision Making in Spain: Heterosexual Couples' Narratives About How They Chose to Have Children." *Journal of Family Issues* 39 (13): 3487–3507. https://doi.org/10.1177/0192513X18783494.
- Alwan, S, IM Yee, M Dybalski, C Guimond, E Dwosh, TM Greenwood, R Butler, and AD Sadovnick. 2013. "Reproductive Decision Making after the Diagnosis of Multiple Sclerosis (MS)." *Multiple Sclerosis Journal* 19 (3): 351–58. https://doi.org/10.1177/1352458512452920.
- Anderson, Austin R., and Eric Knee. 2021. "Queer Isolation or Queering Isolation? Reflecting upon the Ramifications of COVID-19 on the Future of Queer Leisure Spaces." *Leisure Sciences* 43 (1–2): 118–24. https://doi.org/10.1080/01490400.2020.1773992.
- Angrosino, Michael V. 2006. Doing Cultural Anthropology: Projects for Ethnographic Data Collection, Second Edition. Waveland Press.
- Angus Reid Institute. 2022. "As 24 Months of Pandemic Restrictions Lift, Are Canadians Ready or Reluctant to Let Go?" Angus Reid Institute. https://angusreid.org/covid-restrictions-precautions-masking-trudeau/.
- Anifandis, George, Christina I. Messini, Alexandros Daponte, and Ioannis E. Messinis. 2020. "COVID-19 and Fertility: A Virtual Reality." *Reproductive BioMedicine Online* 41 (2): 157–59. https://doi.org/10.1016/j.rbmo.2020.05.001.
- Anton, Jessie. 2022. "Sask. Government under Fire for Funding Independent Schools Run by Anti-LGBTQ Churches | CBC News." *CBC*, June 23, 2022. https://www.cbc.ca/news/canada/saskatchewan/sask-religious-independent-school-funding-anti-lgbtq-1.6498368.

- Archer, Matthew. 2021. "Homonormativity." In *Homonormativity*, edited by Kamden K. Strunk and Stephanie Anne Shelton, 284–89. Netherlands: Brill. https://doi.org/10.1163/9789004506725\_056.
- Arendt, Hannah. 2003. *Responsibility and Judgment*. Edited by Jerome Kohn. New York, NY: Schnocken.
- Årestedt, Liselott, Eva Benzein, Carina Persson, and Margareta RÄmgÅrd. 2016. "A Shared Respite—The Meaning of Place for Family Well-Being in Families Living with Chronic Illness." *International Journal of Qualitative Studies on Health and Well-Being* 11 (1): 30308. https://doi.org/10.3402/qhw.v11.30308.
- Ariizumi, Hideki, Yaqin Hu, and Tammy Schirle. 2015. "Stand Together or Alone? Family Structure and the Business Cycle in Canada." *Review of Economics of the Household* 13 (1): 135–61. https://doi.org/10.1007/s11150-013-9195-8.
- Ariyabandu, Madhavi Malalgoda. 2009. "Sex, Gender and Gender Relations in Disasters." In *Women, Gender and Disaster: Global Issues and Initiatives*, edited by Elaine Enarson and P. G. Dhar Chakrabarti, 5–17. New Delhi: SAGE Publications India.
- Armstrong, Elizabeth A. 2002. Forging Gay Identities: Organizing Sexuality in San Francisco, 1950-1994. Chicago, IL: University of Chicago Press. https://press.uchicago.edu/ucp/books/book/chicago/F/bo3621651.html.
- Atkinson, Sean R., and Darren Russell. 2015. "Gender Dysphoria." *Australian Family Physician* 44 (11): 792–96. https://doi.org/10.3316/informit.585153959808244.
- Atter, Heidi. 2021. "Homophobic Incident at Saskatoon Rainbow Crosswalk Shows Need for Pride, Says Transgender Man | CBC News." *CBC*, June 11, 2021. https://www.cbc.ca/news/canada/saskatoon/pride-crosswalk-incident-saskatoon-1.6061517.
- Aven, Terje, and Ortwin Renn. 2009. "On Risk Defined as an Event Where the Outcome Is Uncertain." *Journal of Risk Research* 12 (1): 1–11. https://doi.org/10.1080/13669870802488883.
- Avramopoulou, Eirini. 2022. "The Everyday Makings of Heteronormativity: Cross-Cultural Explorations of Sex, Gender, and Sexuality." *Journal of Middle East Women's Studies* 18 (1): 137–40. https://doi.org/10.1215/15525864-9494206.
- Aznar, Ana, Paul Sowden, Sarah Bayless, Kirsty Ross, Amy Warhurst, and Dimitra Pachi. 2021. "Home-Schooling during COVID-19 Lockdown: Effects of Coping Style, Home Space, and Everyday Creativity on Stress and Home-Schooling Outcomes." *Couple and Family Psychology: Research and Practice* 10 (4): 294–312. https://doi.org/10.1037/cfp0000182.
- Badgett, M V Lee, Laura E Durso, and Alyssa Schneebaum. 2013. "New Patterns of Poverty in the Lesbian, Gay, and Bisexual Community." The Williams Institute. https://williamsinstitute.law.ucla.edu/wp-content/uploads/Poverty-LGB-Jun-2013.pdf.
- Baggaley, Rebecca F, Richard G White, and Marie-Claude Boily. 2010. "HIV Transmission Risk through Anal Intercourse: Systematic Review, Meta-Analysis and Implications for HIV Prevention." *International Journal of Epidemiology* 39 (4): 1048–63. https://doi.org/10.1093/ije/dyq057.
- Ball, Kirstie, Nicola Green, Hille Koskela, and David J. Phillips. 2009. "Editorial: Surveillance Studies Needs Gender and Sexuality." *Surveillance & Society* 6 (4): 352–55.
- Barker, Meg-John. 2016. Queer: A Graphic History. Icon Books.
- Barnhart, Robert, and Sol Steinmetz, eds. 1999. *Chambers Dictionary of Etymology*. Illustrated edition. Edinburgh: Quercus.

- Barrow, Tyler. 2022. "Performers Excited to Be Back at Saskatchewan Jazz Festival." *CTV News*, July 2, 2022, sec. Saskatoon. https://saskatoon.ctvnews.ca/performers-excited-to-be-back-at-saskatchewan-jazz-festival-1.5971981.
- Bastos, Ana Cecília de Sousa, Vivian Volkmer-Pontes, Pedro Gomes Brasileiro, and Helena Martinelli Serra. 2013. "Fathering in Brazil: A Diverse and Unknown Reality." In *Fathers in Cultural Context*, 228–49. New York, NY, US: Routledge/Taylor & Francis Group.
- Battell Lowman, Emma, and Adam J. Barker. 2015. Settler: Identity and Colonialism in 21st Century Canada. Halifax: Fernwood Publishing.
- Battin, Margaret Pabst. 1997. "Sex and Consequences: World Population Growth vs. Reproductive Rights." *TRUE*, January. https://soar.suny.edu/handle/20.500.12648/3160.
- Bawer, Bruce. 2008. *Place at the Table: The Gay Individual in American Society*. Simon and Schuster.
- Beagan, Brenda L., Stephanie R. Bizzeth, Tara M. Pride, and Kaitlin R. Sibbald. 2022. "LGBTQ+ Identity Concealment and Disclosure within the (Heteronormative) Health Professions: 'Do I? Do I Not? And What Are the Potential Consequences?" *SSM* - *Qualitative Research in Health* 2 (December): 100114. https://doi.org/10.1016/j.ssmqr.2022.100114.
- Bear, Laura. 2016. "Time as Technique." *Annual Review of Anthropology* 45 (1): 487–502. https://doi.org/10.1146/annurev-anthro-102313-030159.
- Bell, Jonathan. 2020. "Between Private and Public: AIDS, Health Care Capitalism, and the Politics of Respectability in 1980s America." *Journal of American Studies* 54 (1): 159–83. https://doi.org/10.1017/S0021875818000518.
- Ben Daniel, Tallie, and Hilary Berwick. 2020. "Queer In/Security." *GLQ: A Journal of Lesbian and Gay Studies* 26 (1): 129–40. https://doi.org/10.1215/10642684-7929157.
- Bengtson, Vern L. 2001. "Beyond the Nuclear Family: The Increasing Importance of Multigenerational Bonds." *Journal of Marriage and Family* 63 (1): 1–16. https://doi.org/10.1111/j.1741-3737.2001.00001.x.
- Bertone, Chiara, and Maria Pallotta-Chiarolli. 2014. "Putting Families of Origin into the Queer Picture: Introducing This Special Issue." *Journal of GLBT Family Studies* 10 (1–2): 1–14. https://doi.org/10.1080/1550428X.2013.857494.
- Bhagat, Rajesh K., M. S. Davies Wykes, Stuart B. Dalziel, and P. F. Linden. 2020. "Effects of Ventilation on the Indoor Spread of COVID-19." *Journal of Fluid Mechanics* 903 (November). https://doi.org/10.1017/jfm.2020.720.
- Bingaman, Sandra. 2013. "Historical Report on Provincial Elections in Saskatchewan: Conducting Provincial Elections in Saskatchewan 1905-2013." Elections Saskatchewan. https://www.elections.sk.ca/what-we-do/historical-report/.
- Bissell, David. 2021. "A Changing Sense of Place: Geography and COVID-19." *Geographical Research* 59 (2): 150–59. https://doi.org/10.1111/1745-5871.12465.
- Bodin, Maja, and Jenny Björklund. 2022. "Can I Take Responsibility for Bringing a Person to This World Who Will Be Part of the Apocalypse!?": Ideological Dilemmas and Concerns for Future Well-Being When Bringing the Climate Crisis into Reproductive Decision-Making." *Social Science & Medicine* 302 (June): 114985. https://doi.org/10.1016/j.socscimed.2022.114985.
- Boellstorff, Tom. 2007. "Queer Studies in the House of Anthropology." *Annual Review of Anthropology* 36 (1): 17–35. https://doi.org/10.1146/annurev.anthro.36.081406.094421.
- Bogaert, Anthony F. 2015. Understanding Asexuality. Rowman & Littlefield.

- Borbasi, Sally, Ysanne Chapman, Lee-Anne Gassner, Sandra Dunn, and Kathy Read. 2002. "Perceptions of the Researcher: In-Depth Interviewing in the Home." *Contemporary Nurse* 14 (1): 24–37. https://doi.org/10.5172/conu.14.1.24.
- Borell, Klas. 2003. "Family and Household. Family Research and Multi-Household Families." *International Review of Sociology* 13 (3): 467. https://doi.org/10.1080/0390670032000139476.
- Bott Spillius, Elizabeth, ed. 2013. *Family and Social Network: Roles, Norms and External Relationships in Ordinary Urban Families*. London: Routledge. https://doi.org/10.4324/9781315824130.
- Boulet, Virginie, and Nadine Badets. 2017. "Early Motherhood among Off-Reserve First Nations, Métis and Inuit Women." December 1, 2017. http://www150.statcan.gc.ca/n1/pub/75-006-x/2017001/article/54877-eng.htm.
- Bourdieu, Pierre. 1990. The Logic of Practice. Stanford University Press.
- Bowen, Glenn A. 2006. "Grounded Theory and Sensitizing Concepts." *International Journal of Qualitative Methods* 5 (3): 12–23. https://doi.org/10.1177/160940690600500304.
- Boyd, Robert, and Peter J. Richerson. 1994. "The Evolution of Norms: An Anthropological View." *Journal of Institutional and Theoretical Economics (JITE) / Zeitschrift Für Die Gesamte Staatswissenschaft* 150 (1): 72–87.
- Boyle, Joyceen. 1994. "Styles of Ethnography." In *Critical Issues in Qualitative Research Methods*, edited by Janice M. Morse. New York: SAGE.
- Boyle, Micheal H., and Ellen L. Lipman. 2002. "Do Places Matter? Socioeconomic Disadvantage and Behavioral Problems of Children in Canada." *Journal of Consulting and Clinical Psychology* 70 (2): 378–89. https://doi.org/10.1037/0022-006X.70.2.378.
- Brennan, David J., Kiffer G. Card, David Collict, Jody Jollimore, and Nathan J. Lachowsky. 2020. "How Might Social Distancing Impact Gay, Bisexual, Queer, Trans and Two-Spirit Men in Canada?" *AIDS and Behavior* 24 (9): 2480–82. https://doi.org/10.1007/s10461-020-02891-5.
- Brown, Michael F. 2008. "Cultural Relativism 2.0" 49 (3): 363–83. https://doi.org/10.1086/529261.
- Browne, Evie. 2019. "Gender Norms, LGBTQI Issues and Development: A Topic Guide." London, UK: Advancing Learning and Innovation in Gender Norms. https://www.alignplatform.org/sites/default/files/2019-01/gender\_norms\_and\_lgbtqi\_issues.pdf.
- Browne, Kath, Leela Bakshi, and Jason Lim. 2011. "It's Something You Just Have to Ignore': Understanding and Addressing Contemporary Lesbian, Gay, Bisexual and Trans Safety Beyond Hate Crime Paradigms." *Journal of Social Policy* 40 (4): 739–56. https://doi.org/10.1017/S0047279411000250.
- Bryant, Rebecca. 2002. "The Purity of Spirit and the Power of Blood: A Comparative Perspective on Nation, Gender and Kinship in Cyprus." *Journal of the Royal Anthropological Institute* 8 (3): 509–30. https://doi.org/10.1111/1467-9655.00120.
- Bubandt, Nils. 2005. "Vernacular Security: The Politics of Feeling Safe in Global, National and Local Worlds." *Security Dialogue* 36 (3): 275–96. https://doi.org/10.1177/0967010605057015.
- Burki, Talha. 2020. "The Indirect Impact of COVID-19 on Women." *The Lancet Infectious Diseases* 20 (8): 904–5. https://doi.org/10.1016/S1473-3099(20)30568-5.

- Butler, Judith. 2006. *Gender Trouble: Feminism and the Subversion of Identity*. 1st edition. New York: Routledge.
- . 2009. "Performativity, Precarity and Sexual Politics." *Revista de Antropología Iberoamericana* 4 (3): i–xiii. https://doi.org/10.11156/aibr.040303e.
- Byne, William, Dan H. Karasic, Eli Coleman, A. Evan Eyler, Jeremy D. Kidd, Heino F.L. Meyer-Bahlburg, Richard R. Pleak, and Jack Pula. 2018. "Gender Dysphoria in Adults: An Overview and Primer for Psychiatrists." *Transgender Health* 3 (1): 57-A3. https://doi.org/10.1089/trgh.2017.0053.
- Cachón-Zagalaz, Javier, María Sánchez-Zafra, Déborah Sanabrias-Moreno, Gabriel González-Valero, Amador J. Lara-Sánchez, and María Luisa Zagalaz-Sánchez. 2020. "Systematic Review of the Literature About the Effects of the COVID-19 Pandemic on the Lives of School Children." *Frontiers in Psychology* 11. https://www.frontiersin.org/articles/10.3389/fpsyg.2020.569348.
- Caffrey, N., M. Goldman, A. Lewin, Y. Grégoire, Q-L. Yi, and S. F. O'Brien. 2022. "Removing the Men Who Have Sex with Men Blood Donation Deferral: Informing Risk Models Using Canadian Public Health Surveillance Data." *Transfusion Clinique et Biologique* 29 (3): 198–204. https://doi.org/10.1016/j.tracli.2022.06.001.
- Callaghan, Tonya D. 2016. "Young, Queer, and Catholic: Youth Resistance to Homophobia in Catholic Schools." *Journal of LGBT Youth* 13 (3): 270–87. https://doi.org/10.1080/19361653.2016.1185758.
- ——. 2019. "Homophobia in the Hallways: LGBTQ People at Risk in Catholic Schools." *National Post*, January 17, 2019. https://nationalpost.com/pmn/news-pmn/homophobia-in-the-hallways-lgbtq-people-at-risk-in-catholic-schools.
- Cameron, Angela. 2008. "Regulating the Queer Family: The Assisted Human Reproduction Act Case Comment." *Canadian Journal of Family Law* 24 (1): 101–22.
- Cameron, John. 2021. "Mapping the Connections between Anti-Queer, Anti-Trans Speakers at Regina's Conversion Therapy Ban Council Meeting." *Sask Dispatch*, May 8, 2021. https://saskdispatch.com/articles/view/mapping-the-connections-between-anti-queer-anti-trans-speakers-at-reginas-conversion-therapy-ban-council-meeting.
- Campion-Smith, Bruce. 2012. "Canadian Families Growing More Diverse, Census Data Shows." *Toronto Star*, September 20, 2012, sec. Canada. https://www.thestar.com/news/canada/2012/09/20/canadian\_families\_growing\_more\_diverse\_census\_data\_shows.html.
- Canadian Blood Services. 2023. "Sexual Behaviour-Based Screening." 2023. https://www.blood.ca/en/blood/am-i-eligible-donate-blood/sexual-behaviour-based-screening.
- Canadian Institute for Health Information. 2022. "Induced Abortions Reported in Canada in 2020." https://www.cihi.ca/sites/default/files/document/induced-abortions-reported-in-canada-2020-en.xlsx.
- Canetto, Silvia Sara. 1996. "What Is a Normal Family? Common Assumptions and Current Evidence." *Journal of Primary Prevention* 17 (1): 31–46. https://doi.org/10.1007/BF02262737.
- Caputo, Virginia. 1999. "At 'Home' and 'Away': Reconfiguring the Field for Late Twentieth-Century Anthropology." In *Constructing the Field*. Routledge.
- Carli, Linda L. 2020. "Women, Gender Equality and COVID-19." *Gender in Management: An International Journal* 35 (7/8): 647–55. https://doi.org/10.1108/GM-07-2020-0236.

- Carpenter, Emma. 2021. "The Health System Just Wasn't Built for Us': Queer Cisgender Women and Gender Expansive Individuals' Strategies for Navigating Reproductive Health Care." *Women's Health Issues* 31 (5): 478–84. https://doi.org/10.1016/j.whi.2021.06.004.
- Carsten, Janet. 2000. *Cultures of Relatedness: New Approaches to the Study of Kinship*. Cambridge University Press.
- ——. 2004. *After Kinship*. Cambridge University Press.
- Cartwright, Elizabeth. 2019. "The Medical Anthropology of Climate Change: Eco-Risks and the Body Environmental." *Medical Anthropology* 38 (5): 436–39. https://doi.org/10.1080/01459740.2019.1621866.
- Casey, Edward S. 1993. Getting Back Into Place: Toward a Renewed Understanding of the Place-World. Indiana University Press.
- CBS News. 2016. "Gay Marriage Goes Dutch CBS News." *CBS News*, January 28, 2016. https://web.archive.org/web/20160128181758/http://www.cbsnews.com/news/gay-marriage-goes-dutch/.
- Centers for Disease Control and Prevention. 2021. "HIV Surveillance Report 2019." 32. HIV Surveillance Report. Atlanta, Georgia. https://www.cdc.gov/hiv/pdf/library/reports/surveillance/cdc-hiv-surveillance-report-2018-updated-vol-32.pdf.
- Chapman, Rachel R. 2006. "Chikotsa—Secrets, Silence, and Hiding: Social Risk and Reproductive Vulnerability in Central Mozambique." *Medical Anthropology Quarterly* 20 (4): 487–515. https://doi.org/10.1525/maq.2006.20.4.487.
- Chbat, Marianne, and Isabel Côté. 2022. "Lesbian and Queer Non-Birthing Mothers in France, Switzerland, and Québec: Differentiated Access to Reproductive Rights." *Sexuality, Gender & Policy* 5 (1): 26–42. https://doi.org/10.1002/sgp2.12043.
- Cheng, Zhiming, Silvia Mendolia, Alfredo R. Paloyo, David A. Savage, and Massimiliano Tani. 2021. "Working Parents, Financial Insecurity, and Childcare: Mental Health in the Time of COVID-19 in the UK." *Review of Economics of the Household* 19 (1): 123–44. https://doi.org/10.1007/s11150-020-09538-3.
- Choi, Soon Kyu, Shahrzad Divsalar, Jennifer Flórez-Donado, Krystal Kittle, Andy Lin, Ilan H Meyer, and Prince Torres-Salazar. 2019. "Stress, Health, and Well-Being of LGBT People in Colombia: Results from a National Survey." UCLA. https://www.ohchr.org/sites/default/files/Documents/Issues/SexualOrientation/IESOGI/Ac ademics/1912\_Colombia\_Report\_English\_FINAL.pdf.
- Chung, Andrew. 2022. "Roe v. Wade Decision Could Put Same-Sex Marriage, Other Rights at Risk, Experts Say." *Global News*, May 4, 2022. https://globalnews.ca/news/8808354/roe-v-wade-potential-impact-expert-analysis/.
- Clare, Stephanie. 2019. "Reimagining Biological Relatedness: Epigenetics and Queer Kin." *Signs: Journal of Women in Culture and Society* 45 (1): 51–73. https://doi.org/10.1086/703495.
- Clarke, Erin. 2021. "Indigenous Women and the Risk of Reproductive Healthcare: Forced Sterilization, Genocide, and Contemporary Population Control." *Journal of Human Rights and Social Work* 6 (2): 144–47. https://doi.org/10.1007/s41134-020-00139-9.
- Clarke, Victoria, Nikki Hayfield, Sonja J. Ellis, and Gareth Terry. 2018. "Lived Experiences of Childfree Lesbians in the United Kingdom: A Qualitative Exploration." *Journal of Family Issues* 39 (18): 4133–55. https://doi.org/10.1177/0192513X18810931.

- Collins, Joshua C. 2013. "Stress and Safety for Gay Men at Work Within Masculinized Industries." *Journal of Gay & Lesbian Social Services* 25 (3): 245–68. https://doi.org/10.1080/10538720.2013.806878.
- Connolly, Creighton, S Harris Ali, and Roger Keil. 2020. "On the Relationships between COVID-19 and Extended Urbanization." *Dialogues in Human Geography* 10 (2): 213–16. https://doi.org/10.1177/2043820620934209.
- Cooper, David Heath, and Joane Nagel. 2021. "Lessons from the Pandemic: Climate Change and COVID-19." *International Journal of Sociology and Social Policy* 42 (3/4): 332–47. https://doi.org/10.1108/IJSSP-07-2020-0360.
- Cowden, Richard G., Victor Counted, and Haywantee Ramkissoon. 2021. "Place Attachment and Suffering During a Pandemic." In *Place and Post-Pandemic Flourishing: Disruption, Adjustment, and Healthy Behaviors*, edited by Victor Counted, Richard G. Cowden, and Haywantee Ramkissoon, 45–54. SpringerBriefs in Psychology. Cham: Springer International Publishing. https://doi.org/10.1007/978-3-030-82580-5 4.
- Cox, Cece. 2005. "To Have and to Hold Or Not: The Influence of the Christian Right on Gay Marriage Laws in the Netherlands, Canada, and the United States." *Law & Sexuality: A Review of Lesbian, Gay, Bisexual, and Transgender Legal Issues* 14: 1–50.
- Crary, David, and Luis Andres Henao. 2021. "LGBTQ Catholics Stung by Vatican Rebuff of Same-Sex Unions." *CityNews Vancouver*, March 15, 2021, sec. World. https://vancouver.citynews.ca/2021/03/15/lgbtq-catholics-stung-by-vatican-rebuff-of-same-sex-unions/.
- Craven, Christa. 2019. *Reproductive Losses: Challenges to LGBTQ Family-Making*. Routledge. Craven, Christa, and Elizabeth Peel. 2014. "Stories of Grief and Hope: Queer Experiences of Reproductive Loss." In , edited by Margaret F. Gibson, 97–110. Ontario, Canada: Demeter Press. https://eprints.worc.ac.uk/3459/.
- Cresswell, Tim. 2009. "Place." In *International Encyclopedia of Human Geography*, edited by Nigel Thrift and Rob Kitchen, 169–77. Oxford, England: Elsevier. https://pure.royalholloway.ac.uk/en/publications/place.
- ——. 2014. *Place: An Introduction*. John Wiley & Sons.
- Croll, Julie, Laura Sanapo, and Ghada Bourjeily. 2022. "LGBTQ+ Individuals and Pregnancy Outcomes: A Commentary." *BJOG: An International Journal of Obstetrics & Gynaecology* 0 (0): 1–5. https://doi.org/10.1111/1471-0528.17131.
- Cucciari, Salvatore. 1981. "The Gender Revolution and the Transition from Bisexual Horde to Patrilocal Band: The Origins of Gender Hierarchy." In *Sexual Meanings: The Cultural Construction of Gender and Sexuality*, edited by Sherry B. Ortner and Harriet Whitehead, 31–79. Cambridge: Cambridge University Press.
- Cumming-Potvin, Wendy, and Wayne Martino. 2014. "Teaching about Queer Families: Surveillance, Censorship, and the Schooling of Sexualities." *Teaching Education* 25 (3): 309–33. https://doi.org/10.1080/10476210.2014.889672.
- Da Costa, Jade Crimson Rose. 2020. "Pride Parades in Queer Times: Disrupting Time, Norms, and Nationhood in Canada." *Journal of Canadian Studies* 54 (2–3): 434–58. https://doi.org/10.3138/jcs-2020-0045.
- Dahl, Ulrika. 2014. "Not Gay as in Happy, but Queer as in Fuck You: Notes on Love and Failure in Queer(Ing) Kinship." *Lambda Nordica* 3 (4): 143–68.

- Dahlseide, Dallas. 2022. "A Return to Normal': Live Festival Shows Are Back in Downtown Saskatoon." 650 CKOM, June 14, 2022. https://www.ckom.com/2022/06/14/a-return-to-normal-live-festival-shows-are-back-in-downtown-saskatoon/.
- Dakis, Alvin Cloyd and International Gay and Lesbian Human Rights Commission. 2012. Human Rights Violations on the Basis of Sexual Orientation, Gender Identity, and Homosexuality in the Philippines Coalition Report. https://doi.org/10.13140/RG.2.2.31215.74409.
- Dasgupta, Rohit K. 2021. "Queer Patchworks: Liveability, Creative Work and Survival in the Time of COVID." In *COVID-19 Assemblages*. New Delhi: Routledge India. https://www-taylorfrancis-com.cyber.usask.ca/chapters/edit/10.4324/9781003262251-11/queer-patchworks-rohit-dasgupta.
- Dator, Jim. 2022. "Weirding the Queer." In *Beyond Identities: Human Becomings in Weirding Worlds*, edited by Jim Dator, 197–99. Anticipation Science. Cham: Springer International Publishing. https://doi.org/10.1007/978-3-031-11732-9\_14.
- Dayal, Pratyush. 2021. "More Students Enrolled in Saskatoon, Regina School Divisions than Last Year | CBC News." *CBC News*, October 21, 2021. https://www.cbc.ca/news/morestudents-enrolled-in-saskatoon-regina-school-divisions-than-last-year-1.6218838.
- ——. 2022. "12.5 per Cent of Sask. Population Is Now Immigrants: 2021 Census." *CBC News*, October 27, 2022. https://www.cbc.ca/news/canada/saskatoon/immigration-is-driving-population-growth-in-saskatchewan-1.6630693.
- Dean, Wesley R. 2002. "Broken Promises: The Canadian Tainted Blood Scandal." Doctor of Philosophy, Edmonton, Alberta: University of Alberta. https://era.library.ualberta.ca/items/33e530e0-5f14-466f-ad20-7121a26dfbcf/download/2be05f78-82f0-4b72-8466-a54262746429.
- Decker, Julie Sondra. 2015. *The Invisible Orientation: An Introduction to Asexuality*. New York: Simon and Schuster.
- Demšar, Karin, Matija Svetina, Ivan Verdenik, Natasa Tul, Isaac Blickstein, and Vislava Globevnik Velikonja. 2018. "Tokophobia (Fear of Childbirth): Prevalence and Risk Factors." *Journal of Perinatal Medicine* 46 (2): 151–54. https://doi.org/10.1515/jpm-2016-0282.
- Devine-Wright, Patrick, Laís Pinto de Carvalho, Andrés Di Masso, Maria Lewicka, Lynne Manzo, and Daniel R. Williams. 2020. "'Re-Placed' Reconsidering Relationships with Place and Lessons from a Pandemic." *Journal of Environmental Psychology* 72 (December): 101514. https://doi.org/10.1016/j.jenvp.2020.101514.
- Devries, Karen M., and Caroline J. Free. 2011. "It's Not Something You Have to Be Scared About': Attitudes towards Pregnancy and Fertility among Canadian Aboriginal Young People." *International Journal of Indigenous Health* 7 (1): 8–15. https://doi.org/10.18357/ijih71201112349.
- Dewaele, Alexis, Nele Cox, Wim Van den Berghe, and John Vincke. 2011. "Families of Choice? Exploring the Supportive Networks of Lesbians, Gay Men, and Bisexuals1." *Journal of Applied Social Psychology* 41 (2): 312–31. https://doi.org/10.1111/j.1559-1816.2010.00715.x.
- Dhejne, Cecilia, Roy Van Vlerken, Gunter Heylens, and Jon Arcelus. 2016. "Mental Health and Gender Dysphoria: A Review of the Literature." *International Review of Psychiatry* 28 (1): 44–57. https://doi.org/10.3109/09540261.2015.1115753.

- Di Ciaula, Agostino, Marcin Krawczyk, Krzysztof J. Filipiak, Andreas Geier, Leonilde Bonfrate, and Piero Portincasa. 2021. "Noncommunicable Diseases, Climate Change and Iniquities: What COVID-19 Has Taught Us about Syndemic." *European Journal of Clinical Investigation* 51 (12): e13682. https://doi.org/10.1111/eci.13682.
- Di Marco, Moreno, Michelle L. Baker, Peter Daszak, Paul De Barro, Evan A. Eskew, Cecile M. Godde, Tom D. Harwood, et al. 2020. "Sustainable Development Must Account for Pandemic Risk." *Proceedings of the National Academy of Sciences* 117 (8): 3888–92. https://doi.org/10.1073/pnas.2001655117.
- Dilley, R. M. 2002. "The Problem of Context in Social and Cultural Anthropology." *Language & Communication* 22 (4): 437–56. https://doi.org/10.1016/S0271-5309(02)00019-8.
- Dixon, Jenny. 2018. "Looking Out From the Family Closet: Discourse Dependence and Queer Family Identity in Workplace Conversation." *Management Communication Quarterly* 32 (2): 271–75. https://doi.org/10.1177/0893318917744067.
- Djuric, Mickey. 2021. "Saskatoon Couple's Fight with the Law Brings Change to Saskatchewan Birth Certificates." *CBC News*, August 31, 2021. https://www.cbc.ca/news/canada/saskatchewan/couple-fight-brings-change-to-birth-certificates-1.6159745.
- ——. 2022. "Winter of Misery Is Here': Saskatchewan Epidemiologist Warns of Spike in COVID Cases." *National Post*, January 4, 2022. https://nationalpost.com/pmn/news-pmn/canada-news-pmn/saskatchewan-reports-1954-new-cases-over-four-day-period-opens-up-fourth-doses.
- Docherty, The Hon. Mark. 2020. "Fourth Session Twenty-Eighth Legislature of the Legislative Assembly of Saskatchewan Debates and Proceedings." The Legislative Assembly of Saskatchewan. https://docs.legassembly.sk.ca/legdocs/Legislative%20Assembly/Hansard/28L4S/200302D ebates.pdf.
- Dorow, Sara K. 2006. Transnational Adoption: A Cultural Economy of Race, Gender, and Kinship. NYU Press.
- Doubilet, Peter M., Carol B. Benson, Tom Bourne, and Michael Blaivas. 2013. "Diagnostic Criteria for Nonviable Pregnancy Early in the First Trimester." Edited by Edward W. Campion. *New England Journal of Medicine* 369 (15): 1443–51. https://doi.org/10.1056/NEJMra1302417.
- Downe, Pamela J. 1999. "Laughing When It Hurts: Humor and Violence in the Lives of Costa Rican Prostitutes." *Women's Studies International Forum* 22 (1): 63–78. https://doi.org/10.1016/S0277-5395(98)00109-5.
- Downe, Pamela J. 2020. *Collective Care: Indigenous Motherhood, Family, and HIV/AIDS*. Toronto: University of Toronto Press. https://utorontopress.com/ca/collective-care-2.
- Drapeau, Sylvie, Marie Simard, Madeleine Beaudry, and Cecile Charbonneau. 2000. "Siblings in Family Transitions\*." *Family Relations* 49 (1): 77–85. https://doi.org/10.1111/j.1741-3729.2000.00077.x.
- Drushel, Bruce E. 2019. "The Evolution Will Not Be Broadcast (or Published): Social Capital, Assimilation, and the Changing Queer Community." *Journal of Homosexuality* 66 (12): 1756–68. https://doi.org/10.1080/00918369.2018.1511132.
- Dryden, OmiSoore H., and Suzanne Lenon. 2015. Disrupting Queer Inclusion: Canadian Homonationalisms and the Politics of Belonging. UBC Press.

- Dubrofsky, Rachel E., and Shoshana Amielle Magnet, eds. 2015. *Feminist Surveillance Studies*. *Feminist Surveillance Studies*. Duke University Press. https://doi.org/10.1515/9780822375463.
- Dyck, Erika, and Maureen Lux. 2016. "Population Control in the 'Global North'?: Canada's Response to Indigenous Reproductive Rights and Neo-Eugenics." *Canadian Historical Review* 97 (4): 481–512. https://doi.org/10.3138/chr.Dyck.
- Ebaugh, Helen Rose, and Mary Curry. 2000. "Fictive Kin as Social Capital in New Immigrant Communities." *Sociological Perspectives* 43 (2): 189–209. https://doi.org/10.2307/1389793.
- Edkins, T, T Peter, J Veale, R Malone, and E Saewyc. 2015. "BEING SAFE, BEING ME IN THE PRAIRIE PROVINCES: Results of the Canadian Trans Youth Health Survey Xin Saskatchewan & Manitoba." Vancouver: University of British Columbia. https://apsc-saravyc.sites.olt.ubc.ca/files/2018/04/SARAVYC TYHS Prairies-Final-WEB.pdf.
- Emery, Tom, and Judith C. Koops. 2022. "The Impact of COVID-19 on Fertility Behaviour and Intentions in a Middle Income Country." *PLOS ONE* 17 (1): e0261509. https://doi.org/10.1371/journal.pone.0261509.
- Enson, Stephanie. 2015. "Causes and Consequences of Heteronormativity in Healthcare and Education." *British Journal of School Nursing* 10 (2): 73–78. https://doi.org/10.12968/bjsn.2015.10.2.73.
- Epstein, Rachel. 2005. "Queer Parenting in the New Millenium: Resisting Normal." *Canadian Woman Studies* 24: 6–14.
- ——. 2018. "Space Invaders: Queer and Trans Bodies in Fertility Clinics." *Sexualities* 21 (7): 1039–58. https://doi.org/10.1177/1363460717720365.
- Epstein, Ronald M., Cleveland G. Shields, Peter Franks, Sean C. Meldrum, Mitchell Feldman, and Richard L. Kravitz. 2007. "Exploring and Validating Patient Concerns: Relation to Prescribing for Depression." *Annals of Family Medicine* 5 (1): 21–28. https://doi.org/10.1370/afm.621.
- Erikson, Susan L. 2012. "Social Embodiments: Prenatal Risk in Postsocialist Germany." *Anthropologica* 54 (1): 83–94.
- Evans, Bronwynne C., David W. Coon, Michael J. Belyea, and Ebere Ume. 2017. "Collective Care: Multiple Caregivers and Multiple Care Recipients in Mexican American Families." *Journal of Transcultural Nursing* 28 (4): 398–407. https://doi.org/10.1177/1043659616657878.
- Ewing, Lori. 2020. "Their Babies Are Almost Here but This 'magical' Time Is Tainted by COVID-19 | CBC News." *CBC*, April 1, 2020. https://www.cbc.ca/news/canada/calgary/babies-covid-calgary-toronto-birth-risk-1.5517560.
- Faas, A.J., and Roberto E. Barrios. 2015. "Applied Anthropology of Risk, Hazards, and Disasters." *Human Organization* 74 (4): 287–95. https://doi.org/10.17730/0018-7259-74.4.287.
- Faier, Lieba, and Lisa Rofel. 2014. "Ethnographies of Encounter." *Annual Review of Anthropology* 43 (1): 363–77. https://doi.org/10.1146/annurev-anthro-102313-030210.
- Farmer, Paul. 2004. "An Anthropology of Structural Violence." *Current Anthropology* 45 (3): 305–23.
- Fatmi, Mahmudur Rahman. 2020. "COVID-19 Impact on Urban Mobility." *Journal of Urban Management* 9 (3): 270–75. https://doi.org/10.1016/j.jum.2020.08.002.

- Fatmi, Mahmudur Rahman, Corrie Thirkell, and Md Shahadat Hossain. 2021. "COVID-19 and Travel: How Our Out-of-Home Travel Activity, In-Home Activity, and Long-Distance Travel Have Changed." *Transportation Research Interdisciplinary Perspectives* 10 (June): 100350. https://doi.org/10.1016/j.trip.2021.100350.
- Faubion, James. 2001. *The Ethics of Kinship: Ethnographic Inquiries*. Rowman & Littlefield Publishers.
- Feeley-Harnik, Gillian. 1999. "Communities of Blood': The Natural History of Kinship in Nineteenth-Century America." *Comparative Studies in Society and History* 41 (02). https://doi.org/10.1017/S0010417599002078.
- Feinberg, Mark E., Jacqueline A. Mogle, Jin-Kyung Lee, Samantha L. Tornello, Michelle L. Hostetler, Joseph A. Cifelli, Sunhye Bai, and Emily Hotez. 2022. "Impact of the COVID-19 Pandemic on Parent, Child, and Family Functioning." *Family Process* 61 (1): 361–74. https://doi.org/10.1111/famp.12649.
- Ferguson, Asher, and Lyric Ferguson. 2022. "LGBTQ+ Travel Safety 203 Best & Worst Countries Ranked (2022)." Asher & Lyric. June 13, 2022. https://www.asherfergusson.com/lgbtq-travel-safety/.
- Ferreira-Filho, Edson Santos, Nilson Roberto de Melo, Isabel Cristina Esposito Sorpreso, Luis Bahamondes, Ricardo Dos Santos Simões, José Maria Soares-Júnior, and Edmund Chada Baracat. 2020. "Contraception and Reproductive Planning during the COVID-19 Pandemic." *Expert Review of Clinical Pharmacology* 13 (6): 615–22. https://doi.org/10.1080/17512433.2020.1782738.
- Filice, Michelle. 2020. "Two-Spirit." In *The Canadian Encyclopedia*. Toronto: Historica Canada. https://www.thecanadianencyclopedia.ca/en/article/two-spirit.
- Findley, Lisa R. 1993. "Paths: The Nature of Linear Public Space." Places 8 (4): 72–75.
- Fine, Gary Alan, and Corey M. Abramson. 2020. "Ethnography in the Time of Covid-19. Vectors and the Vulnerable." *Etnografia e Ricerca Qualitativa*, no. 2/2020. https://doi.org/10.3240/97802.
- Fineman, Martha Albertson. 1993. "Our Sacred Institution: The Ideal of the Family in American Law and Society." *Utah Law Review* 1993 (2): 387–406.
- Finlay, Linda. 2014. "Engaging Phenomenological Analysis." *Qualitative Research in Psychology* 11 (2): 121–41. https://doi.org/10.1080/14780887.2013.807899.
- Fischer, Olivia J. 2021. "Non-Binary Reproduction: Stories of Conception, Pregnancy, and Birth." *International Journal of Transgender Health* 22 (1–2): 77–88. https://doi.org/10.1080/26895269.2020.1838392.
- Fixmer-Oraiz, Natalie, and Shui-yin Sharon Yam. 2021. "Queer(Ing) Reproductive Justice." Oxford Research Encyclopedia of Communication. November 29, 2021. https://doi.org/10.1093/acrefore/9780190228613.013.1195.
- Fletcher, Christopher M. 2006. "Environmental Sensitivity: Equivocal Illness in the Context of Place." *Transcultural Psychiatry* 43 (1): 86–105. https://doi.org/10.1177/1363461506061759.
- Flowers, Paul, Jonathan A. Smith, Paschal Sheeran, and Nigel Beail. 1997. "Health and Romance: Understanding Unprotected Sex in Relationships between Gay Men." *British Journal of Health Psychology* 2 (1): 73–86. https://doi.org/10.1111/j.2044-8287.1997.tb00524.x.

- Folgerø, Tor. 2008. "Queer Nuclear Families? Reproducing and Transgressing Heteronormativity." *Journal of Homosexuality* 54 (1–2): 124–49. https://doi.org/10.1080/00918360801952028.
- Fontaine, Kim. 2020. "Going Remote: How USask Rallied Together in the Face of the Pandemic." *University of Saskatchewan News*, July 10, 2020. https://news.usask.ca/articles/general/2020/going-remote-how-usask-rallied-together-in-the-face-of-the-pandemic.php.
- Fore, Henrietta. 2020. "Don't Let Children Be the Hidden Victims of COVID-19 Pandemic." April 9, 2020. https://www.unicef.org/press-releases/dont-let-children-be-hidden-victims-covid-19-pandemic.
- Fortunato, Laura. 2017. "Insights From Evolutionary Anthropology on the (Pre)History of the Nuclear Family." *Cross-Cultural Research* 51 (2): 92–116. https://doi.org/10.1177/1069397117691006.
- Fostik, Ana. 2021. "COVID-19 and Fertility in Canada: A Commentary." *Canadian Studies in Population* 48 (2): 217–24. https://doi.org/10.1007/s42650-021-00054-y.
- Fox, Bonnie, and Elena Neiterman. 2015. "Embodied Motherhood: Women's Feelings about Their Postpartum Bodies." *Gender & Society* 29 (5): 670–93. https://doi.org/10.1177/0891243215591598.
- Francisco, Pedro Augusto P., and Robert Muggah. 2020. "Brazil's LGBTQ Community Faces Surging Violence, but They're Fighting Back." OpenDemocracy. December 9, 2020. https://www.opendemocracy.net/en/democraciaabierta/violencia-anti-lgbtq-brasil-en/?utm\_campaign=clipping\_institucional\_dia\_a\_dia&utm\_medium=email&utm\_source=t w.
- Fredriksen-Goldsen, Karen I., Hyun-Jun Kim, Chengshi Shui, and Amanda E. B. Bryan. 2017. "Chronic Health Conditions and Key Health Indicators Among Lesbian, Gay, and Bisexual Older US Adults, 2013–2014." *American Journal of Public Health* 107 (8): 1332–38. http://dx.doi.org/10.2105/AJPH.2017.303922.
- Freed, Benjamin, Sarah Hillman, Saran Shantikumar, Debra Bick, Jeremy Dale, and Julia Gauly. 2021. "The Impact of Disasters on Contraception in OECD Member Countries: A Scoping Review." *The European Journal of Contraception & Reproductive Health Care* 26 (5): 429–38. https://doi.org/10.1080/13625187.2021.1934440.
- Freeman, Elizabeth. 2007. "Queer Belongings: Kinship Theory and Queer Theory." In *A Companion to Lesbian, Gay, Bisexual, Transgender, and Queer Studies*, edited by George E. Haggerty and Molly McGarry, 293–314. John Wiley & Sons, Ltd. https://doi.org/10.1002/9780470690864.ch15.
- Frew, Nicholas. 2020. "Report Card Grades Sask. Political Parties on Platforms for LGBTQ Issues." *CBC*, October 20, 2020. https://www.cbc.ca/news/canada/saskatchewan/saskatchewan-parties-lgbtq2-2020-election-1.5769883.
- Fukumura, Yoko E., Joseph M. Schott, Gale M. Lucas, Burcin Becerik-Gerber, and Shawn C. Roll. 2021. "Negotiating Time and Space When Working From Home: Experiences During COVID-19." *OTJR: Occupation, Participation and Health* 41 (4): 223–31. https://doi.org/10.1177/15394492211033830.
- Gachago, Daniela, Lindsay Clowes, and Janet Condy. 2018. "Family Comes in All Forms, Blood or Not': Disrupting Dominant Narratives around the Patriarchal Nuclear Family." *Gender and Education* 30 (8): 966–81. https://doi.org/10.1080/09540253.2016.1259464.

- Gadermann, Anne C., Kimberly C. Thomson, Chris G. Richardson, Monique Gagné, Corey McAuliffe, Saima Hirani, and Emily Jenkins. 2021. "Examining the Impacts of the COVID-19 Pandemic on Family Mental Health in Canada: Findings from a National Cross-Sectional Study." *BMJ Open* 11 (1): e042871. https://doi.org/10.1136/bmjopen-2020-042871.
- Garcia-Subirats, Irene, Ingrid Vargas, Amparo Susana Mogollón-Pérez, Pierre De Paepe, Maria Rejane Ferreira da Silva, Jean Pierre Unger, and María Luisa Vázquez. 2014. "Barriers in Access to Healthcare in Countries with Different Health Systems. A Cross-Sectional Study in Municipalities of Central Colombia and North-Eastern Brazil." *Social Science & Medicine* 106 (April): 204–13. https://doi.org/10.1016/j.socscimed.2014.01.054.
- Gardikiotis, Antonis. 2011. "Minority Influence." *Social and Personality Psychology Compass* 5 (9): 679–93. https://doi.org/10.1111/j.1751-9004.2011.00377.x.
- Garwood, Eliza. 2016. "Reproducing the Homonormative Family: Neoliberalism, Queer Theory and Same-Sex Reproductive Law." *Journal of International Women's Studies* 17 (2): 14.
- ——. 2022. "Queering the Kinship Story: Constructing Connection through LGBTQ Family Narratives." *Feminist Theory*, February, 14647001211059520. https://doi.org/10.1177/14647001211059521.
- Gell, Alfred. 2021. *The Anthropology of Time: Cultural Constructions of Temporal Maps and Images*. Routledge.
- Gentry, Caron. 2020. Disordered Violence: How Gender, Race and Heteronormativity Structure Terrorism. Edinburgh University Press.
- Gerstein, Josh, and Alexander Ward. 2022. "Exclusive: Supreme Court Has Voted to Overturn Abortion Rights, Draft Opinion Shows." *POLITICO*, May 2, 2022. https://www.politico.com/news/2022/05/02/supreme-court-abortion-draft-opinion-00029473.
- Getsy, David J. 2017. "Queer Relations." *ASAP/Journal* 2 (2): 254–57. https://doi.org/10.1353/asa.2017.0041.
- Ghania, Yasmine. 2022. "'Abortion Is Healthcare' Billboard Goes up in Sask. to Counter Anti-Abortion Messages." *CBC News*, November 1, 2022. https://www.cbc.ca/news/canada/saskatchewan/abortion-is-health-care-billboard-anti-abortion-messages-1.6635516.
- Gibbard, Robyn. 2018. "Meeting the Care Needs of Canada's Aging Population July 2018." Ottawa: The Conference Board of Canada. https://www.cma.ca/sites/default/files/2018-11/Conference%20Board%20of%20Canada%20-%20Meeting%20the%20Care%20Needs%20of%20Canada%27s%20Aging%20Population%20%281%29.PDF.
- Ginsburg, Faye, and Rayna Rapp. 1991. "The Politics of Reproduction." *Annual Review of Anthropology* 20 (1): 311–43. https://doi.org/10.1146/annurev.an.20.100191.001523.
- Giritli Nygren, Katarina, Susanna Öhman, and Anna Olofsson. 2017. "Doing and Undoing Risk: The Mutual Constitution of Risk and Heteronormativity in Contemporary Society." *Journal of Risk Research* 20 (3): 418–32. https://doi.org/10.1080/13669877.2015.1088056.
- Giritli Nygren, Katarina, and Anna Olofsson. 2020. "Managing the Covid-19 Pandemic through Individual Responsibility: The Consequences of a World Risk Society and Enhanced Ethopolitics." *Journal of Risk Research* 23 (7/8): 1031–35. https://doi.org/10.1080/13669877.2020.1756382.

- GLAAD. 2022a. "GLAAD Media Reference Guide Transgender Terms." GLAAD. February 22, 2022. https://www.glaad.org/reference/trans-terms.
- ——. 2022b. "GLAAD Media Reference Guide LGBTQ Terms." GLAAD. February 24, 2022. https://www.glaad.org/reference/terms.
- Glazer, Deborah. 2014. "LGBT Transitions to Parenthood." *Journal of Gay & Lesbian Mental Health* 18 (2): 213–21. https://doi.org/10.1080/19359705.2014.883668.
- Goldberg, Abbie E. 2010. *Lesbian and Gay Parents and Their Children: Research on the Family Life Cycle*. Division 44: Contemporary Perspectives on Lesbian, Gay, and Bisexual Psychology. US, Washington, DC: American Psychological Association. https://doi.org/10.1037/12055-000.
- Goldberg, Sanford C. 2020. Conversational Pressure: Normativity in Speech Exchanges. Oxford University Press.
- Gosine, Andil. 2010. "Non-White Reproduction and Same-Sex Eroticism: Queer Acts against Nature." In *Queer Ecologies: Sex, Nature, Politics, Desire*, edited by Catriona Mortimer-Sandilands and Bruce Erickson, 149–72. Indiana University Press.
- Government of Canada. 2016. "Action for Seniors Report." Research. https://www.canada.ca/en/employment-social-development/programs/seniors-action-report.html.
- ——. 2017. "The Human Rights of Lesbian, Gay, Bisexual, Transgender, Queer, 2-Spirit and Intersex Persons." GAC. February 21, 2017. https://www.international.gc.ca/world-monde/issues\_development-enjeux\_development/human\_rights-droits\_homme/rights\_lgbti-droits\_lgbti.aspx?lang=eng.
- ——. 2020. "Consolidated Federal Laws of Canada, THE CONSTITUTION ACTS, 1867 to 1982." August 7, 2020. https://laws-lois.justice.gc.ca/eng/const/page-3.html.
- Government of Canada, Statistics Canada. 2022. "Canada Tops G7 Growth despite COVID." February 9, 2022. http://www150.statcan.gc.ca/n1/daily-quotidien/220209/dq220209aeng.htm.
- Government of Saskatchewan. 2020. "COVID-19: Saskatchewan Declares State of Emergency, Imposes Additional Measures to Protect Saskatchewan Residents | News and Media." https://www.saskatchewan.ca/government/news-and-media/2020/march/18/covid-19-state-of-emergency.
- Grace, Daniel, Mark Gaspar, Benjamin Klassen, David Lessard, Praney Anand, David J. Brennan, Nathan Lachowsky, et al. 2021. "Stepping Stones or Second Class Donors?: A Qualitative Analysis of Gay, Bisexual, and Queer Men's Perspectives on Plasma Donation Policy in Canada." *BMC Public Health* 21 (1): 444. https://doi.org/10.1186/s12889-021-10480-x.
- Grace, Daniel, Mark Gaspar, Benjamin Klassen, David Lessard, David J. Brennan, Nathan J. Lachowsky, Barry D. Adam, et al. 2020. "It's in Me to Give: Canadian Gay, Bisexual, and Queer Men's Willingness to Donate Blood If Eligible Despite Feelings of Policy Discrimination." *Qualitative Health Research* 30 (14): 2234–47. https://doi.org/10.1177/1049732320952314.
- Grace, Daniel, Mark Gaspar, David Lessard, Benjamin Klassen, David J. Brennan, Barry D. Adam, Jody Jollimore, Nathan J. Lachowsky, and Trevor A. Hart. 2019. "Gay and Bisexual Men's Views on Reforming Blood Donation Policy in Canada: A Qualitative Study." *BMC Public Health* 19 (1): 772. https://doi.org/10.1186/s12889-019-7123-4.

- Graham, Mark. 2016. *Anthropological Explorations in Queer Theory*. New York, NY: Routledge. https://library.oapen.org/bitstream/handle/20.500.12657/52750/9781317180494.pdf?seque nce=1&isAllowed=y.
- Grant, Jon E., Brian L. Odlaug, Katherine Derbyshire, Liana R. N. Schreiber, Katherine Lust, and Gary Christenson. 2014. "Mental Health and Clinical Correlates in Lesbian, Gay, Bisexual, and Queer Young Adults." *Journal of American College Health* 62 (1): 75–78. https://doi.org/10.1080/07448481.2013.844697.
- Gray, Deven, and Joanna Mishtal. 2019. "Managing an Epidemic: Zika Interventions and Community Responses in Belize." *Global Public Health* 14 (1): 9–22. https://doi.org/10.1080/17441692.2018.1471146.
- Green, Adam Isaiah. 2007. "Queer Theory and Sociology: Locating the Subject and the Self in Sexuality Studies." *Sociological Theory* 25 (1): 26–45. https://doi.org/10.1111/j.1467-9558.2007.00296.x.
- Green, Ashley. 2019. "By Definition They're Not the Same Thing': Analyzing Methods of Meaning Making for Pansexual Individuals." In *Expanding the Rainbow: Exploring the Relationships of Bi+*, *Polyamorous, Kinky, Ace, Intersex, and Trans People*, edited by Brandy L. Simula, J.E. Sumerau, and Andrea Miller, 23–33. Brill. https://brill.com/view/book/edcoll/9789004414105/BP000012.xml.
- Greenglass, Esther, Daniel Chiacchia, and Lisa Fiskenbaum. 2022. "Investigating COVID-19 Stress and Coping: Substance Use and Behavioural Disengagement." *International Journal of Psychology* 57 (3): 325–35. https://doi.org/10.1002/jjop.12820.
- Greensmith, Cameron, and Sulaimon Giwa. 2013. "Challenging Settler Colonialism in Contemporary Queer Politics: Settler Homonationalism, Pride Toronto, and Two-Spirit Subjectivities." Edited by Patrick Wolfe. *American Indian Culture and Research Journal* 37 (2): 129–48. https://doi.org/10.17953/aicr.37.2.p4q2r84l12735117.
- Grindstaff, Davin Allen. 2014. Rhetorical Secrets: Mapping Gay Identity and Queer Resistance in Contemporary America. University of Alabama Press.
- Grundmann, Reiner. 2021. "COVID and Climate: Similarities and Differences." *WIREs Climate Change* 12 (6): e737. https://doi.org/10.1002/wcc.737.
- Gunawan, Joko, Siripattra Juthamanee, and Yupin Aungsuroch. 2020. "Current Mental Health Issues in the Era of Covid-19." *Asian Journal of Psychiatry* 51 (June): 102103. https://doi.org/10.1016/j.ajp.2020.102103.
- Günel, Gökçe, Saiba Varma, and Chika Watanabe. 2020. "A Manifesto for Patchwork Ethnography." Society for Cultural Anthropology. June 9, 2020. https://culanth.org/fieldsights/a-manifesto-for-patchwork-ethnography.
- Hackford-Peer, Kim. 2010. "In the Name of Safety: Discursive Positionings of Queer Youth." *Studies in Philosophy and Education* 29 (6): 541–56. https://doi.org/10.1007/s11217-010-9197-4.
- Hall, Jeffrey, Natalie Pennington, and Amanda Holmstrom. 2021. "Connecting Through Technology During COVID-19." *Human Communication & Technology* 2 (1). https://doi.org/10.17161/hct.v3i1.15026.
- Hallström, Sofia, Hanna Grundström, Anna Malmquist, Matilda Eklind, and Katri Nieminen. 2022. "Fear of Childbirth and Mental Health among Lesbian, Bisexual, Transgender and Queer People: A Cross-Sectional Study." *Journal of Psychosomatic Obstetrics & Gynecology* 0 (0): 1–6. https://doi.org/10.1080/0167482X.2022.2089555.

- Halperin, David M. 2009. "Thirteen Ways of Looking at a Bisexual." *Journal of Bisexuality* 9 (3–4): 451–55. https://doi.org/10.1080/15299710903316679.
- Harder, Lois. 2011. "After the Nuclear Age? Some Contemporary Developments in Families and Family Law in Canada." Ottawa: The Vanier Institute of the Family. https://vanierinstitute.ca/wp-content/uploads/2015/12/CFT 2011-10-00 EN.pdf.
- ——. 2021. "How Queer!? Canadian Approaches to Recognizing Queer Families in the Law." Whatever. A Transdisciplinary Journal of Queer Theories and Studies 4 (April): 303–28. https://doi.org/10.13131/2611-657X.whatever.v4i1.125.
- Harrington, Lee. 2016. *Traversing Gender: Understanding Transgender Realities*. Mystic Productions Press.
- Hartmann, Betsy. 1995. Reproductive Rights and Wrongs: The Global Politics of Population Control. South End Press.
- Hartnett, Caroline Sten, Lisa L. Lindley, and Katrina M. Walsemann. 2017. "Congruence across Sexual Orientation Dimensions and Risk for Unintended Pregnancy among Adult U.S. Women." *Women's Health Issues* 27 (2): 145-151.e2. https://doi.org/10.1016/j.whi.2016.10.010.
- Harvey, David. 1996. "David, Goliath and HIV-Infected Blood." *Canadian HIV/AIDS Policy and Law Newsletter* 2 (2): 9–11.
- Hashem, Nesrein M., Sameh A. Abdelnour, Ahmad R. Alhimaidi, and Ayman A. Swelum. 2021. "Potential Impacts of COVID-19 on Reproductive Health: Scientific Findings and Social Dimension." *Saudi Journal of Biological Sciences* 28 (3): 1702–12. https://doi.org/10.1016/j.sjbs.2020.12.012.
- Haukelid, K. 2008. "Theories of (Safety) Culture Revisited—An Anthropological Approach." *Safety Science*, Regulatory Issues, Safety Climate, Culture and Management Papers selected from the third international conference Working on Safety (WOS2006), September 12-15th, 2006, Zeewolde, The Netherland, 46 (3): 413–26. https://doi.org/10.1016/j.ssci.2007.05.014.
- Have, Henk ten, and Maria do Céu Patrão Neves. 2021. "Adoption." In *Dictionary of Global Bioethics*, edited by Henk ten Have and Maria do Céu Patrão Neves, 93–94. Cham: Springer International Publishing. https://doi.org/10.1007/978-3-030-54161-3\_43.
- Hay, Iain. 2020. "Zoom and Place: Video Conferencing and Virtual Geography." *South Australian Geographical Journal* 116 (1): 7–11. https://doi.org/10.3316/informit.719009113072212.
- Health Canada. 2022. "Health Canada Authorizes Canadian Blood Services' Submission to Eliminate Donor Deferral Period for Men Who Have Sex with Men." Statements. April 28, 2022. https://www.canada.ca/en/health-canada/news/2022/04/health-canada-authorizes-canadian-blood-services-submission-to-eliminate-donor-deferral-period-for-men-who-have-sex-with-men.html.
- Hechter, Michael, and Karl-Dieter Opp, eds. 2001. *Social Norms*. Russell Sage Foundation. Heiss, Jan Patrick, and Albert Piette. 2015. "Individuals in Anthropology." *Zeitschrift Für Ethnologie* 140 (1): 5–17.
- Hekma, Gert. 2000. "Queering Anthropology." In *Lesbian and Gay Studies: An Introductory, Interdisciplinary Approach*, edited by Theo Sandfort, Judith Schuyf, Jan Willem Duyvendak, and Jeffrey Weeks, 81–97. SAGE.
- Herman, Ellen. 2009. *Kinship by Design: A History of Adoption in the Modern United States. Kinship by Design*. University of Chicago Press. https://doi.org/10.7208/9780226328072.

- Herz, Marcus, and Thomas Johansson. 2015. "The Normativity of the Concept of Heteronormativity." *Journal of Homosexuality* 62 (8): 1009–20. https://doi.org/10.1080/00918369.2015.1021631.
- Herzog, Hanna. 2012. "Interview Location and Its Social Meaning." In *The SAGE Handbook of Interview Research: The Complexity of the Craft*, edited by Jaber F. Gubrium, James A. Holstein, Amir B. Marvasti, and Karyn D. McKinney, 207–17. California: SAGE Publications.
- Heyhoe, Jane, Yvonne Birks, Reema Harrison, Jane K O'Hara, Alison Cracknell, and Rebecca Lawton. 2016. "The Role of Emotion in Patient Safety: Are We Brave Enough to Scratch beneath the Surface?" *Journal of the Royal Society of Medicine* 109 (2): 52–58. https://doi.org/10.1177/0141076815620614.
- Hicks, Stephen. 2005. "Is Gay Parenting Bad for Kids? Responding to the 'Very Idea of Difference' in Research on Lesbian and Gay Parents." *Sexualities* 8 (2): 153–68. https://doi.org/10.1177/1363460705050852.
- Hjorth, Larissa. 2022. "Careful Digital Kinship: Understanding Multispecies Digital Kinship, Choreographies of Care and Older Adults During the Pandemic in Australia." *Communication, Culture and Critique* 15 (2): 227–43. https://doi.org/10.1093/ccc/tcac008.
- Hoffma, Susanna M., and Roberto E. Barrios. 2019. *Disaster Upon Disaster: Exploring the Gap Between Knowledge, Policy and Practice*. Berghahn Books.
- Holmes, J. Teresa. 2009. "When Blood Matters: Making Kinship in Colonial Kenya." In *Kinship and Beyond: The Genealogical Model Reconsidered*, edited by Sandra Bamford and James Leach, 15:50–83. London: Berghahn Books.
- Horne, Christine. 2001. "Sociological Perspectives on the Emergence of Norms." In *Social Norms*, edited by Michael Hechter and Karl-Dieter Opp, 3–34. New York: Russell Sage Foundation.
- Housman, Patty. 2022. "Roe v Wade Overturned: What It Means, What's Next." *American University*, June 29, 2022. https://www.american.edu/cas/news/roe-v-wade-overturned-what-it-means-whats-next.cfm.
- Hubbard, Ruth. 1996. "Gender and Genitals: Constructs of Sex and Gender." In *Science Wars*, edited by Andrew Ross. Durham, NC: Duke University Press. http://read.dukeupress.edu/books/book/1822/chapter/186910/Gender-and-GenitalsConstructs-of-Sex-and-Gender.
- Hudak, Nicole. 2021. "Negotiating Identity and Intimacy in Birth Control and Queer Pregnancy Research." In *Communicating Intimate Health*, edited by Angela Cooke-Jackson and Valerie Rubinsky. 251–262. New York: Rowman & Littlefield.
- Hudak, Nicole C. 2019. "Communicating Heterosexism in Queer Pregnancies: A Multiadic Interview Study." Doctorial Thesis, Athens, OH: Ohio University. https://search.proquest.com/openview/95f4efca32e68b119508094cc4c744a0/1?pq-origsite=gscholar&cbl=18750&diss=y&casa\_token=XoQ9k7Gn5h0AAAAA:y3Vymfhn4QZvVlaBJ5mqTf9HKZcWTHht6nCYHEwGob-FD1CsrpUSpjJbhBuriFQUNF6PKQG1Bb8.
- Huertas-Zurriaga, Ariadna, Patrick A Palmieri, Mariela P Aguayo-Gonzalez, Karen A Dominguez-Cancino, Cristina Casanovas-Cuellar, Kara L Vander Linden, Sandra K Cesario, Joan E Edwards, and Juan M Leyva-Moral. 2022. "Reproductive Decision-Making of Black Women Living with HIV: A Systematic Review." *Women's Health* 18 (January): 174550572210908. https://doi.org/10.1177/17455057221090827.

- Hultman, Martin, and Jonas Anshelm. 2017. "Masculinities of Global Climate Change: Exploring Ecomodern, Industrial and Ecological Masculinity." In *Climate Change and Gender in Rich Countries*. Routledge.
- Human Rights Watch. 2017. "'Just Let Us Be': Discrimination Against LGBT Students in the Philippines." Human Rights Watch. https://www.hrw.org/report/2017/06/21/just-let-us-be/discrimination-against-lgbt-students-philippines.
- Humphreys, Kathryn L., Myo Thwin Myint, and Charles H. Zeanah. 2020. "Increased Risk for Family Violence During the COVID-19 Pandemic." *Pediatrics* 146 (1): e20200982. https://doi.org/10.1542/peds.2020-0982.
- Hunt, Sarah, and Cindy Holmes. 2015. "Everyday Decolonization: Living a Decolonizing Queer Politics." *Journal of Lesbian Studies* 19 (2): 154–72. https://doi.org/10.1080/10894160.2015.970975.
- Hunter, Adam. 2022. "Woman Told to Change 'abortion Is Health Care' Shirt before Entering Sask. Legislature." *CBC News*, November 10, 2022. https://www.cbc.ca/news/canada/saskatchewan/sask-abortion-rights-1.6646675.
- Hutta, J Simon. 2009. "Geographies of Geborgenheit: Beyond Feelings of Safety and the Fear of Crime." *Environment and Planning D: Society and Space* 27 (2): 251–73. https://doi.org/10.1068/d3308.
- Huxley, Caroline, Victoria Clarke, and Emma Halliwell. 2014. "Resisting and Conforming to the 'Lesbian Look': The Importance of Appearance Norms for Lesbian and Bisexual Women." *Journal of Community & Applied Social Psychology* 24 (3): 205–19. https://doi.org/10.1002/casp.2161.
- Idier, Antoine. 2020. "What's a 'Norm' After Queer Movements?" Oxford Research Encyclopedia of Politics. October 27, 2020. https://doi.org/10.1093/acrefore/9780190228637.013.1191.
- Ingraham, Chrys. 1997. "The Heterosexual Imaginary Feminist Sociology and Theories of Gender." In *Materialist Feminism: A Reader in Class, Difference, and Women's Lives*, edited by Rosemary Hennessy and Chrys Ingraham, Sussex, UK, 275–90. Psychology Press.
- Jackman, Michael Connors, and Nishant Upadhyay. 2014. "Pinkwatching Israel, Whitewashing Canada: Queer (Settler) Politics and Indigenous Colonization in Canada." *Women's Studies Ouarterly* 42 (3/4): 195–210.
- Jackson, Taylor, Jason Clemens, and Milagros Palacios. 2017. "Canada's Aging Population and Implications for Government Finances." Fraser Institute. https://www.fraserinstitute.org/sites/default/files/canadas-aging-population-and-implications-for-government-finances.pdf.
- Jadoo, Saad Ahmed Ali. 2020. "COVID -19 Pandemic Is a Worldwide Typical Biopsychosocial Crisis." *Journal of Ideas in Health* 3 (2): 152–54. https://doi.org/10.47108/jidhealth.Vol3.Iss2.58.
- Jagose, Annamarie. 2013. Queer Theory. Melbourne: Melbourne University Publishing.
- James-Abra, S., L.A. Tarasoff, d. green, R. Epstein, S. Anderson, S. Marvel, L.S. Steele, and L.E. Ross. 2015. "Trans People's Experiences with Assisted Reproduction Services: A Qualitative Study." *Human Reproduction* 30 (6): 1365–74. https://doi.org/10.1093/humrep/dev087.

- Javaid, Aliraza. 2018. "Out of Place: Sexualities, Sexual Violence, and Heteronormativity." *Aggression and Violent Behavior* 39 (March): 83–89. https://doi.org/10.1016/j.avb.2018.02.007.
- Jiao, Wen Yan, Lin Na Wang, Juan Liu, Shuan Feng Fang, Fu Yong Jiao, Massimo Pettoello-Mantovani, and Eli Somekh. 2020. "Behavioral and Emotional Disorders in Children during the COVID-19 Epidemic." *The Journal of Pediatrics* 221 (June): 264-266.e1. https://doi.org/10.1016/j.jpeds.2020.03.013.
- Johnson, John M., and Timothy Rowlands. 2012. "The Interpersonal Dynamics of In-Depth Interviewing." In *The SAGE Handbook of Interview Research: The Complexity of the Craft*, edited by Jaber F. Gubrium, James A. Holstein, Amir B. Marvasti, and Karyn D. McKinney, 2nd ed., 99–113. Thousand Oaks, CA: SAGE. https://link.gale.com/apps/doc/CX3721700016/GVRL?sid=bookmark-GVRL&xid=3840ac19.
- Johnson, Sally. 2018. "'I See My Section Scar like a Battle Scar': The Ongoing Embodied Subjectivity of Maternity." *Feminism & Psychology* 28 (4): 470–87. https://doi.org/10.1177/0959353518769920.
- Jomeen, J., C. R. Martin, C. Jones, C. Marshall, S. Ayers, K. Burt, L. Frodsham, et al. 2021. "Tokophobia and Fear of Birth: A Workshop Consensus Statement on Current Issues and Recommendations for Future Research." *Journal of Reproductive and Infant Psychology* 39 (1): 2–15. https://doi.org/10.1080/02646838.2020.1843908.
- Jones, Abeni. 2020. "How to Take Care of Each Other: Community Care In Times Of Crisis." *Autostraddle* (blog). June 8, 2020. https://www.autostraddle.com/how-to-take-care-of-each-other-community-care-in-times-of-crisis-425239/.
- Jong, Elisabeth M. de, Niklas Ziegler, and Michaéla C. Schippers. 2020. "From Shattered Goals to Meaning in Life: Life Crafting in Times of the COVID-19 Pandemic." *Frontiers in Psychology* 11. https://www.frontiersin.org/articles/10.3389/fpsyg.2020.577708.
- Joy, Phillip, Megan Aston, Sheri Price, Meaghan Sim, Rachel Ollivier, Britney Benoit, Neda Akbari-Nassaji, and Damilola Iduye. 2020. "Blessings and Curses: Exploring the Experiences of New Mothers during the COVID-19 Pandemic." *Nursing Reports* 10 (2): 207–19. https://doi.org/10.3390/nursrep10020023.
- Jubran, Bellal, Maxime Billick, Gabriel Devlin, Jeremy Cygler, and Bertrand Lebouché. 2016. "Reevaluating Canada's Policy for Blood Donations from Men Who Have Sex with Men (MSM)." *Journal of Public Health Policy* 37 (4): 428–39. https://doi.org/10.1057/s41271-016-0032-1.
- Karaian, Lara. 2013. "Pregnant Men: Repronormativity, Critical Trans Theory and the Re(Conceive)Ing of Sex and Pregnancy in Law." *Social & Legal Studies* 22 (2): 211–30. https://doi.org/10.1177/0964663912474862.
- Karataş, Zeynep, and Özlem Tagay. 2021. "The Relationships between Resilience of the Adults Affected by the Covid Pandemic in Turkey and Covid-19 Fear, Meaning in Life, Life Satisfaction, Intolerance of Uncertainty and Hope." *Personality and Individual Differences* 172 (April): 110592. https://doi.org/10.1016/j.paid.2020.110592.
- Karki, Utkarsh, Gunjan Dhonju, and Arun Raj Kunwar. 2020. "Parenting During the COVID-19 Pandemic." *JNMA: Journal of the Nepal Medical Association* 58 (231): 957–59. https://doi.org/10.31729/jnma.5319.

- Katczynski, Amelie, Elaine Stratford, and Pauline Marsh. 2022. "Tracing Memories and Meanings of Festival Landscapes during the COVID-19 Pandemic." *Emotion, Space and Society*, June, 100903. https://doi.org/10.1016/j.emospa.2022.100903.
- Katsabian, Tammy. 2020. "The Telework Virus: How the COVID-19 Pandemic Has Affected Telework and Exposed Its Implications for Privacy and Equality." SSRN Scholarly Paper. Rochester, NY. https://doi.org/10.2139/ssrn.3684702.
- Kay, Jilly Boyce. 2020. "'Stay the Fuck at Home!': Feminism, Family and the Private Home in a Time of Coronavirus." *Feminist Media Studies* 20 (6): 883–88. https://doi.org/10.1080/14680777.2020.1765293.
- Kazyak, Emily, Brandi Woodell, Kristin Scherrer, and Emma Finken. 2018. "Law and Family Formation Among LGBQ-Parent Families." *Family Court Review* 56 (3): 364–73. https://doi.org/10.1111/fcre.12353.
- Kelly, Fiona. 2013. "One of These Families Is Not Like the Others: The Legal Response to Non-Normative Queer Parenting in Canada." *Alberta Law Review* 51 (1): 1–22.
- Kentlyn, Sue. 2008. "The Radically Subversive Space of the Queer Home: 'Safety House' and 'Neighbourhood Watch." *Australian Geographer* 39 (3): 327–37. https://doi.org/10.1080/00049180802270523.
- Khader, Serene J. 2013. "Intersectionality and the Ethics of Transnational Commercial Surrogacy." *IJFAB: International Journal of Feminist Approaches to Bioethics* 6 (1): 68–90. https://doi.org/10.3138/ijfab.6.1.68.
- Kia, Hannah, Margaret Robinson, Jenna MacKay, and Lori E. Ross. 2021. "Poverty in Lesbian, Gay, Bisexual, Transgender, Queer, Two-Spirit, and Other Sexual and Gender Minority (LGBTQ2S+) Communities in Canada: Implications for Social Work Practice." *Research on Social Work Practice* 31 (6): 584–98. https://doi.org/10.1177/1049731521996814.
- King, Wesley M., Jaclyn M. W. Hughto, and Don Operario. 2020. "Transgender Stigma: A Critical Scoping Review of Definitions, Domains, and Measures Used in Empirical Research." *Social Science & Medicine* 250 (April): 112867. https://doi.org/10.1016/j.socscimed.2020.112867.
- Kirsch, Max H. 2000. *Queer Theory and Social Change*. Opening Up. London: Routledge. https://doi.org/10.4324/9780203131060.
- Kislenko, Cassandra. 2022. "Is Canada's Blood Ban Really over? | Xtra Magazine." December 6, 2022. https://xtramagazine.com/health/blood-ban-donating-blood-in-canada-241695.
- Klesse, Christian. 2019. "Polyamorous Parenting: Stigma, Social Regulation, and Queer Bonds of Resistance." *Sociological Research Online* 24 (4): 625–43. https://doi.org/10.1177/1360780418806902.
- Klobas, Jane E., and Icek Ajzen. 2015. "Making the Decision to Have a Child." In *Reproductive Decision-Making in a Macro-Micro Perspective*, edited by Dimiter Philipov, Aart C. Liefbroer, and Jane E. Klobas. Dordrecht: Springer Netherlands. https://doi.org/10.1007/978-94-017-9401-5.
- Korinek, Valerie J. 2018. *Prairie Fairies: A History of Queer Communities and People in Western Canada, 1930-1985.* Toronto: University of Toronto Press. https://books-scholarsportal-info.cyber.usask.ca/en/read?id=/ebooks/ebooks3/utpress/2018-08-15/1/9781487518172.
- Kradin, Richard. 2009. "The Family Myth: Its Deconstruction and Replacement with a Balanced Humanized Narrative." *Journal of Analytical Psychology* 54 (2): 217–32. https://doi.org/10.1111/j.1468-5922.2009.01771.x.

- Kurz, Larissa. 2022. "Not Quite There: Abortion Pill Access in Sask. Still Limited." *Regina Leader-Post*, June 18, 2022. https://leaderpost.com/news/saskatchewan/not-quite-there-abortion-pill-access-in-sask-still-limited.
- Lacey, Sheryl de. 2014. "Recruiting Men to Research about Reproduction: A Fruitless Goal or a Challenge?" *Australian Nursing and Midwifery Journal* 22 (5): 43. https://doi.org/10.3316/informit.688311774883773.
- Lam, Peggy. 2022. "Canadian Blood Services to End Blood Ban for Men Who Have Sex with Men | CBC News." *CBC*, April 28, 2022. https://www.cbc.ca/news/health/health-canada-blood-ban-1.6434078.
- Langager, Brody. 2022. "Sask. Government among Others Added to \$25 Million Church-Run School Lawsuit | Globalnews.Ca." *Global News*, December 13, 2022. https://globalnews.ca/news/9344857/sask-government-25-million-church-school-lawsuit/.
- Lateef, Rusan, Ramona Alaggia, and Delphine Collin-Vézina. 2021. "A Scoping Review on Psychosocial Consequences of Pandemics on Parents and Children: Planning for Today and the Future." *Children and Youth Services Review* 125 (June): 106002. https://doi.org/10.1016/j.childyouth.2021.106002.
- Lavery, Irelyne. 2022. "LGBTQ2 People Will Face Disproportionate Impacts If Roe v. Wade Is Overturned: Experts." *Global News*, May 7, 2022. https://globalnews.ca/news/8814001/lgbtq2s-people-disproportionate-impacts-roe-v-wade-overturned-experts/.
- Law, Caroline. 2019. "Men on the Margins? Reflections on Recruiting and Engaging Men in Reproduction Research." *Methodological Innovations* 12 (1): 2059799119829425. https://doi.org/10.1177/2059799119829425.
- Lawson, Danielle F., Kathryn T. Stevenson, M. Nils Peterson, Sarah J. Carrier, Erin Seekamp, and Renee Strnad. 2019. "Evaluating Climate Change Behaviors and Concern in the Family Context." *Environmental Education Research* 25 (5): 678–90. https://doi.org/10.1080/13504622.2018.1564248.
- Lebow, Jay L. 2020. "Family in the Age of COVID-19." *Family Process* 59 (2): 309–12. https://doi.org/10.1111/famp.12543.
- Lehr, Valerie. 1999. *Queer Family Values: Debunking the Myth of the Nuclear Family*. Temple University Press.
- Leibetseder, Doris, and Gabriele Griffin. 2018. "Introduction: Queer and Trans Reproduction with Assisted Reproductive Technologies (ART), in Europe." *Journal of International Women's Studies* 20 (1): 10.
- Leonetti, Donna L., and Benjamin Chabot-Hanowell. 2011. "The Foundation of Kinship." *Human Nature* 22 (1): 16–40. https://doi.org/10.1007/s12110-011-9111-y.
- Leroux, Chad. 2021. "Discrimination Still Happens despite Growing LGBTQ2S+ Acceptance: OUTSaskatoon." *CTV News*, May 17, 2021. https://saskatoon.ctvnews.ca/discrimination-still-happens-despite-growing-lgbtq2s-acceptance-outsaskatoon-1.5431913.
- Levine, Nancy E. 2008. "Alternative Kinship, Marriage, and Reproduction." *Annual Review of Anthropology* 37 (1): 375–89. https://doi.org/10.1146/annurev.anthro.37.081407.085120.
- Levitt, Peggy. 2018. "Creation National and Global Citizens: What Role Can Museums Play?" In *Museums in a Time of Migration: Rethinking Museum's Roles, Representations, Collections, and Collaborations*, edited by Pieter Bevelander and Christina Johansson, 31–57. Nordic Academic Press, Sweden.

- Leyva-Moral, Juan M., Patrick A. Palmieri, María Feijoo-Cid, Sandra K. Cesario, Nataly Julissa Membrillo-Pillpe, Patricia N. Piscoya-Angeles, Marilyn Goff, Ana Toledo-Chavarri, and Joan E. Edwards. 2018. "Reproductive Decision-Making in Women Living with Human Immunodeficiency Virus: A Systematic Review." *International Journal of Nursing Studies* 77 (January): 207–21. https://doi.org/10.1016/j.ijnurstu.2017.10.012.
- Library and Archives Canada. 2021. "Acts of Divorce, 1841-1968 Library and Archives Canada." April 30, 2021. https://www.bac-lac.gc.ca/eng/discover/vital-statistics-births-marriages-deaths/divorce-1841-1968/Pages/acts-divorce-1841-1968.aspx.
- Lilja, Mona. 2013. Resisting Gendered Norms: Civil Society, the Juridical and Political Space in Cambodia. New York: Taylor & Francis. http://www.routledge.com/Resisting-Gendered-Norms-Civil-Society-the-Juridical-and-Political-Space/Lilja/p/book/9781138278998.
- Lin, Tracy Kuo, Rachel Law, Jessica Beaman, and Diana Greene Foster. 2021. "The Impact of the COVID-19 Pandemic on Economic Security and Pregnancy Intentions among People at Risk of Pregnancy." *Contraception* 103 (6): 380–85. https://doi.org/10.1016/j.contraception.2021.02.001.
- Linton, Steven J., Ida K. Flink, Emma Nilsson, and Sara Edlund. 2017. "Can Training in Empathetic Validation Improve Medical Students' Communication with Patients Suffering Pain? A Test of Concept." *Pain Reports* 2 (3): e600. https://doi.org/10.1097/PR9.0000000000000000000.
- Long, Nicholas J., and Henrietta L. Moore. 2013. *Sociality: New Directions*. Berghahn Books. Loughlin, Gerard. 2018. "Catholic Homophobia." *Theology* 121 (3): 188–96. https://doi.org/10.1177/0040571X17749147.
- Love, Gillian. 2022. "Trans Pregnancy in a Repronormative World." In *Towards Gender Equality in Law: An Analysis of State Failures from a Global Perspective*, edited by Gizem Guney, David Davies, and Po-Han Lee, 35–58. Cham: Springer International Publishing. https://doi.org/10.1007/978-3-030-98072-6\_3.
- Lunny, Allyson M. 2013. "Look, a Faggot!': The Scopic Economies of Cruising, Queer Bashing and Law." *Social & Legal Studies* 22 (2): 231–45. https://doi.org/10.1177/0964663912465638.
- Luttik, Marie Louise, Cristina Garcia-Vivar, Hanne Konradsen, Romy Mahrer-Imhof, Lorenz Imhof, Anne Brodsgaard, Birte Ostergaard, Karin Dieperink, and Erla Kolbrun-Svavarsdottir. 2020. "The COVID-19 Pandemic: A Family Affair." *Journal of Family Nursing* 26 (2): 87–89. https://doi.org/10.1177/1074840720920883.
- Maas, Jolanda, Peter Spreeuwenberg, Marijke van Winsum-Westra, Robert A Verheij, Sjerp Vries, and Peter P Groenewegen. 2009. "Is Green Space in the Living Environment Associated with People's Feelings of Social Safety?" *Environment and Planning A: Economy and Space* 41 (7): 1763–77. https://doi.org/10.1068/a4196.
- Macintosh, Heather, Elke D Reissing, and Heather Andruff. 2010. "Same-Sex Marriage in Canada: The Impact of Legal Marriage on the First Cohort of Gay and Lesbian Canadians to Wed" 19 (3): 14.
- Maestripieri, Lara. 2021. "The Covid-19 Pandemics: Why Intersectionality Matters." *Frontiers in Sociology* 6. https://www.frontiersin.org/articles/10.3389/fsoc.2021.642662.
- Mahler, Sarah J., Mayurakshi Chaudhuri, and Vrushali Patil. 2015. "Scaling Intersectionality: Advancing Feminist Analysis of Transnational Families." *Sex Roles* 73 (3): 100–112. https://doi.org/10.1007/s11199-015-0506-9.

- Maier, Karl J. 2021. "Equity, Environment, and the Biopsychosocial Ecology of the COVID-19 Syndemic." *Psychosomatic Medicine* 83 (9): 1089–91. https://doi.org/10.1097/PSY.000000000001020.
- Malicka, Izabela, Monika Mynarska, and Joanna Świderska. 2021. "Perceived Consequences of the COVID-19 Pandemic and Childbearing Intentions in Poland." *Journal of Family Research* 33 (3): 674–702. https://doi.org/10.20377/jfr-666.
- Mamo, Laura. 2007. *Queering Reproduction: Achieving Pregnancy in the Age of Technoscience*. Duke University Press.
- Mamo, Laura, and Eli Alston-Stepnitz. 2015. "Queer Intimacies and Structural Inequalities: New Directions in Stratified Reproduction." *Journal of Family Issues* 36 (4): 519–40. https://doi.org/10.1177/0192513X14563796.
- Manning, Jimmie, and Danielle M. Stern. 2018. "Heteronormative Bodies, Queer Futures: Toward a Theory of Interpersonal Panopticism." *Information, Communication & Society* 21 (2): 208–23. https://doi.org/10.1080/1369118X.2016.1271901.
- Manzanedo, Rubén D., and Peter Manning. 2020. "COVID-19: Lessons for the Climate Change Emergency." *Science of The Total Environment* 742 (November): 140563. https://doi.org/10.1016/j.scitotenv.2020.140563.
- Marazziti, Donatella, Paolo Cianconi, Federico Mucci, Lara Foresi, Ilaria Chiarantini, and Alessandra Della Vecchia. 2021. "Climate Change, Environment Pollution, COVID-19 Pandemic and Mental Health." *Science of The Total Environment* 773 (June): 145182. https://doi.org/10.1016/j.scitotenv.2021.145182.
- Martin, Emily. 2001. The Woman in the Body: A Cultural Analysis of Reproduction. Beacon Press.
- Martin, Karin A. 2009. "Normalizing Heterosexuality: Mothers' Assumptions, Talk, and Strategies with Young Children." *American Sociological Review* 74 (2): 190–207. https://doi.org/10.1177/000312240907400202.
- Martin, Karin A., and Katherine Luke. 2010. "Gender Differences in the ABC's of the Birds and the Bees: What Mothers Teach Young Children About Sexuality and Reproduction." *Sex Roles* 62 (3): 278–91. https://doi.org/10.1007/s11199-009-9731-4.
- Marvel, Stu. 2016. "Laws of Conception: A Queer Genealogy of Canada's Assisted Human Reproduction Act Symposium: New Approaches & Challenges to Reproductive Justice." *FIU Law Review* 12 (1): 81–116.
- Massey, Doreen. 1991. "A Global Sense Of Place." Marxism Today 38: 24–29.
- Matlock, Kate. 2012. "Digital Anthropology." In *Share This Too*, 41–48. John Wiley & Sons, Ltd. https://doi.org/10.1002/9781119207993.ch5.
- Matsuno, Emmie, and Stephanie L. Budge. 2017. "Non-Binary/Genderqueer Identities: A Critical Review of the Literature." *Current Sexual Health Reports* 9 (3): 116–20. https://doi.org/10.1007/s11930-017-0111-8.
- Mattingly, Cheryl. 2019. "Defrosting Concepts, Destabilizing Doxa: Critical Phenomenology and the Perplexing Particular." *Anthropological Theory* 19 (4): 415–39. https://doi.org/10.1177/1463499619828568.
- Mavuso, Jabulile Mary-Jane Jace. 2022. "Repronormativity in Cisgender Men's Reasons Why They Would Not Use Womb Transplant Technology to Become Pregnant." *Sociology Compass*, e13054. https://doi.org/10.1111/soc4.13054.
- Mazzei, Julie, and Erin E. O'Brien. 2009. "You Got It, So When Do You Flaunt It?: Building Rapport, Intersectionality, and the Strategic Deployment of Gender in the Field." *Journal*

- of Contemporary Ethnography 38 (3): 358–83. https://doi.org/10.1177/0891241608330456.
- McCann, Kristian. 2018. "Fun Fact: Belgium Was the World's Second Country to Legalise Same-Sex Marriage." *Brussels Express*, February 15, 2018. https://brussels-express.eu/fun-fact-belgium-was-the-worlds-second-country-to-legalise-same-sex-marriage/.
- McDermott, Elizabeth, Elizabeth Hughes, and Victoria Rawlings. 2018. "Norms and Normalisation: Understanding Lesbian, Gay, Bisexual, Transgender and Queer Youth, Suicidality and Help-Seeking." *Culture, Health & Sexuality* 20 (2): 156–72. https://doi.org/10.1080/13691058.2017.1335435.
- McDonald, James. 2013. "Conforming to and Resisting Dominant Gender Norms: How Male and Female Nursing Students Do and Undo Gender." *Gender, Work & Organization* 20 (5): 561–79. https://doi.org/10.1111/j.1468-0432.2012.00604.x.
- McKenzie, Lara. 2022. "Parenthood: Beyond Maternity and Paternity." *Feminist Anthropology* 3 (2): 299–306. https://doi.org/10.1002/fea2.12105.
- Mellström, Ulf. 2020. "COVID-19, Masculinity and Risk/at Risk." *NORMA* 15 (2): 94–96. https://doi.org/10.1080/18902138.2020.1762307.
- Merriam-Webster. 2021a. "Definition of RISK." In . Encyclopædia Britannica. https://www.merriam-webster.com/dictionary/risk.
- ——. 2021b. "Definition of Safe." In *Merriam-Webster*. Encyclopædia Britannica. https://www.merriam-webster.com/dictionary/safe.
- Meulen, Emily van der, and Robert Heynen, eds. 2016. Expanding the Gaze: Gender and the Politics of Surveillance. University of Toronto Press.
- Meyer, David D. 2006. "Parenthood in a Time of Transition: Tensions between Legal, Biological, and Social Conceptions of Parenthood." *The American Journal of Comparative Law* 54: 125–44.
- Meyer, Doug. 2015. Violence Against Queer People: Race, Class, Gender, and the Persistence of Anti-LGBT Discrimination. Rutgers University Press.
- Michelson, Rebecca, Akeiylah DeWitt, Ria Nagar, Alexis Hiniker, Jason Yip, Sean A. Munson, and Julie A. Kientz. 2021. "Parenting in a Pandemic: Juggling Multiple Roles and Managing Technology Use in Family Life During COVID-19 in the United States." *Proceedings of the ACM on Human-Computer Interaction* 5 (CSCW2): 402:1-402:39. https://doi.org/10.1145/3479546.
- Miller, Shaeleya D., Verta Taylor, and Leila J. Rupp. 2016. "Social Movements and the Construction of Queer Identity." In *New Directions in Identity Theory and Research*, edited by Jan E. Stets and Richard T. Serpe, 443–69. Oxford University Press.
- Miller, W. B. 1994. "Reproductive Decisions: How We Make Them and How They Make Us." *Advances in Population: Psychosocial Perspectives* 2: 1–27.
- Minister of Education. 2012. "OC 415/2012 The Education Funding Amendment Regulations, 2012 (Minister of Education)." https://publications.saskatchewan.ca/#/products/64424.
- . 2018. "OC 234/2018 The Registered Independent Schools Regulations/Règlement Sur Les Écoles Indépendantes Inscrites (Minister of Education)." https://publications.saskatchewan.ca/#/products/90023.
- ——. 2022. "Registered Independent Schools Regulations, E-0.2 Reg 27." https://publications.saskatchewan.ca/#/products/91108.

- Modjeski, Morgan. 2016. "Victim of Alleged Gay Bashing Calls for Education, Not Violence." *The StarPhoenix*, September 13, 2016. https://thestarphoenix.com/news/local-news/police-encourage-reporting-despite-incidents-of-hate-crime-being-very-rare.
- Mollon, Lea. 2012. "The Forgotten Minorities: Health Disparities of the Lesbian, Gay, Bisexual, and Transgendered Communities." *Journal of Health Care for the Poor and Underserved* 23 (1): 1–6. https://doi.org/10.1353/hpu.2012.0009.
- Montero-Ossandón, Luz Eugenia, Paz Valverde-Forttes, Angelina María Dois-Castellón, Mauricio Bicocca-Gino, and Carmen Aída Domínguez-Hidalgo. 2017. "Sex Education: A Challenge for Catholic Education." *Educación y Educadores* 20 (3): 341–63. https://doi.org/10.5294/edu.2017.20.3.1.
- Moore, Julia. 2021. "The Fixed Childfree Subjectivity: Performing Meta-Facework about Sterilization on Reddit." *Health Communication* 36 (12): 1527–36. https://doi.org/10.1080/10410236.2020.1773697.
- Moore, Mignon R., and Michael Stambolis-Ruhstorfer. 2013. "LGBT Sexuality and Families at the Start of the Twenty-First Century." *Annual Review of Sociology* 39 (1): 491–507. https://doi.org/10.1146/annurev-soc-071312-145643.
- Moran, Leslie J. 2002. "The Poetics of Safety: Lesbians, Gay Men and Home." In *Crime and Insecurity*, edited by Adam Crawford. London: Willan.
- Moreau, Greg, and Statistics Canada. 2021. "Police-Reported Hate Crime in Canada, 2019." May 29, 2021. http://www150.statcan.gc.ca/n1/pub/85-002-x/2021001/article/00002-eng.htm.
- Moreton-Robinson, Aileen. 2004. "Whiteness, Epistemology and Indigenous Representation." In *Whitening Race: Essays in Social and Cultural Criticism*, 75–88. Aboriginal Studies Press.
- Morgan, Lewis Henry. 1871. *Systems of Consanguinity and Affinity of the Human Family*. Washington: Smithsonian Institution.
- Mulder, Clara H. 2006. "Home-Ownership and Family Formation." *Journal of Housing and the Built Environment* 21 (3): 281–98. https://doi.org/10.1007/s10901-006-9050-9.
- Munn, Nancy D. 1990. "Constructing Regional Worlds in Experience: Kula Exchange, Witchcraft and Gawan Local Events." *Man* 25 (1): 1–17. https://doi.org/10.2307/2804106.
- Musial, Jennifer. 2014. "From 'Madonna' to 'Whore': Sexuality, Pregnancy, and Popular Culture." *Sexualities* 17 (4): 394–411. https://doi.org/10.1177/1363460713516335.
- Mykhalovskiy, Eric, Cécile Kazatchkine, Annie Foreman-Mackey, Alexander McClelland, Ryan Peck, Colin Hastings, and Richard Elliott. 2020. "Human Rights, Public Health and COVID-19 in Canada." *Canadian Journal of Public Health* 111 (6): 975–79. https://doi.org/10.17269/s41997-020-00408-0.
- Nanda, Kavita, Elena Lebetkin, Markus J. Steiner, Irina Yacobson, and Laneta J. Dorflinger. 2020. "Contraception in the Era of COVID-19." *Global Health: Science and Practice* 8 (2): 166–68. https://doi.org/10.9745/GHSP-D-20-00119.
- Naya, Christine H., Darby E. Saxbe, and Genevieve F. Dunton. 2021. "Early Effects of the COVID-19 Pandemic on Fertility Preferences in the United States: An Exploratory Study." *Fertility and Sterility* 116 (4): 1128–38. https://doi.org/10.1016/j.fertnstert.2021.05.092.
- Neiterman, Elena. 2013. "Pregnant Bodies in Social Context: Natural, Disruptive, and Unrecognized Pregnancy." *Symbolic Interaction* 36 (3): 335–50. https://doi.org/10.1002/symb.71.
- Nejaime, Douglas. 2017. "The Nature of Parenthood." *The Yale Law Journal* 126 (8): 2260–2381.

- Nelson, Joshua. 2020. "Petro-Masculinity and Climate Change Denial among White, Politically Conservative American Males." *International Journal of Applied Psychoanalytic Studies* 17 (4): 282–95. https://doi.org/10.1002/aps.1638.
- Newman, Brooke N. 2018. *Dark Inheritance: Blood, Race, and Sex in Colonial Jamaica*. Yale University Press.
- Noelle, Monique. 2002. "The Ripple Effect of the Matthew Shepard Murder: Impact on the Assumptive Worlds of Members of the Targeted Group." *American Behavioral Scientist* 46 (1): 27–50. https://doi.org/10.1177/0002764202046001004.
- Nygren, Katarina Giritli, Susanna Öhman, and Anna Olofsson. 2016. "Everyday Places, Heterosexist Spaces and Risk in Contemporary Sweden." *Culture, Health & Sexuality* 18 (1): 45–57. https://doi.org/10.1080/13691058.2015.1063814.
- Oliffe, John L., Mary T. Kelly, Gabriela Gonzalez Montaner, and Wellam F. Yu Ko. 2021. "Zoom Interviews: Benefits and Concessions." *International Journal of Qualitative Methods* 20 (January): 16094069211053522. https://doi.org/10.1177/16094069211053522.
- Olofsson, Anna, Jens O. Zinn, Gabriele Griffin, Katarina Giritli Nygren, Andreas Cebulla, and Kelly Hannah-Moffat. 2014. "The Mutual Constitution of Risk and Inequalities: Intersectional Risk Theory." *Health, Risk & Society* 16 (5): 417–30. https://doi.org/10.1080/13698575.2014.942258.
- Ontario Human Rights Commission. 2006. "The Changing Face of Canadian Families." 2006. https://www.ohrc.on.ca/en/human-rights-and-family-ontario/changing-face-canadian-families.
- Oriana Gonzalez, Danielle Alberti. 2022. "The Political Leanings of the Supreme Court Justices." Axios. January 26, 2022. https://www.axios.com/2019/06/01/supreme-court-justices-ideology.
- Orlando, Laura, Ashley N. Rousson, Susan Barkan, Kristen Greenley, Alyssa Everitt, and Emiko A. Tajima. 2021. "Responding to COVID-19's Impact on Supervised Family Time: The Supportive Virtual Family Time Model." *Developmental Child Welfare* 3 (1): 3–19. https://doi.org/10.1177/25161032211001078.
- Ortner, Sherry B. 2016. "Dark Anthropology and Its Others." *HAU: Journal of Ethnographic Theory* 6 (1): 47–73. https://doi.org/10.14318/hau6.1.004.
- Otto, Dianne. 2017. "Resisting the Heteronormative Imaginary of the Nation-State: Rethinking Kinship and Border Protection." In *Queering International Law*. Routledge.
- OUTSaskatoon. 2020. "Election Report Card on 2SLGBTQ Topics." Google Docs. October 20, 2020.
  - https://docs.google.com/document/d/1pTqtm9HjlWPzC7yUsapHXDTyG4mWAQkp8kkt WPr9AKM/edit?usp=embed facebook.
- Paine, Sarah-Jane, Monique Priston, T Leigh Signal, Bronwyn Sweeney, and Diane Muller. 2013. "DEVELOPING NEW APPROACHES FOR THE RECRUITMENT AND RETENTION OF INDIGENOUS PARTICIPANTS IN LONGITUDINAL RESEARCH: Lessons from E Moe, Ma–Ma–: Maternal Sleep." *MAI Journal* 2 (2): 12.
- Palmer, Lisa, and Susanna Barnes, dirs. 2021. *HOLDING TIGHTLY Custom and Healing in Timor-Leste*. https://vimeo.com/ondemand/holdingtightly.
- Patton, Michael Quinn. 1980. *Qualitative Evaluation Methods*. Beverly Hills CA: SAGE Publications.

- Peletz, Michael G. 2001. "Ambivalence in Kinship since the 1940s." In *Relative Values: Reconfiguring Kinship Studies*, edited by Sarah Franklin and Susan McKinnon. Durham, NC: Duke University Press.
- Peltz, Jack S., Dev Crasta, Jennifer S. Daks, and Ronald D. Rogge. 2021. "Shocks to the System: The Influence of COVID-19–Related Stressors on Coparental and Family Functioning." *Developmental Psychology* 57 (10): 1693–1707. https://doi.org/10.1037/dev0001206.
- Pender, Lisa Jane. 2007. "Birds & Bees: How Nature and Kinship Are Mobilized to Support Nuclear Family Narratives on Fertility Clinic Websites." Thesis. https://dspace.library.uvic.ca/handle/1828/1252.
- Pereira, Naveen L., Ferhaan Ahmad, Mirnela Byku, Nathan W. Cummins, Alanna A. Morris, Anjali Owens, Sony Tuteja, and Sharon Cresci. 2021. "COVID-19: Understanding Inter-Individual Variability and Implications for Precision Medicine." *Mayo Clinic Proceedings* 96 (2): 446–63. https://doi.org/10.1016/j.mayocp.2020.11.024.
- Peters, Wendy. 2001. "Queer Identities: Rupturing Identity Categories and Negotiating Meanings of Queer." *Canadian Woman Studies* 24 (2, 3): 102–7.
- Petersen, Cynthia. 1991. "A Queer Response to Bashing: Legislating against Hate Symposium on Lesbian and Gay Legal Issues." *Queen's Law Journal* 16 (2): 237–60.
- Peterson, Cassie. 2013. "The Lies That Bind: Heteronormative Constructions of 'Family' in Social Work Discourse." *Journal of Gay & Lesbian Social Services* 25 (4): 486–508. https://doi.org/10.1080/10538720.2013.829394.
- Pettinicchio, David. 2010. "Public and Elite Policy Preferences: Gay Marriage in Canada." *International Journal of Canadian Studies / Revue Internationale d'études Canadiennes*, no. 42: 125–53. https://doi.org/10.7202/1002175ar.
- Philipov, Dimiter, Aart C. Liefbroer, and Jane E. Klobas, eds. 2015. *Reproductive Decision-Making in a Macro-Micro Perspective*. Dordrecht: Springer Netherlands. https://doi.org/10.1007/978-94-017-9401-5.
- Phillips, Chas. 2009. "Difference, Disagreement and the Thinking of Queerness." *Borderlands* 8 (2): 17.
- Pitman, Gayle E. 2019. The Stonewall Riots: Coming Out in the Streets. Abrams.
- Planned Parenthood. 2022. "What Do Transgender and Cisgender Mean?" 2022. https://www.plannedparenthood.org/learn/teens/all-about-sex-gender-and-gender-identity/what-do-transgender-and-cisgender-mean.
- n.d. "What Is Intersex? Definition of Intersexual." Accessed August 4, 2022. https://www.plannedparenthood.org/learn/gender-identity/sex-gender-identity/whats-intersex.
- Platto, Sara, Jinfeng Zhou, Yanqing Wang, Huo Wang, and Ernesto Carafoli. 2021. "Biodiversity Loss and COVID-19 Pandemic: The Role of Bats in the Origin and the Spreading of the Disease." *Biochemical and Biophysical Research Communications*, COVID-19, 538 (January): 2–13. https://doi.org/10.1016/j.bbrc.2020.10.028.
- Podjed, Dan. 2021. "Renewal of Ethnography in the Time of the COVID-19 Crisis." *Sociologija i Prostor* 59 (March): 267–84. https://doi.org/10.5673/sip.59.0.10.
- Pohshna, Lede E Miki. 2020. "There Is No Home, Pig': Examining the Dilemma of Northeast Queer in Time of Covid-19." *Rupkatha Journal on Interdisciplinary Studies in Humanities* 12 (5). https://doi.org/10.21659/rupkatha.v12n5.rioc1s17n6.
- Pool, Robert. 2017. "The Verification of Ethnographic Data." *Ethnography* 18 (3): 281–86. https://doi.org/10.1177/1466138117723936.

- Portillo, Shannon K., Nicole Humphrey, and Domonic A. Bearfield. 2022. *The Myth of Bureaucratic Neutrality: An Examination of Merit and Representation*. Taylor & Francis.
- Poushter, Jacob, and Nicholas Kent. 2020. "The Global Divide on Homosexuality Persists." *Pew Research Center's Global Attitudes Project* (blog). June 25, 2020.
  - https://www.pewresearch.org/global/2020/06/25/global-divide-on-homosexuality-persists/.
- Power, Kate. 2020. "The COVID-19 Pandemic Has Increased the Care Burden of Women and Families." *Sustainability: Science, Practice and Policy* 16 (1): 67–73. https://doi.org/10.1080/15487733.2020.1776561.
- Prime, Heather, Mark Wade, and Dillon T. Browne. 2020. "Risk and Resilience in Family Well-Being during the COVID-19 Pandemic." *American Psychologist* 75 (5): 631–43. https://doi.org/10.1037/amp0000660.
- Public Health Agency of Canada. 2022. "Summary of Evidence Supporting COVID-19 Public Health Measures." Transparency other. June 27, 2022. https://www.canada.ca/en/public-health/services/diseases/2019-novel-coronavirus-infection/guidance-documents/summary-evidence-supporting-covid-19-public-health-measures.html.
- Quathamer, Natalie, and Phillip Joy. 2022. "Being in a Queer Time: Exploring the Influence of the COVID-19 Pandemic on LGBTQ+ Body Image." *Nutrition & Dietetics* 79 (3): 400–410. https://doi.org/10.1111/1747-0080.12699.
- Rabo, Annika. 1997. "Free to Make the Right Choice?" In *Anthropology of Policy*, edited by Cris Shore and Susan Wright, 83–103. Routledge.
- Rasche, Teagan. 2022a. "Possible Overturn of Roe v. Wade Reigniting Abortion Access Conversation in Saskatchewan." *Global News*, May 6, 2022. https://globalnews.ca/news/8816337/possible-overturn-roe-v-wade-abortion-access-saskatchewan/.
- 2022b. "Saskatchewan Abortion Rights Advocates Concerned about Impact of Decision to Overturn Roe v. Wade." *Global News*, June 24, 2022. https://globalnews.ca/news/8946567/saskatchewan-abortion-rights-advocates-concerned-about-impact-of-decision-to-overturn-roe-v-wade/.
- Rashawn, Ray, and Andre M. Perry. 2020. "Why We Need Reparations for Black Americans." *Brookings* (blog). April 15, 2020. https://www.brookings.edu/policy2020/bigideas/whywe-need-reparations-for-black-americans/.
- Razak, Fahad, Saeha Shin, C. David Naylor, and Arthur S. Slutsky. 2022. "Canada's Response to the Initial 2 Years of the COVID-19 Pandemic: A Comparison with Peer Countries." *CMAJ* 194 (25): E870–77. https://doi.org/10.1503/cmaj.220316.
- Read, Dwight, Michael D. Fischer, and F. K. Lehman (Chit Hlaing). 2014. "The Cultural Grounding of Kinship." *L'Homme. Revue Française d'anthropologie*, no. 210 (May): 63–89. https://doi.org/10.4000/lhomme.23550.
- Redfern, Corinne. 2021. "'I'm Scared Every Damn Day': In the Philippines, Violence Shadows Trans Lives." *The Fuller Project*, January 7, 2021. https://fullerproject.org/story/im-scared-every-damn-day-in-the-philippines-violence-shadows-trans-lives/.
- Reid, Graeme, and Samuel Ritholtz. 2020. "A Queer Approach to Understanding LGBT Vulnerability during the COVID-19 Pandemic." *Politics & Gender* 16 (4): 1101–9. https://doi.org/10.1017/S1743923X20000707.
- Reimann, Professor Renate. 2001. Queer Families, Queer Politics: Challenging Culture and the State. Columbia University Press.

- Riggs, Damien W. 2012. "Paradoxes of Visibility': Lesbian and Gay Parents in the Australian Print Media." *Jindal Global Law Review* 4 (1): 19.
- Rio, Carlos del, Saad B. Omer, and Preeti N. Malani. 2022. "Winter of Omicron—The Evolving COVID-19 Pandemic." *JAMA* 327 (4): 319–20. https://doi.org/10.1001/jama.2021.24315.
- Risch, Gail S., and Michael G. Lawler. 2003. "Sexuality Education and the Catholic Teenager: A Report." *Catholic Education: A Journal of Inquiry and Practice* 7 (1). https://ejournals.bc.edu/index.php/cej/article/view/475.
- Robinson, Brandon Andrew. 2016. "Heteronormativity and Homonormativity." In *The Wiley Blackwell Encyclopedia of Gender and Sexuality Studies*, edited by Angela Wong, Maithree Wickramasinghe, renee hoogland, and Nancy A Naples, 1–3. Singapore: John Wiley & Sons, Ltd. https://doi.org/10.1002/9781118663219.wbegss013.
- Rodman, Margaret C. 1992. "Empowering Place: Multilocality and Multivocality." *American Anthropologist* 94 (3): 640–56.
- Ross, Loretta, and Rickie Solinger. 2017. *Reproductive Justice: An Introduction*. Oakland, California: Univ of California Press.
- Ross, Lori, Rachel Epstein, Corrie Goldfinger, Leah Steele, Scott Anderson, and Carol Strike. 2008. "Lesbian and Queer Mothers Navigating the Adoption System: The Impacts on Mental Health." *Health Sociology Review* 17 (3): 254–66. https://doi.org/10.5172/hesr.451.17.3.254.
- Ross, Lori, Amy Siegel, Cheryl Dobinson, Rachel Epstein, and Leah S. Steele. 2012. "I Don't Want to Turn Totally Invisible': Mental Health, Stressors, and Supports among Bisexual Women during the Perinatal Period." *Journal of GLBT Family Studies* 8 (2): 137–54. https://doi.org/10.1080/1550428X.2012.660791.
- Rothblum, Esther D., ed. 2020. *The Oxford Handbook of Sexual and Gender Minority Mental Health*. Oxford: Oxford University Press.
- Rotman, Youval. 2021. "The Relational Mind: In between History, Psychology and Anthropology." *History of Psychology* 24: 142–63. https://doi.org/10.1037/hop0000175.
- Roulston, Chris. 2021. "Queer Parenting and the Challenge to Queer Theory." *Studies in Canadian Literature / Études En Littérature Canadienne* 46 (1): 117–44. https://doi.org/10.7202/1086613ar.
- Rowe, Aimee Carrillo. 2017. "A Queer Indigenous Manifesto." *QED: A Journal in GLBTQ Worldmaking* 4 (2): 93–99. https://doi.org/10.14321/qed.4.2.0093.
- Roy, Modhumita, and Mary Thompson, eds. 2019. *The Politics of Reproduction: Adoption, Abortion, and Surrogacy in the Age of Neoliberalism*. The Ohio State University Press. https://doi.org/10.26818/9780814214152.
- Ruiz-Sánchez, Héctor-Camilo. 2021. "Facing the Plagues Alone. Men Reshaping the HIV and Heroin Epidemics in Colombia." Ph.D., Pittsburgh, Pennsylvania: University of Pittsburgh. https://d-scholarship.pitt.edu/40829/.
- Russell, Stephen T., and Jessica N. Fish. 2016. "Mental Health in Lesbian, Gay, Bisexual, and Transgender (LGBT) Youth." *Annual Review of Clinical Psychology* 12 (March): 465–87. https://doi.org/10.1146/annurev-clinpsy-021815-093153.
- Sandbakken, Ella Marie, and Sigrun Marie Moss. 2021. "Now We Are All in the Same Boat. At the Same Time, We Are Not.' Meaning-Making and Coping Under COVID-19 Lockdown in Norway." *Human Arenas*, May. https://doi.org/10.1007/s42087-021-00208-z.

- Sargsyan, Nelli. 2019. "Experience-Sharing as Feminist Praxis: Imagining a Future of Collective Care." *History and Anthropology* 30 (1): 67–90. https://doi.org/10.1080/02757206.2018.1531287.
- Saskatchewan Bureau of Statistics. 2022. "Saskatchewan Immigration and Ethnocultural Diversity 2021 Census of Canada." October 26, 2022. https://pubsaskdev.blob.core.windows.net/pubsask-prod/137595/2021%252BCensus%252BSaskatchewan%252Bimmigration%252Band%252BEthnocultural%252BDiversity.pdf.
- Saskatchewan Catholic School Boards Association. n.d. "History of the SCBSA." Accessed March 7, 2022. https://www.scsba.ca/about-us/history-of-the-scbsa/.
- Schaffnit, Susan B., and Rebecca Sear. 2017. "Support for New Mothers and Fertility in the United Kingdom: Not All Support Is Equal in the Decision to Have a Second Child." *Population Studies* 71 (3): 345–61. https://doi.org/10.1080/00324728.2017.1349924.
- Scheffler, H.W. 1973. "Kinship, Descent, and Alliance." In *Handbook of Social and Cultural Anthropology*, edited by John J. Honigmann and Alexander Alland. Rand McNally Anthropology Series. Chicago: Rand McNally Co.
- Scheper-Hughes, Nancy. 1993. *Death Without Weeping: The Violence of Everyday Life in Brazil.* University of California Press.
- Scheper-Hughes, Nancy, and Margaret M. Lock. 1987. "The Mindful Body: A Prolegomenon to Future Work in Medical Anthropology." *Medical Anthropology Quarterly* 1 (1): 6–41.
- Schneider-Mayerson, Matthew. 2022. "The Environmental Politics of Reproductive Choices in the Age of Climate Change." *Environmental Politics* 31 (1): 152–72. https://doi.org/10.1080/09644016.2021.1902700.
- Schneider-Mayerson, Matthew, and Kit Ling Leong. 2020. "Eco-Reproductive Concerns in the Age of Climate Change." *Climatic Change* 163 (2): 1007–23. https://doi.org/10.1007/s10584-020-02923-y.
- Schormans, Ann Fudge, Sue Hutton, Marissa Blake, Kory Earle, and Kevin John Head. 2021. "Social Isolation Continued: Covid-19 Shines a Light on What Self-Advocates Know Too Well." *Qualitative Social Work* 20 (1–2): 83–89. https://doi.org/10.1177/1473325020981755.
- Schwartz, Elizabeth F. 2016. "LGBT Issues in Surrogacy: Present and Future Challenges." In *Handbook of Gestational Surrogacy: International Clinical Practice and Policy Issues*, edited by E. Scott Sills, 55–61. Cambridge: Cambridge University Press.
- Scoffield, Heather. 2012. "Census: Nuclear Family Is No Longer the Norm in Canada." *Toronto Star*, September 19, 2012, sec. Canada. https://www.thestar.com/news/canada/2012/09/19/census\_nuclear\_family\_is\_no\_longer\_thenorm in canada.html.
- Seidman, Steven. 1997. *Difference Troubles: Queering Social Theory and Sexual Politics*. Cambridge Cultural Social Studies. Cambridge; New York: Cambridge University Press.
- Senderowicz, Leigh, and Jenny Higgins. 2022. "Reproductive Autonomy Is Nonnegotiable, Even in the Time of COVID-19." *International Perspectives on Sexual and Reproductive Health* 46: 147–51.
- Sharif, Nadim, and Shuvra Kanti Dey. 2021. "Impact of Population Density and Weather on COVID-19 Pandemic and SARS-CoV-2 Mutation Frequency in Bangladesh." *Epidemiology & Infection* 149. https://doi.org/10.1017/S0950268821000029.

- Sharma, Rahul. 2013. "The Family and Family Structure Classification Redefined for the Current Times." *Journal of Family Medicine and Primary Care* 2 (4): 306. https://doi.org/10.4103/2249-4863.123774.
- Shead, Sam. 2021. "Climate Change Is Making People Think Twice about Having Children." *CNBC*, August 12, 2021. https://www.cnbc.com/2021/08/12/climate-change-is-making-people-think-twice-about-having-children.html.
- Shek, Daniel T. L. 2021. "COVID-19 and Quality of Life: Twelve Reflections." *Applied Research in Quality of Life* 16 (1): 1–11. https://doi.org/10.1007/s11482-020-09898-z.
- Sheppard, Amanda J., Gabriel D. Shapiro, Tracey Bushnik, Russell Wilkins, Serenity Perry, Jay S. Kaufman, Michael S. Kramer, and Seungmi Yang. 2017. "Birth Outcomes among First Nations, Inuit and Métis Populations." November 15, 2017. http://www150.statcan.gc.ca/n1/pub/82-003-x/2017011/article/54886-eng.htm.
- Shivaram, Deepa. 2022. "Abortion Wasn't Always the Politically Charged Issue It Is Today." *NPR*, May 4, 2022, sec. Politics. https://www.npr.org/2022/05/04/1096719971/abortion-wasn-t-always-the-politically-charged-issue-it-is-today.
- Shore, Cris, Victoria A. Goddard, and Josep R. Llobera. 2021. *The Anthropology of Europe: Identities and Boundaries in Conflict.* Routledge.
- Sienicka, Agnieszka, Agata Pisula, Katarzyna Karina Pawlik, Joanna Kacperczyk-Bartnik, Pawel Bartnik, Agnieszka Dobrowolska-Redo, and Ewa Romejko-Wolniewicz. 2022. "The Impact of COVID-19 Pandemic on Reproductive Intentions among the Polish Population." *Ginekologia Polska* 93 (5): 345–50. https://doi.org/10.5603/GP.a2021.0135.
- Singh, Shweta, Deblina Roy, Krittika Sinha, Sheeba Parveen, Ginni Sharma, and Gunjan Joshi. 2020. "Impact of COVID-19 and Lockdown on Mental Health of Children and Adolescents: A Narrative Review with Recommendations." *Psychiatry Research* 293 (November): 113429. https://doi.org/10.1016/j.psychres.2020.113429.
- Siu, Judy Yuen-man. 2008. "The SARS-Associated Stigma of SARS Victims in the Post-SARS Era of Hong Kong." *Qualitative Health Research* 18 (6): 729–38. https://doi.org/10.1177/1049732308318372.
- Sjöberg, Ida, and Katarina Giritli Nygren. 2021. "Contesting City Safety Exploring (Un)Safety and Objects of Risk from Multiple Viewpoints." *Journal of Risk Research* 24 (10): 1251–65. https://doi.org/10.1080/13669877.2020.1819391.
- Smith, Jonathan A., and Pnina Shinebourne. 2012. "Interpretative Phenomenological Analysis." In *APA Handbook of Research Methods in Psychology, Vol 2: Research Designs: Quantitative, Qualitative, Neuropsychological, and Biological*, 73–82. APA Handbooks in Psychology®. Washington, DC, US: American Psychological Association. https://doi.org/10.1037/13620-005.
- Smith, Julia. 2022. "From 'Nobody's Clapping for Us' to 'Bad Moms': COVID-19 and the Circle of Childcare in Canada." *Gender, Work & Organization* 29 (1): 353–67. https://doi.org/10.1111/gwao.12758.
- Sodik, Muhammad Ali, and Simon Martin Manyanza Nzilibili. 2017. "The Role Of Health Promotion And Family Support With Attitude Of Couples Childbearing Age In Following Family Planning Program In Health." *Journal of Global Research in Public Health* 2 (2): 82–89. https://doi.org/10.30994/jgrph.v2i2.105.
- Spadafora, Natalie, Caroline Reid-Westoby, Molly Pottruff, and Magdalena Janus. 2022. "Family Responsibilities and Mental Health of Kindergarten Educators during the First

- COVID-19 Pandemic Lockdown in Ontario, Canada." *Teaching and Teacher Education* 115 (July): 103735. https://doi.org/10.1016/j.tate.2022.103735.
- Spartacus. 2021. "Gay Travel Index 2021." Spartacus International Gay Guide. March 31, 2021. https://spartacus.gayguide.travel/gaytravelindex.pdf.
- Spradley, James P. 2016. The Ethnographic Interview. Waveland Press.
- Standing Senate Committee on Human Rights. 2021. "Forced and Coerced Sterilization of Persons in Canada." https://sencanada.ca/content/sen/committee/432/RIDR/reports/2021-06-03 ForcedSterilization E.pdf.
- Stanley, B. Liahnna, Alaina C. Zanin, Brianna L. Avalos, Sarah J. Tracy, and Sophia Town. 2021. "Collective Emotion During Collective Trauma: A Metaphor Analysis of the COVID-19 Pandemic." *Qualitative Health Research* 31 (10): 1890–1903. https://doi.org/10.1177/10497323211011589.
- Stardust, Zahra, Carla Treloar, Elena Cama, and Jules Kim. 2021. "I Wouldn't Call the Cops If I Was Being Bashed to Death': Sex Work, Whore Stigma and the Criminal Legal System." *International Journal for Crime, Justice and Social Democracy* 10 (3): [142]-157. https://doi.org/10.3316/informit.026527683507995.
- Statistics Canada. 2012. "Portrait of Families and Living Arrangements in Canada." 2012. http://www12.statcan.gc.ca/census-recensement/2011/as-sa/98-312-x/98-312-x2011001-eng.cfm.
- . 2016. "Aboriginal Peoples: Fact Sheet for Saskatchewan." March 14, 2016. http://www150.statcan.gc.ca/n1/pub/89-656-x/89-656-x2016009-eng.htm.
- ——. 2017. "2016 Census: The Canadian Families of Today and Yesteryear." August 2, 2017. https://www.statcan.gc.ca/en/sc/video/2016census-families.
- ——. 2021a. "A Statistical Portrait of Canada's Diverse LGBTQ2+ Communities." June 15, 2021. http://www150.statcan.gc.ca/n1/daily-quotidien/210615/dq210615a-eng.htm.
- ——. 2021b. "Crime Severity Index and Weighted Clearance Rates, Police Services in Saskatchewan." July 27, 2021.
  - http://www150.statcan.gc.ca/t1/tbl1/en/tv.action?pid=3510006101.
- ——. 2021c. "Crude Birth Rate, Age-Specific Fertility Rates and Total Fertility Rate (Live Births)." September 29, 2021.
  - http://www150.statcan.gc.ca/t1/tbl1/en/tv.action?pid=1310041801.
- ——. 2021d. "Mean Age of Mother at Time of Delivery (Live Births)." September 29, 2021. http://www150.statcan.gc.ca/t1/tbl1/en/tv.action?pid=1310041701.
- ——. 2022a. "Research to Insights: Immigration as a Source of Labour Supply." June 22, 2022. http://www150.statcan.gc.ca/n1/pub/11-631-x/11-631-x2022003-eng.htm.
- ———. 2022b. "The Daily Indigenous Population Continues to Grow and Is Much Younger than the Non-Indigenous Population, Although the Pace of Growth Has Slowed." September 21, 2022. http://www150.statcan.gc.ca/n1/daily-quotidien/220921/dq220921a-eng.htm.
- Stevenson, Allyson. 2020. *Intimate Integration: A History of the Sixties Scoop and the Colonization of Indigenous Kinship*. University of Toronto Press.
- Stone, Linda. 2002. New Directions in Anthropological Kinship. Rowman & Littlefield Publishers.

- Stoneman, Paul, Nick Allum, Niazy Hazeldine, and Andrew King. 2020. "Vigilance Within the LGBTQ+ Community and Its Effect on Mental Wellbeing." SSRN Scholarly Paper. Rochester, NY. https://doi.org/10.2139/ssrn.3745320.
- Strathern, Marilyn. 1992. Reproducing the Future: Essays on Anthropology, Kinship and the New Reproductive Technologies. Manchester University Press.
- Stults, Christopher B., Sandra A. Kupprat, Kristen D. Krause, Farzana Kapadia, and Perry N. Halkitis. 2017. "Perceptions of Safety among LGBTQ People Following the 2016 Pulse Nightclub Shooting." *Psychology of Sexual Orientation and Gender Diversity* 4 (3): 251–56. https://doi.org/10.1037/sgd0000240.
- Sullivan, Maureen. 2004. The Family of Woman: Lesbian Mothers, Their Children, and the Undoing of Gender.
- Sullivan, Nikki, and Sara Davidmann. 2015. "Reimag(in)Ing Life Making, or Queering the Somatechnics of Reproductive Futurity." In *Critical Kinship Studies*, edited by Charlotte Kroløkke, Lene Myong, Stine Willum Adrian, and Tine Tjørnhøj-Thomsen, 239–54. Rowman & Littlefield.
- Sullivan, T. Richard. 2013. *Queer Families, Common Agendas: Gay People, Lesbians, and Family Values*. Routledge. https://doi.org/10.4324/9780203057025.
- Sultana, Farhana. 2021. "Climate Change, COVID-19, and the Co-Production of Injustices: A Feminist Reading of Overlapping Crises." *Social & Cultural Geography* 22 (4): 447–60. https://doi.org/10.1080/14649365.2021.1910994.
- Swift, Jonathan. 1729. A Modest Proposal for Preventing the Children of Poor People from Being a Burthen to Their Parents, or the Country, and for Making Them Beneficial to the Publick. Eighteenth Century Collections Online. S. Harding. http://library.usask.ca/scripts/remote?URL=http://find.galegroup.com/ecco/infomark.do?contentSet=ECCOArticles&docType=ECCOArticles&bookId=0099302700&type=getFullCitation&tabID=T001&prodId=ECCO&docLevel=TEXT\_GRAPHICS&version=1.0&source=library&userGroupName=usaskmain.
- Taekema, Dan. 2020. "5 Babies Test Negative for COVID-19 after Outbreak at Hospital Unit for Premature Infants | CBC News." *CBC*, April 8, 2020. https://www.cbc.ca/news/canada/hamilton/nursery-covid-babies-1.5526238.
- Tam, Benita Y., Leanne C. Findlay, and Dafna E. Kohen. 2017. "Indigenous Families: Who Do You Call Family?" *Journal of Family Studies* 23 (3): 243–59. https://doi.org/10.1080/13229400.2015.1093536.
- Tam, Michelle W. 2021. "Queering Reproductive Access: Reproductive Justice in Assisted Reproductive Technologies." *Reproductive Health* 18 (1): 164. https://doi.org/10.1186/s12978-021-01214-8.
- Tamis-LeMonda, Catherine S., Shuyuan Wang, Eugenia Koutsouvanou, and Martina Albright. 2002. "Childrearing Values in Greece, Taiwan, and the United States." *Parenting* 2 (3): 185–208. https://doi.org/10.1207/S15327922PAR0203\_01.
- Tasker, John Paul. 2022. "Canadian Conservatives Reluctant to Comment on Report That U.S. Supreme Court Will Overturn Abortion Law." *CBC*, May 3, 2022. https://www.cbc.ca/news/politics/canadian-conservatives-supreme-court-roe-wade-1.6439660.
- Tate, Charlotte Chuck. 2012. "Considering Lesbian Identity from a Social-Psychological Perspective: Two Different Models of 'Being a Lesbian." *Journal of Lesbian Studies* 16 (1): 17–29. https://doi.org/10.1080/10894160.2011.557639.

- Testa, Maria Rita, Tomáš Sobotka, and Philip S. Morgan. 2011. "Reproductive Decision-Making: Towards Improved Theoretical, Methodological and Empirical Approaches." *Vienna Yearbook of Population Research* 9: 1–9.
- Trnka, Susanna, and Sharyn Graham Davies. 2021. "Blowing Bubbles: COVID-19, New Zealand's Bubble Metaphor, and the Limits of Households as Sites of Responsibility and Care." In *COVID-19: Global Pandemic, Societal Responses, Ideological Solutions*, edited by J. Michael Ryan. Vol. 1. London: Taylor & Francis.
- Tudge, Jonathan R.H., Diane M. Hogan, Irina A. Snezhkova, Natalya N. Kulakova, and Kathleen E. Etz. 2000. "Parents' Child-Rearing Values and Beliefs in the United States and Russia: The Impact of Culture and Social Class." *Infant and Child Development* 9 (2): 105–21. https://doi.org/10.1002/1522-7219(200006)9:2<105::AID-ICD222>3.0.CO;2-Y.
- Umereweneza, Patience, Marie Lovrod, Isobel M Findlay, Crystal Giesbrecht, Manuela Valle-Castro, Natalya Mason, and Renee Hoffart. 2020. "Sexual Violence in Saskatchewan: Voices, Stories, Insights, and Actions from the Front Lines." Saskatchewan: Community-University Institute for Social Research. https://artsandscience.usask.ca/news/files/900/SASS-CUISR\_Sexual\_Viiolence\_Final\_Report\_April\_30\_(1).pdf.
- Valdez, Natali, and Daisy Deomampo. 2019. "Centering Race and Racism in Reproduction." *Medical Anthropology* 38 (7): 551–59. https://doi.org/10.1080/01459740.2019.1643855.
- Valentine, Gill. 2002. "Queer Bodies and the Production of Space." In *Handbook of Lesbian and Gay Studies*, edited by Diane Richardson and Steven Seidman, 145–60. SAGE.
- Van Eeden-Moorefield, Brad, Christopher R. Martell, Mark Williams, and Marilyn Preston. 2011. "Same-Sex Relationships and Dissolution: The Connection Between Heteronormativity and Homonormativity." *Family Relations* 60 (5): 562–71. https://doi.org/10.1111/j.1741-3729.2011.00669.x.
- Veltman, Albina, and Gary Chaimowitz. 2014. "Mental Health Care for People Who Identify as Lesbian, Gay, Bisexual, Transgender, and (or) Queer." *Canadian Journal of Psychiatry*. *Revue Canadienne de Psychiatrie* 59 (11): 1–7.
- Vescera, Zak. 2022. "Abortion in Saskatchewan Not yet Accessible for All, Doctors Say." *The StarPhoenix*, May 28, 2022. https://thestarphoenix.com/news/saskatchewan/abortion-in-saskatchewan-not-yet-accessible-for-all-doctors-say.
- Vieira, Mauro Luís, Maria Aparecida Crepaldi, Carina Nunes Bossardi, Lauren Beltrão Gomes, Simone Dill Azeredo Bolze, and Cesar Augusto Piccinini. 2013. *Paternity in the Brazilian Context. Parenting in South American and African Contexts*. IntechOpen. https://doi.org/10.5772/57027.
- Vlieger, Dan de. 2007. "Political History of Saskatchewan." In *The Encyclopedia of Saskatchewan*. Regina, Sask.: University of Regina. https://esask.uregina.ca/entry/political history of saskatchewan.jsp.
- Wainberg, Mark A., Talia Shuldiner, Karine Dahl, and Norbert Gilmore. 2010. "Reconsidering the Lifetime Deferral of Blood Donation by Men Who Have Sex with Men." *CMAJ* 182 (12): 1321–24. https://doi.org/10.1503/cmaj.091476.
- Waldram, James B., and Andrew R. Hatala. 2022. "Does 'Susto' Really Exist? Indigenous Knowledge and Fright Disorders Among Q'eqchi' Maya in Belize." *Culture, Medicine, and Psychiatry*, March. https://doi.org/10.1007/s11013-022-09777-2.
- Walsh, Froma. 2020. "Loss and Resilience in the Time of COVID-19: Meaning Making, Hope, and Transcendence." *Family Process* 59 (3): 898–911. https://doi.org/10.1111/famp.12588.

- Wang, Shuyuan, and Catherine S. Tamis-Lemonda. 2003. "Do Child-Rearing Values in Taiwan and the United States Reflect Cultural Values of Collectivism and Individualism?" *Journal of Cross-Cultural Psychology* 34 (6): 629–42. https://doi.org/10.1177/0022022103255498.
- Ward, Jane. 2016. "Dyke Methods: A Meditation on Queer Studies and the Gay Men Who Hate It." *WSQ: Women's Studies Quarterly* 44 (3): 68–85. https://doi.org/10.1353/wsq.2016.0036.
- Ward, Sally. 2003. "On Shifting Ground: Changing Formulations of Place in Anthropology." *The Australian Journal of Anthropology* 14 (1): 80–96. https://doi.org/10.1111/j.1835-9310.2003.tb00222.x.
- Warick, Jason. 2016. "The Hate Still Exists': Saskatoon Man Says He Was Beaten for Being Gay." *The StarPhoenix*, September 12, 2016. https://thestarphoenix.com/news/crime/the-hate-still-exists-saskatoon-man-says-he-was-beaten-for-being-gay.
- Warner, Michael. 2000. *The Trouble with Normal: Sex, Politics, and the Ethics of Queer Life.* Harvard University Press.
- Warren, Carol A.B. 2012. "Interviewing as Social Interaction." In *The SAGE Handbook of Interview Research: The Complexity of the Craft*, edited by Jaber F. Gubrium, James A. Holstein, Amir B. Marvasti, and Karyn D. McKinney, 2nd ed., 129–42. Thousand Oaks, CA: SAGE. https://link.gale.com/apps/doc/CX3721700018/GVRL?sid=bookmark-GVRL&xid=1a0055de.
- Watkinson, Alisa. 2022. "Ailsa Watkinson: Christian Schools Funding in Sask. Requires Debate." *The StarPhoenix*, November 24, 2022. https://thestarphoenix.com/opinion/columnists/ailsa-watkinson-christian-schools-funding-in-sask-requires-debate.
- Watson, Ash, Deborah Lupton, and Mike Michael. 2021. "Enacting Intimacy and Sociality at a Distance in the COVID-19 Crisis: The Sociomaterialities of Home-Based Communication Technologies." *Media International Australia* 178 (1): 136–50. https://doi.org/10.1177/1329878X20961568.
- Weems, Lisa. 2019. "Thinking Queer About the Space of School Safety: Violence and Dis/Placement of LGBTQ Youth of Color." In *Queer Pedagogies: Theory, Praxis, Politics*, edited by Cris Mayo and Nelson M. Rodriguez, 93–108. Critical Studies of Education. Cham: Springer International Publishing. https://doi.org/10.1007/978-3-030-27066-7\_7.
- Weissman, Anna L. 2017. "Repronormativity and the Reproduction of the Nation-State: The State and Sexuality Collide." *Journal of GLBT Family Studies* 13 (3): 277–305. https://doi.org/10.1080/1550428X.2016.1210065.
- Welldon, Estela V. 1988. *Mother, Madonna, Whore: The Idealization and Denigration of Motherhood.* London: Routledge. https://doi.org/10.4324/9780429477348.
- Wenham, Clare, Julia Smith, Sara E. Davies, Huiyun Feng, Karen A. Grépin, Sophie Harman, Asha Herten-Crabb, and Rosemary Morgan. 2020. "Women Are Most Affected by Pandemics Lessons from Past Outbreaks." *Nature* 583 (7815): 194–98. https://doi.org/10.1038/d41586-020-02006-z.
- Weston, Kath. 1991. Families We Choose: Lesbians, Gays, Kinship. Between Men--between Women. New York: Columbia University Press.
- White-Crumney, Arthur. 2018. "To Pray or Not to Pray? The Place of God in the Sask. Legislature | Regina Leader Post." *Regina Leader-Post*, November 2, 2018. https://leaderpost.com/news/saskatchewan/to-pray-or-not-to-pray-the-place-of-god-in-the-legislature.

- Wiegers, Wanda. 2010. "Fatherhood and Misattributed Genetic Paternity in Family Law." *Queen's Law Journal* 36 (2): 623–72.
- Wilkinson, Eleanor. 2020. "Never after? Queer Temporalities and the Politics of Non-Reproduction." *Gender, Place & Culture* 27 (5): 660–76. https://doi.org/10.1080/0966369X.2019.1598941.
- Williams, Alex. 2021. "To Breed or Not to Breed?" *The New York Times*, November 20, 2021, sec. Style. https://www.nytimes.com/2021/11/20/style/breed-children-climate-change.html.
- Williams, Raymond. 1977. Marxism and Literature. New York: OUP Oxford.
- Williamson, Linzi EA, Karen L. Lawson, Pamela J. Downe, and Roger A. Pierson. 2014. "Informed Reproductive Decision-Making: The Impact of Providing Fertility Information on Fertility Knowledge and Intentions to Delay Childbearing." *Journal of Obstetrics and Gynaecology Canada* 36 (5): 400–405. https://doi.org/10.1016/S1701-2163(15)30585-5.
- Wilson, Ara. 2019. "Queer Anthropology." *Cambridge Encyclopedia of Anthropology*, July. https://www.anthroencyclopedia.com/entry/queer-anthropology.
- Wingo, Erin, Natalie Ingraham, and Sarah C. M. Roberts. 2018. "Reproductive Health Care Priorities and Barriers to Effective Care for LGBTQ People Assigned Female at Birth: A Qualitative Study." *Women's Health Issues* 28 (4): 350–57. https://doi.org/10.1016/j.whi.2018.03.002.
- Xu, Yanfeng, Qi Wu, Sue E. Levkoff, and Merav Jedwab. 2020. "Material Hardship and Parenting Stress among Grandparent Kinship Providers during the COVID-19 Pandemic: The Mediating Role of Grandparents' Mental Health." *Child Abuse & Neglect*, Protecting children from maltreatment during COVID-19: First volume, 110 (December): 104700. https://doi.org/10.1016/j.chiabu.2020.104700.
- Yan, Yunxiang. 2001. "Practicing Kinship in Rural North China." In *Relative Values:* Reconfiguring Kinship Studies, edited by Sarah Franklin and Susan McKinnon, 224–45. Duke University Press.
- Yazzie, Robert. 2000. "Indigenous Peoples and Postcolonial Colonialism." In *Reclaiming Indigenous Voice and Vision*, edited by Marie Ann Battiste, 39–49. Vancouver: UBC Press.
- Younkin, Felisha L, Michael Firmin, and Taylor Zurlinden. 2022. "Engaged Students' Perspectives: Stability as the Key Factor in Family Planning." Defense Technical Information Center. https://apps.dtic.mil/sti/pdfs/AD1169545.pdf.
- Zamarro, Gema, Francisco Perez-Arce, and Maria Jose Prados. 2020. "Gender Differences in the Impact of COVID-19." University of Southern California.
- Zang, Sheryl M., Ivy Benjenk, Suellen Breakey, Eleonor Pusey-Reid, and Patrice K. Nicholas. 2021. "The Intersection of Climate Change with the Era of COVID-19." *Public Health Nursing* 38 (2): 321–35. https://doi.org/10.1111/phn.12866.
- Zhu, Chenfeng, Jiahao Wu, Yan Liang, Li Yan, Chuqing He, Luting Chen, and Jian Zhang. 2020. "Fertility Intentions among Couples in Shanghai under COVID-19: A Cross-Sectional Study." *International Journal of Gynecology & Obstetrics* 151 (3): 399–406. https://doi.org/10.1002/ijgo.13366.
- Zimmerman, Linnea A., Celia Karp, Mary Thiongo, Peter Gichangi, Georges Guiella, Alison Gemmill, Caroline Moreau, and Suzanne O. Bell. 2022. "Stability and Change in Fertility Intentions in Response to the COVID-19 Pandemic in Kenya." *PLOS Global Public Health* 2 (3): e0000147. https://doi.org/10.1371/journal.pgph.0000147.

Zimonjic, Peter. 2022. "Number of Singles, Common-Law Relationships and Roommates Rises as Canada's Households Evolve | CBC News." *CBC*, July 13, 2022. https://www.cbc.ca/news/politics/canada-changing-households-military-1.6519440.

#### **APPENDIX**

# A - Recruitment Advertisement

# Seeking LGBTQ2S+ volunteers for research on decisions about parenthood

If you are an LGBTQ2S+ person in Saskatchewan between the ages of 18 and 35, we are interested in hearing how and when you make decisions regarding parenthood and the factors that might play a role in these decisions. Whether you currently have children, intend to have children, or have decided against having children, we invite you to share your story.

As a participant in this study, you would be asked to take part in a 60-90 minute remote interview about how and when you made your decisions regarding parenthood. You will receive a \$25 honorarium in thanks for your time and sharing.

This study has been reviewed by, and received approval through the Research Ethics Office, University of Saskatchewan.

### Please contact:

Jessica Jack or Dr. Pamela Downe

Department of Archaeology and Anthropology, University of Saskatchewan

Provide your contact info at:

https://www.surveymonkey.ca/r/DecisionsParenthood Email: jessica.jack@usask.ca or pamela.downe@usask.ca

Phone: 306 966 1974

# B - Interview Guide

Today I would like to talk to you about how you arrived at your current parenthood status (whether you have children or not). There are many aspects of this that I would like to discuss with you, and there are no right or wrong answers. I would like to hear about your thoughts and experiences.

- First, I'd like to learn about you. How do you describe yourself?
  - What is your age?
  - What is your gender identity? / Do you identify as a man, a woman, or another gender identity?
  - What is your sexual orientation? / Do you identify as straight, gay, bisexual, queer, or another sexual orientation?
  - What do you do for a living?
  - Where were you born? (if not in Saskatchewan, how long have you lived in Saskatchewan? Do you have close ties to your birth country?)
  - Where do you currently live? How long have you lived there? Why do you live there? Have you ever lived anywhere else?
  - Where were your parents born? (if not in Saskatchewan, how long have they lived in Saskatchewan?)
- Tell me about your life in Saskatchewan.
  - Who are the most important people in your life?
  - What did your family look like when you were growing up?
  - What aspects of living in Saskatchewan are important to you?
- How has COVID-19 in Saskatchewan impacted your life?
  - What changes, if any, have you experienced to your life as a result of COVID-19?
  - Have you faced any hardships from the impacts of COVID-19?

- Have there been any positive effects in your life from the social response to the pandemic?
- How much control do you feel like you have over the way COVID-19 has impacted your life? In what areas of your life do you feel like you have control?
- How do you feel about the idea of becoming a parent?
  - If positive, why do you want to have children? If negative, why not?
  - If yes, when would you like to have them? Why that time and context?
  - Have your feelings about becoming / not becoming a parent changed over time? If so, why?
  - What are some important factors in your thoughts about becoming / not becoming a parent?
  - Is there anything about your gender/sexual identity that has impacted your feelings about parenthood?
  - How does the idea of parenthood fit into other plans or goals in your life?
  - In a perfect world, what would your life look like in terms of being a parent / not being a parent?
- What do you think about being a parent in Saskatchewan?
  - How does your life in Saskatchewan fit into your thoughts about becoming / not becoming a parent?
  - Are there benefits to being a parent in this province? Are there drawbacks?
  - How well do you think Saskatchewan supports people like you in their plans about being a parent / not being a parent?
  - If you could change anything about parenthood in Saskatchewan, what would you change?
- How does COVID-19 and its impacts on life in Saskatchewan factor into your thoughts about parenthood?

- You had mentioned [important factor from previous questions] has that changed at all because of COVID-19?
- What changes resulting from COVID-19 have had the most serious impact on your thoughts about parenthood?
- If the COVID-19 pandemic hadn't happened, do you think you would feel differently about parenthood?
- Do you think parenthood will be different after COVID-19 is no longer a serious problem in Saskatchewan? What do you think will change?